

https://theses.gla.ac.uk/

Theses Digitisation:

https://www.gla.ac.uk/myglasgow/research/enlighten/theses/digitisation/

This is a digitised version of the original print thesis.

Copyright and moral rights for this work are retained by the author

A copy can be downloaded for personal non-commercial research or study, without prior permission or charge

This work cannot be reproduced or quoted extensively from without first obtaining permission in writing from the author

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the author

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given

Enlighten: Theses <u>https://theses.gla.ac.uk/</u> research-enlighten@glasgow.ac.uk

Regulation of Phosphoenolpyruvate Carboxylase in Higher Plants

Katrina M. Paterson

Thesis submitted for the degree of doctor of philosophy

Division of Biochemistry and Molecular Biology Institute of Biomedical and Life Sciences University of Glasgow July, 1999

© Katrina M. Paterson, 1999

i ·

いたい ぼうえい 変化的 いわりがく

ProQuest Number: 10391213

All rights reserved

INFORMATION TO ALL USERS

The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



ProQuest 10391213

Published by ProQuest LLC (2017). Copyright of the Dissertation is held by the Author.

All rights reserved.

This work is protected against unauthorized copying under Title 17, United States Code Microform Edition © ProQuest LLC.

ProQuest LLC. 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106 – 1346



, , The work described in Chapter Three of this thesis has been accepted for publication and is in press as:

Paterson, KM and Nimmo, HG (2000) Effects of pH on the induction of phosphoenolpyruvate carboxylase kinase in *Kalanchoë fedtschenkoi*. Plant Science.



4.5.

Contents of Thesis

Y

· ·

			Page
Title			i
Publica	ations		й
Conter	nts		ili
List of	figures		viii
Abbrev	viations		xi
Ackno	wledgem	ients	xiii
Abstra	ct		xiv
Chapt	ter One:	Introduction	
1.1	The re	gulation of plant enzyme activity	1
1.2	Protei	n phosphorylation in plants	2
	1.2.1	Plant protein kinases	3
	1.2.2	Plant protein phosphatases	7
1.3	Phosp	hoenolpyruvate carboxylase in higher plants	9
	1.3.1	The reaction catalyzed by higher plants	9
	1.3.2	The regulation of PEPc	9
	1.3.3	The function of PEPc	11
	1.3.4	The structure of PEPc	12
1.4	Photo	synthesis and photorespiration	14
	1.4.1	The C ₃ pathway of photosynthesis	14
	1.4.2	Photorespiration	14
1.5	PEPc	and Crassulacean Acid Metabolism	16
	1.5.1	The role of PEPc in CAM	16
	1.5.2	The regulation of PEPc in the CAM plant K. fedtschenkoi	21

	1	The regulation of this of the CAW plant K. Jeuischenwit	A.L
1.6	PEPc	and C ₄ photosynthesis	26
	1.6.1	The role of PEPc in C ₄ photosynthesis	26
	1.6.2	The regulation of PEPc in C_4 plants	29
1.7	PEPc	and C ₃ photosynthesis	32
	1 .7.1	The role of PEPc in C ₃ photosynthesis	32

	1.7.2 The regulation of PEPc in C_3 plants	34
1.8	Molecular properties of phosphoenolpyruvate carboxylase	
(PEPc) k	tinase	39
1.9	The coordinate regulation of carbon and nitrogen metabolism	
in plants	by protein phosphorylation	39
1.10	Project objectives	41
Chapter	Two: Materials and Methods	
2.1	Materials and material sources	48
2.2	Experimental tissues	49
	2.2.1 Culture material	49
	2.2.2 Plant material	50
2.3	General biochemical methods	51
	2.3.1 pH calibrations	51
	2.3.2 Glassware and plastics	51
	2.3.3 Spectrophotometric assays	51
	2.3.4 Centrifugation	51
	2.3.5 Micropipetting	51
	2.3.6 Chromatographic materials	52
2.4	Buffers and solutions	52
2.5	Methods for the extraction of enzyme activities	53
	2.5.1 A. thaliana cell culture	53
	2.5.2 A. thaliana leaf extracts	54
	2.5.3 K. fedtschenkoi leaves and leaf disks	54
	2.5.4 Desalting crude extracts	54
2.6	Assay procedures	55
	2.6.1 Determination of PEPc activity and the apparent K_i of	
	PEPc for L-malate	55
	2.6.2 PEPc kinase activity	55
	2.6.3 Malate Dehydrogenase (MDH) activity	56
2.7	Gel electrophoresis techniques	56
	2.7.1 SDS/polyacrylamide gel electrophoresis	56

iv

	2.7.2	Coomassie staining of SDS/polyacrylamide gels	57
	2.7.3	Drying and autoradiography and phosphorimaging of gels	57
	2.7.4	Agarose gel electrophoresis of nucleic acids	57
	2.7.5	Denaturing agarose gel electrophoresis of RNA	58
2.8	North	ern and Southern blotting	58
	2.8.1	Capillary transfer of nucleic acids onto nylon membrane	58
	2.8.2	Random priming of cDNAs to synthesize radioactive probes	59
	2.8.3	Separation of labelled DNA from un-incorporated	
	radion	ucleotides using spin columns	59
	2.8.4	Hybridization of radiolabelled DNA probes to Northern	
	and So	outhern blots	60
	2.8.5	Stripping Hybond-N membranes of radioactive nucleotides	60
2.9	DNA	experiments	61
	2.9.1	Prevention of degradation of DNA samples by	
	DNas	e contamination	61
	2.9.2	Isolation of plasmid DNA	61
	2.9.3	Determination of nucleic acid concentration and quality	61
	2.9.4	Digestion of DNA with restriction endonucleases	62
	2.9.5	Polymerase Chain Reaction (PCR)	62
	2.9.6	Sequencing of DNA	62
	2.9.7	Design of primers	63
2.10	RNA	experiments	63
	2.10.1	Prevention of RNA sample degradation by	
	RNase	e contamination	63
	2.10.2	Isolation of RNA from plant tissue and cell culture	63
	2.10.3	Isolation of poly A ⁺ RNA	67
	2.10.4	In vitro translation of RNA samples	67
	2.10.5	RT-PCR	68
	2.10.6	In vitro transcription of A. thaliana PEPc kinase cDNA	68
2.11	Clonir	ng techniques	69
	2.11.1	Preparation of plating bacteria	69
	2.11.2	2 Titration of λ PRL2 cDNA library and λ ZIPlox recombinants	69

ν

	2.11.3 Plasmid preparation of A. thaliana ESTs	70
	2.11.4 Immobilization of λ ZIPLox recombinants onto	
	nitrocellulose membrane	70
	2.11.5 Hybridization using radiolabelled PEPc probe	70
	2.11.6 Isolation of putative positive phage	71
	2.11.7 In vivo excision of recombinant pKP42	71
2.12	Bradford assay for protein determination	71
2.13	Gas exchange measurements	72
2.14	Measurement of total leaf malate content	72

Chapter Three: The regulation of phosphoenolpyruvate carboxylase from *Kalanchoë fedtschenkoi* by phosphorylation

3.1	Introdu	uction	73
3.2	Result	s	74
	3.2.1	Cytosolic alkalinization has no affect on the malate	
	sensitiv	vity of PEPc and PEPc kinase in K. fedtschenkoi leaf disks	74
	3.2.2	Cytosolic acidification prevents the nocturnal increase in	
	the app	parent K _i of PEPc for L-malate and PEPc kinase activity	
	in K. fe	edtschenkoi leaf disks	74
	3.2.3	Optimisation of the incubation conditions	75
	3.2.4	Acetic acid affects PEPc via the acidification of cytosolic pH	80
	3.2.5	Time course of the effect of 3 mM acetic acid	84
	3.2.6	Cytosolic acidification prevents decarboxylation of	
	intrace	ilular malate	88
	3.2.7	PEPc inhibitors fail to prevent dark CO ₂ fixation	91
	3.2.8	The involvement of light in the effect of 3 mM acctic acid	
	on the	malate sensitivity of PEPc and malate content in	
	K. fed	<i>tschenkoi</i> disks	96
3.3	Discus	ssion	99

æ 1

.

Chapter Four: Cloning and sequencing of first full-length

phosphoenolpyruvate carboxylase cDNA from Arabidopsis thaliana

4.1	Introduction		108
4.2	Result	s	108
	4.2.1	Cloning of a full-length putative PEPc from an A. thaliana	
	cDNA	library	108
	4.2.2	Sequencing of the putative full length PEPc cDNA	111
	4.2.3	Sequence analysis of full-length A. thaliana PEPc cDNA	114
	4,2,4	Tissue expression study of A. thaliana PEPc clone	134
	4.2.5	Phylogenetic analysis of A. thaliana PEPc	138
4.3	Discus	sion	138

- 3

$Chapter\ Five:\ Characterization\ of\ phosphoenol pyruvate\ carboxylase$

and its kinase from the C_3 species Arabidopsis thaliana

5.1	Introd	uction	145
5.2	Result	s	1 46
	5.2.1	Development of an extraction protocol for PEPc activity	
	from A	t. thaliana cell culture	146
	5.2.2	Characterization of A. thaliana cell culture growth, PEPc	
	activity	y and the malate sensitivity of PEPe	148
	5.2.3	Manipulation of the malate sensitivity of PEPc in	
	A. tha	liana: effects of the protein phosphatase inhibitor	
	cantha	ridin and light	150
	5.2.4	PEPc and PEPc kinase transcript levels in A. thaliana	
	cell cu	lture and plant tissue	156
	5.2.5	The effect of sucrose and nitrate on the transcript levels	
	ofPEI	Pc and PEPc kinase in A. thaliana cell culture	161
5.3	Discus	sion	171
Chapte	r Six: E	Discussion	177

Chapter Seven: References

vii

List of Figures

		Page
Chapt	er One: Introduction	
1.1	The reaction catalyzed by PEPc	10
1.2	The reductive pentose phosphate pathway (RPP) cycle	15
1.3	Photosynthetic carbon oxidation (PCO) cycle	17
1.4	Compartmentation of CO_2 fixation reactions in C_4 and	
CAM	species	20
1.5	Diurnal variation in PEPc kinase protein, translatable mRNA	
and PE	Pc K, for L-malate	24
1.6	C_4 PEPc transduction pathway	33
1.7	Coordinate regulation of PEPc, SPS and NR in C3 leaves	42
1.8	Photograph of Kalanchoë (Bryophyllum) fedtschenkoi	44
1.9	Photograph of Arabidopsis thaliana var. Landsberg erecta	46
1.10	Photograph of Arabidopsis thaliana photomixotrophic cell	
suspen	sion culture	47
Chapt	er Two: Materials and Methods	
2.1	Alignment of primers for A. thaliana PEPc, PEPc kinase and	
α-tubi	alin	64
Chapt	er Three: The regulation of phosphoenolpyruvate carboxylase	
from <i>I</i>	Kalanchoë fedtschenkoi by phosphorylation	
3.1	The effect of acetic acid on the malate sensitivity of PEPc and	
PEPc 1	cinase activity of K. fedtschenkoi leaf disks	76
3.2	The malate sensitivity of PEPc and PEPc kinase activity of	
K. fedt	schenkoi leaves and leaf disks at night	78
3.3	The effect of shaking on the malate sensitivity of PEPc from	
K. fedt	schenkoi disks and leaves at night	81
3.4	The effect of different acids on the malate sensitivity of PEPc from	
K. fedt	<i>schenkoi</i> disks	82

3.5	The effect of acetic acid treatment length on the malate sensitivity	
ofPEPc	PEPc kinase activity and level of PEPc kinase translatable mRNA	85
3.6	The effect of acetic acid treatment length on the intracellular malate	
concentr	ation of K. fedtschenkoi leaf disks	89
3.7	The effect of CO_2 -free air and 3 mM acetic acid on the intracellular	
malate c	oncentration of K. fedtschenkoi leaf disks	92
3.8	The effect of CO_2 -free air and 3 mM acetic acid on the malate	
sensitivit	y of PEPc from K. fedtschenkoi leaf disks	94
3.9	The effect of DCDP and DMDP on the activity of purified	
K. fedtsc	chenkoi PEPc	95
3.10	The effect of the PEPc inhibitors DCDP and DMDP on the	
circadiar	rhythm of CO2 output in K. fedtschenkoi leaves maintained in	
CO ₂ -free	e air and constant darkness at 15°C	97
3.11	The effect of light and 3 mM acetic acid on the malate sensitivity	
of PEPc	in <i>K. fedtschenkoi</i> disks	100
3.12	The effect of light and 3 mM acetic acid on the intracellular	
malate c	oncentration of K. fedtschenkoi leaf disks	101
Chapter	r Four: Cloning and sequencing of first full-length	
phosph	penolpyruvate carboxylase cDNA from Arabidopsis thaliana	
4.1	Hybridization of radiolabelled PEPc cDNA to PCR products	
from λP	RL2 cDNA library screen	110

19/6 1

- -

4.2	Purified phage containing putative full-length PEPc cDNA inserts	112
4.3	Sal I digest of purified recombinant plasmid pKP42	113
4.4	Interleaved nucleotide and amino acid sequence of a full-length	
PEPc cI	ONA from A. thaliana	115
4.5	Sequence alignments from higher plant PEPcs	118
4.6	Alignment of A. thaliana PEPc cDNA and genomic sequences	126
4.7	A comparison of intron length and location in PEPc genes from	
different	species	133
4.8	Alignment of A. thaliana PEPc cDNA with other A. thaliana	
PEPc se	quences	135
4.9	Labelled photograph of a flowering A. thaliana var. Landsberg	
erecta plant		

4.10	Expression of A. thaliana PEPc transcript in different tissues	
of A. tha	liana var. Landsberg erecta	137
4.11	Phylogenetic tree of A. thaliana PEPc and various other PEPc	
isoforms		139
Chapter	· Five: Characterization of phosphoenolpyruvate	
carboxy	lase and its kinase from the C_3 species <i>Arabidopsis thaliana</i>	
5.1	Determination of the malate sensitivity (apparent K_i) of PEPc in	
A, thalia	ma cell culture	147
5.2	Time course of cell growth in an A. thaliana cell culture	149
5.3	The specific activity and malate sensitivity of PEPc in A. thaliana	
cell cuitu	re	151
5.4	The effect of the phosphatase inhibitor cautharidin on the malate	
sensitivit	y of PEPc	154
5.5	The effect of light and dark on the PEPc malate sensitivity in	
A. thalia	ana cell culture and leaves	155
5.6	RT-PCR of PEPc kinase, PEPc and α -tubulin steady-state transcript	
from A.	thaliana cell culture	157
5.7	Diagram of the A. thaliana PEPc kinase EST in plasmid pZL1	159
5.8	Northern blot analysis of PEPc and PEPc kinase steady-state	
transcrip	t levels in A. thaliana tissue	162
5.9	Northern blot analysis of PEPc and PEPc kinase steady-state	
transcrip	t levels in A. thaliana cell culture	164
5.10	Northern blot analysis of PEPc and PEPc kinase steady-state	
transcrip	t levels in A. thaliana cell culture in response to altered sucrose	
and nitra	ite levels	167
5.11	PEPc and PEPc kinase steady-state transcript levels in A. thaliana	
in respoi	nse to altered sucrose and nitrate levels	i69

х

List of Abbreviations

The abbreviations used in this thesis are described in full below:

AAT, aspartate aminotransferase; ADP, adenosine diphosphate; AMP, adenosine monophosphate; AMPK, AMP-activated protein kinase; ATP, adenosine triphosphate; γ ³²PIATP, radioactively labelled adenosine triphosphate; BAC, bacterial artificial chromosome: bp. base pair: Ca²⁺, calcium ion, CAM, crassulacean acid metabolism: CaMK, calmodulin-dependent protein kinase; cAMP, cyclic adenosine monophosphate; cDNA, complementary deoxyribonucleic acid; CDK, cyclin-dependent kinase; CDPK, calciumdependent protein kinase (or calmodulin-like domain protein kinase); CK, casein kinase; CO₂, carbon dioxide; DCDP, 3,3-dichloro-2-dihydroxyphosphinoylmethyl-2-propenoate; DHAP, dihydroxyacetone phosphate; DCMU, 3(3,4-dichlorophenyl)-1,1-dimethylurea; DMDP, 3,3-dimethyl-2-dihydroxyphosphinoylmethyl-2-propenoate; DMO, 5,5-dimethyl-2,40xazolidinedione; DNA, deoxyribonucleic acid; DTT, dithiothreitol; EDTA, ethylene diamine tetra-acetic acid; EGTA, ethylene glycol-bis(β -aminoethyl ether)N, N, N'. N'-tetra acetic acid; EST, expressed sequence tag; gln, glutamine; glu, glutamate; GOGAT, glutamine 2-oxoglutarate aminotransferase (glutamate synthase); G3P, glyceraldehyde 3phosphate; GS, glutamine synthetase; GSK, glycogen synthasc kinase; GTP, guanosine triphosphate: HCl, hydrochloric acid; HMG-CoA reductase, 3-hydroxy-3-methylglutarylcoenzyme A reductase; KAPP, kinase associated protein phosphatase; kb, kilobase; Ki, inhibition constant; kDa, kilodalton; KNO₃, potassium nitrate; K⁺, potassium ion; LRR, leucine-rich repeat; MAPK, MAPKK or MAPKKK, mitogen activated protein kinase, kinase kinase or kinase kinase; MDH, malate dehydrogenase; ME, malic enzyme; [35S]Met, radioactively labelled methionine; mM, millimolar; mRNA, messenger ribonucleic acid; MSMO, Murashige & Skoog Minimal Organics; NAD⁺/NADH, nicotinamide adenine dinucleotide oxidized/reduced form; NADP⁺/NADPH, nicotinamide adenine dinucleotide phosphate oxidized/reduced form; NH₄Cl, ammonium chloride; NH₄NO₃, ammonium nitrate; NIP, nitrate reductase inhibitor protein; NO₃ nitrate; NR, nitrate reductase; OAA, oxaloacetate; O2, oxygen; PCO, photosynthetic carbon oxidation; PCR, polymerase chain reaction; PEP, phosphoenolpyruvate; PEPc, phosphoenolpyruvate carboxylase; 3-PGA, 3phosphogłycerate; P_i, inorganic phosphate; PPDK, pyruvate, phosphate dikinase; PP_i, inorganic phosphate; PP1, 2A, 2B or 2C, protein phosphatase type-1, -2A, -2B or -2C; RLK, receptor-like protein kinase; RNA, ribonucleic acid; RNase, ribonuclease; RPP, reductive pentose phosphate; RT, reverse transcription; Rubisco, ribulose 1,5-bisphosphate carboxylase/oxygenase; SDS-PAGE, sodium dodecyl sulphate polyacrylamide gel electrophoresis; SNF-1, sucrose non-fermenting protein kinase-1; SPS, sucrose phosphate synthase; SRK, S-locus receptor kinase; TP, triose phosphate; μ Ci, microcurie; UDP, uridine diphosphate; μ E, microEinstein; μ M, micromolar; UTR, untranslated region.

1. gift

Acknowledgements

It is not possible to acknowledge fully each person who has supported me throughout this research. There are many who, at just the right time, have given the appropriate advice or much-needed words of encouragement to keep the wheels turning. A sincere thank you to each and every one. However, special recognition must be given to the following people without whom the work would probably never have been completed.

7

į

•

I would like to thank Prof. Hugh G. Nimmo who has supervised my research. His guidance in experimental design and help in the interpretation of results has been invaluable. His contribution of time and experience to the writing of this thesis has been particularly appreciated.

I would also like to thank the BBSRC for their financial support and the following individuals for their technical support and assistance: Dr. John Christie, Dr. Aideen Gill, Jim Jardine, Carol Johnston, Helena Wade and Sue at MBSU.

Throughout my research experience Dr. James Hartwell has been a strength and guide, never exhausting in his patience and encouragement. His enthusiasm and dedication to furthering all aspects of research in the Bond Lab have been an inspiration and his friendship very much enjoyed. I wish him every success for his future in research and every happiness in the life that he shares with Susie. And I would like to thank the other Bond Lab Boys for the laughs and banter along the way and the continual reassurance that I am still sane.

The friendships of Judith, Niall, Gordon and Mark have kept my life in perspective and given me many laughs - I hope they never end. My friendship with Tim over this last year cannot be forgotten; I thank him for the times that we have shared and the things that he has taught me. The fellowship and prayer support of friends at Greenview Evangelical Church has been a tremendous encouragement and the hospitality and care of Tom and Moira Afleck every Sunday for nearly two years has been a blessing.

My parents are wonderful! Dad has given me the stubborn streak that kept me persevering throughout this research; Mum has given me many packets of Skittles. Both have always supported and always loved - thank you.

And finally, for all those mentioned above, the mind with which I think and the hope with which I live, I thank God, my peace and constant companion throughout it all.

Abstract

The regulation of higher plant phosphoenolpyruvate carboxylase (PEPc) has been the focus of intensive research for many years. In CAM, C₄ and C₃ species the enzyme's sensitivity to feedback inhibition by malate is modulated by reversible phosphorylation of a serine residue by a calcium-independent PEPc kinase. This kinase is controlled by a circadian oscillator in CAM plants whereas in C₄ and C₃ plants the activity is induced by light. In all cases, PEPc kinase activity is regulated by a process involving *de novo* protein synthesis as opposed to a phosphorylation cascade or a second messenger. Nitrate has also been implicated in the regulation of PEPc from C₃ species and the involvement of cytosolic alkalinization and calcium ion movements has been demonstrated in the signalling pathway by which light triggers phosphorylation of PEPc in C₄ plants.

The overall objective of this present research was to investigate the physiological roles of the phosphorylation of PEPe and the signalling pathways that bring this about. In the first part of the work, the involvement of cytosolic pH in the regulation of PEPe was studied in the CAM plant *K. fedtschenkoi*. An increase in cytosolic pH had no affect on the malate sensitivity of PEPe or the PEPe kinase activity. In contrast, treatment of leaf disks with 3 mM acetic acid to reduce cytosolic pH prevented the nocturnal increase in the apparent K_i of PEPe for malate, PEPe kinase activity and PEPe kinase translatable mRNA. The effect was not metabolic but was due to cytosolic acidification. Measurement of the malate content of leaf disks revealed that acetic acid treatment prevented the decarboxylation of malate. It is proposed that when the cytosolic malate concentration is above a certain threshold at the end of a day period, PEPe kinase activity is inhibited during the following night and PEPe remains in a dephosphorylated, inactive form. The effect of acid treatment was slow, requiring 9-12 hours, and was therefore not considered to be involved in CAM PEPe signal transduction.

In the second part of this research aspects of PEPc regulation were studied in the model C_3 species *A. thaliana*. The first full-length cDNA for PEPc from *A. thaliana* was cloned and sequenced. Northern analysis of poly A^+ RNA from different *A. thaliana* plant tissues suggested that the cDNA encoded a root isoform of PEPc. The specific activity and malate sensitivity of PEPc were measured in a photomixotrophic cell culture of *A. thaliana* were measured. The phosphatase inhibitor, cantharidin, decreased the malate sensitivity of PEPc from the cell culture but light did not have any effect on the malate

xiv

sensitivity of PEPc from the cell culture or *A. thaliana* plant tissue. Using Northern analysis of poly A^+ RNA transcripts for PEPc and PEPc kinase were detected in the cell culture. PEPc transcripts were detected in root and flower/bud tissue using a conserved PEPc probe and PEPc kinase transcripts were detected in several different tissues of *A. thaliana*. The effects of sucrose and nitrate on the transcript levels of PEPc and PEPc kinase were higher in 1% sucrose compared with 3% sucrose but no apparent trend was evident in the effect of nitrate on either PEPc or PEPc kinase. Therefore, it can be concluded that PEPc kinase is present in both *A. thaliana* tissue and cell culture and that the malate sensitivity of PEPc from the cell culture is regulated by reversible phosphorylation. The potential use of the *A. thaliana* cell culture for studies of the regulation of PEPc in C₃ plants has been demonstrated.

- e.o. T&P, py-phosphato dependent programations F236-bisi activates tetrance + glycolgsis PP: Settrates denner 4 gluconegoponesis e.g. functionin, ferredoxin thioredoxin reductase and thioredoxin Ford et al., 1987 - protein maintaise', p. - p. interactions' thought to be insched, ambiguites, interactions - marine of the

Chapter One

INTRODUCTION

1.1 The regulation of plant enzyme activity

Plants and animals respond in different ways to their environments. The plant lifestyle, characterized by a lack of movement and the consequent inability to search for food or move away from threatening or adverse situations, necessitates a plasticity of metabolism, growth and development. This requires that external signals are recognised and used to stimulate internal responses; however, the mechanisms of signal transduction are still relatively poorly understood. At one level (e.g. gene expression or proteolysis) of these signalling pathways, changes are relatively slow. However, in contrast to this coarse control, another type of metabolic control exists to effect rapid (i.e. seconds to minutes) and generally energetically inexpensive responses within the plant to external stimuli. This fine control modulates the activity of pre-existing enzyme molecules, and thus the rate of flux through the various metabolic pathways, to meet the requirements of individual cells in response to perceived changes in the plant's environment. Thus, this regulation of enzyme activity is important in the overall metabolic profile of the plant.

The fine control of enzyme regulation can be brought about in different ways, which, it must be emphasized, often operate in concert. One type of fine control is a change in the concentration of a reaction substrate. This is particularly significant for enzymes which show sigmoidal substrate saturation kinetics. Enzyme activity can also be controlled by changes in intracellular pH, the extent to which this is important for each enzyme being determined by how broad or narrow the pH activity profile of that enzyme. Regulation of enzyme activity can also involve reversible associations of subunits within a multimeric protein, normally induced by an effector molecule to increase or decrease activity, or association of enzymes has been proposed as a mechanism facilitating the direct transfer of intermediates between sequential enzymes thus increasing metabolic flux. This enzyme association can also result in the alteration of the kinetic properties of enzymes, presumably by inducing conformational changes in proteins during interactions between different enzymes.

In comparison with the mechanisms of fine metabolic control just mentioned, much more is known about allosteric and covalent regulation of enzyme activity. Allosteric effectors can

alter enzyme activity by binding to a site distinct from the active site and eliciting a precise change in the enzyme's conformation. This conformational change can then either promote or hinder the interaction of the enzyme with its substrate to affect metabolic flux through a pathway. Covalent modification usually operates in conjunction with allosteric regulation, the covalent modification causing a conformational change in the enzyme which affects its substrate binding and can enhance or reduce the affect of allosteric molecules. Reversible covalent modification is the major mechanism of enzyme regulation used by higher eukaryotes to coordinate the appropriate response to external stimuli. Despite there being over one hundred different types of post-translational protein modification known to occur *in vivo*, very few are thought to be important in enzyme regulation and of these dithiol-sulfide interconversion and reversible phosphorylation are the principal mechanisms of enzyme regulation by covalent modification in higher plants.

ş

زم د

ŝ

Ì

. م

Reversible phosphorylation is of particular relevance to the topic of this research as it is a major mechanism of regulating phosphoenolpyruvate carboxylase (PEPc). However, as already mentioned, the different mechanisms of fine control are not mutually exclusive and this concept is illustrated in the research findings described in this thesis. But, before reviewing what is already known about PEPc in higher plants, it will be useful to first consider more thoroughly the process of reversible phosphorylation and the enzymes involved in this method of covalently modifying proteins.

1.2 Protein phosphorylation in plants

The reversible phosphorylation of enzyme protein is a very efficient mechanism of regulation which can yield an enhanced sensitivity of response to a given level of effectors. In some cases, several enzymes in a single or related pathways can be controlled by phosphorylation, by the same or different kinases or phosphatases. Moreover, some enzymes can be phosphorylated at several sites, which may have different effects on activity. Thus phosphorylation can control single pathways or the coordination of pathways in response to different signals.

The phosphorylation state of an enzyme is determined by the balance of kinase and phosphatase activity acting on the protein. The donor for phosphorylation is most often ATP, but can also be GTP, ADP or phosphoenolpyruvate which all have high phosphate group transfer potentials. Dephosphorylation is generally a simple hydrolysis reaction catalyzed by a

protein phosphatase, yielding orthophosphate (P_i). The steady state activity of the target enzyme is adjusted by positive and/or negative effectors influencing the rates of phosphorylation and dephosphorylation. A notable exception to this form of regulation by phosphorylation/dephosphorylation is the chloroplastic enzyme pyruvate, phosphate dikinase (PPDK) which is active when phosphorylated on a catalytic histidine residue and inactive when phosphorylated on a regulatory threonine residue (Roeske and Chollet, 1989; Smith et al., 1994). There is some evidence for lipid-activated kinases (Nanmori et al., 1994; Xing et al., 1996: Subramaniam et al., 1997) although no role has been assigned to them, and a chloroplast kinase responds to redox state in its regulation of the light-harvesting chlorophyll a/b-binding complex (Allen, 1995), but so far, Ca²⁺ is the best characterized second messenger that can affect protein phosphorylation in plants (for review see Hardie, 1999).

1.2.1 Plant protein kinases

It has been estimated that 2-3% of the eukaryotic genome codes for protein kinases. In response to a large and diverse number of signals, these enzymes can phosphorylate proteins on serine, threonine, tyrosine and histidine residues, usually to effect conformational changes in the substrate protein thus altering its properties. Almost all eukaryotic protein kinases contain a kinase catalytic domain of some 260 residues with alternating regions of high and low conservation (Hanks and Hunter, 1995). There are twelve conserved regions, referred to as subdomains, and within these subdomains are to be found some invariant residues which have been shown by crystal structure determinations to be important in catalysis and/or overall structure (for a review, see Wei et al., 1994). Plant protein kinases have been categorized into families based on the eukaryotic protein kinase phylogenetic classification scheme (see Hanks and Hunter, 1995 and Stone and Walker, 1995) and phylogenetic analysis of kinase domains from 89 A. thaliana protein kinase sequences (Hardie, 1999). This latter analysis revealed that most of the A. thaliana sequences clustered together into about 12 major subfamilies which could be used as a basis for discussing the stucture and function of all known plant protein kinases. Those subfamilies for which the most biochemical information is available and which are similar to the conventional protein kinases are outlined below.

The largest well-defined subfamily of the plant protein kinases is the calcium-dependent protein kinases (CDPKs). These are placed in the CaMK group by Hanks and Hunter (1995). However, these differ quite significantly from mammalian members of this family by virtue of

their apparent independence of the calcium-binding protein calmodulin. The DNA sequence of the first plant CDPK clone from soybean (Harmon et al., 1987), was found to encode a kinase catalytic domain followed by a C-terminal domain with 39% identity to calmodulin, with the four "EF-hand" Ca²⁺-binding motifs of calmodulin being conserved. Hence, these plant kinases are also refered to as calmodulin-like domain protein kinases. Several isoforms of CDPK have been found in many plant species, including *A. thaliana*, their existence potentially enabling both specificity and flexibility in the response to different Ca²⁺-elevating stimuli. CDPKs have a junction region between the two domains mentioned above, which is thought to function in a regulatory manner similar to the autoinhibitory region of the calmodulin-dependent protein kinases (CaMK), its effect being relieved upon binding of calcium to the protein. Reversible inactivation of nitrate reductase by phosphorylation has been demonstrated using a CDPK from spinach leaves (Douglas et al., 1998) and sucrose synthase can be phosphorylated by CDPK activity from maize leaf and soybean root nodules (Huber et al., 1996; Zhang and Chollet, 1997). However, there are a few members of this subfamily whose activity is either not independent of calmodulin or varies in their requirement for calcium.

The plant SNF1-like kinases are homologs of the *S. cerevisiae* SNF1 serine/threonine protein kinase required for carbon catabolite derepression. The SNF1/AMPK related protein kinase 1 (SnRK1) subgroup exists as large multimeric complexes regulated by phosphorylation by an upstream kinase kinase. A SnRK1 from spinach has recently been shown to be regulated by AMP in a manner similar to the mammalian SNF1 homologs (Sugden et al., 1999). SnRK1 kinases are also proposed to be activated by stresses like starvation for a carbon source and have similar *in vitro* substrate specificity, phosphorylating and thus inactivating the key regulatory enzymes of isoprenoid and sucrose synthesis and nitrogen assimilation, HMG-CoA reductase, sucrose synthase and nitrate reductase, respectively. No *in vivo* substrates have yet been confirmed for the SnRK1 kinases. Members of the other plant SNF1-like kinase subgroup, SnRK2, contain kinase domains related to those of Snf1 and AMPK, but the C-terminal domains are unrelated and the functions and biochemical properties of these kinases remain unknown.

;

ţ

The receptor-like kinase (RLK) subfamily share many structural features with the mammalian receptor-linked kinases (single transmembrane helix, C-terminal cytoplasmic kinase domain) but so far have no identifiable ligands. However, many interesting members belong to the group including the *Xa21* gene in rice responsible for resistance to some strains of bacterial

pathogen, the leucine-rich repeat (LRR) group which is thought to be involved in proteinprotein interactions and the *S* locus receptor-like kinase (SRK) involved in the genetically well-characterized phenomenon of self-incompatability in *Brassicaceae*. Autophosphorylation occurs on the kinase domains of most RLKs, reminiscent of the better characterized receptorlinked kinases whereby autophosphorylation facilitates the binding of signalling complexes to the kinase domain of the receptor thus initiating the intracellular transduction of a signal. 2

3

The next four subfamilies of plant kinases to be discussed are classed together in the CMGC group according to Hanks and Hunter (1995) and are the mitogen activated protein kinases, the cyclin-dependent kinases, glycogen synthase kinase-3 homologs and the casein kinase II family.

Mitogen-activated protein kinases (MAPK) are serine/threonine kinases that are activated by dual phosphorylation on threonine and tyrosine residues by MAPK kinase (MAPKK). The phosphorylation motif recognised by MAPKK is threonine-X-tyrosine in the activation loop of MAPK. MAPKK is an unusual type of kinase in that it phosphorylates serine, threonine and tyrosine residues. MAPKK is phosphorylated by MAPKK kinase (MAPKKK) and thus by forming these phosphorylation cascades, MAP kinase and associated kinases are able to link extracellular signals to intracellular events such as cell division, gene expression and control of growth, through phosphorylation of yet other molecules like transcription factors, Many plant MAPK homologs have been cloned and identified and have been shown to be activated by plant hormones and environmental stimuli (touch, cold shock, wounding, pathogen infection). The ATMPK1 and ATMPK2 homologs from A. thaliana were able to phosphorylate an artificial MAPK substrate (myelin basic protein, MBP), and this kinase activity was enhanced upon addition of purified Xenopus MAPKK. A putative MAPKK activity also accumulated in response to auxin treatment, stimulating phosphorylation of both the MAPK and its MBP kinase activity (Mizoguchi et al., 1984). Although the activity of MAPKs can be observed in various plant species and in response to various external signals, no plant MAPK cascade has yet been mapped out in any detail.

Cyclin-dependent kinases (CDK) form complexes with cyclins and/or inhibitory subunits to control progression through the cell cycle in eukaryotes. CDKs have a conserved amino acid motif, PSTAIRE, involved in the interaction with cyclin. Full activation of CDKs requires both binding of the appropriate cyclin and phosphorylation on a threonine residue in the activation loop. Phosphorylation of CDK on a tyrosine residue in the ATP-binding loop (P

Lincontration of reaching modelized storage form of glurose, GS adds — + 1152- attack a giveographic to convolve glycographic distribution, glycographic by the latter attack (-12) from glycologies, insular to glycologies spectross by the dolphilistic of glycologies specthold, Cisk-3 is involved in the multiside 2. Then of FPT phonprotesse activity oching on glycogen syndrosu loop) inhibits the kinase function in some mammalian systems. It would seem that an individual CDK may have multiple roles, determined by the cyclin to which it is bound and suggested by the greater number of cyclins than CDKs. Homologs of both cyclin and CDK have been found in many different plant species and direct evidence of CDK involvement in cell cycle regulation is emerging e.g. expression of a dominant negative mutant of *A. thaliana* Cdc2 kinase in tobacco plants disrupted the cell division *in vitro* of leaf mesophyll protoplasts derived from the transgenic plants (Hemerley et al., 1998). Mammalian glycogen synthase kinase-3 (GSK3) is implicated in hormonal responses, phosphorylating and inactivating glycogen synthase. It is part of a signal transduction pathway downstream of the insulin and insulin-like growth factor-1 receptors and is inactivated through phosphorylation by protein kinase B on a serine residue in response to stimulation of the pathway. The phosphorylation requirement of a specific tyrosine residue for maximal enzyme

19.00

2

pathway. The phosphorylation requirement of a specific tyrosine residue for maximal enzyme activity highlights in GSK3 both the importance of protein phosphorylation in the regulation of many signalling pathways and the compexity of regulating different kinase activites acting on the same substrate. Autophosphorylation on serine, threonine and tyrosine residues occurs with GSK3 homologs in *A. thaliana*. Several GSK3 homologs have been identified in plants, and together with mammalian GSK3 have good sequence similarity with the *shaggy* gene from *Drosophila melanogaster* which encodes a protein kinase involved in a pathway controlling development. Therefore, the different isoforms of the plant GSK3/shaggy homologs which show tissue-specific expression patterns have been assumed to contribute to control of plant development although little evidence exists to confirm this.

Casein kinase I and II homologs have been identified in plants. Similar to the mammalian enzymes, both CKI and CKII phosphorylate serine or threonine residues in the context of nearby acidic side chains and phosphorylate many different substrates *in vitro*. Although an *in vivo* function has yet to be characterized for the plant enzymes, and they appear to lack regulation *in vivo*, *Arabidopsis* plant extracts contain a CK2-like activity that affects the formation of a DNA-protein complex containing the transcription factor CCA1 (circadian clock-associated 1). *In vitro* studies showed that CCA1 is able to specifically interact with different submits of CK2 and that recombinant CK2 can phosphorylate CCA1 (Sugano et al., 1998).

Thus, it is apparent that many plant protein kinases are found ubiquitously in other eukaryotes (e.g. SNF1, MAPK and CDK) whilst other kinases are conspicuously absent from plants, such as the cyclic nucleotide-dependent protein kinases and conventional protein tyrosine

kinases. Conversely, higher plants have unique protein kinases distinct from those found in most eukaryotes (e.g. RLKs and CDPKs). Intriguingly, in plants, the protein kinases implicated in the earlier stages of signaling pathways are unique but the different signals often then converge into pathways that use more highly conserved protein kinases that are universal in eukaryotes. In consideration of the many environmental and developmental signals to which plants must respond, it is perhaps understandable why these differences in the early steps of signal transduction exist.

Although somewhat lagging behind mammalian protein kinase research, the knowledge of plant protein kinases has increased dramatically in the last decade. However, much of this knowledge is limited to the amino acid sequence of several hundred plant protein kinases available on database, the result of much cloning by sequence homology. Other molecular approaches like the yeast two-hybrid screen and interactional cloning, complemented by biochemical analysis, are required to learn more about the function and regulation of individual kinases.

and the second second by the second sec

.....

Ì

the state of the state of

•

24

477 -

1.2.2 Plant protein phosphatases

Protein phosphatase activity has been detected in different subcellular compartments of plants (Smith and Walker, 1996). They have been classed into three major groups which are biochemically almost indistinguishable from the mammalian type-1, -2A and -2C protein/serine phosphatases, classed according to their specificities for the substrate phosphorylase kinase (MacKintosh and Cohen, 1989). Some less well characterized plant protein phosphatases may represent novel enzymes that are unique to plants.

Mammalian and fungal PP1 enzyme is a complex of a catalytic subunit and one or more regulatory subunits thought to define specific functions of the catalytic activity of PP1 *in vivo* by controlling the subcellular location and substrate specificity of the enzyme complex. An interesting mammalian example is the regulatory subunit J-2 which is phosphorylated by GSK-3 in the presence of ATP-Mg and is hypothesized to then act as a chaperone that binds and activates newly synthesized PP1 catalytic subunits (Smith and Walker, 1996). At present, very little is known about the structure, function or physiological substrates of PP1 activity in plants, however similarities in biochemical properties and subcellular distribution with animal PP1 may give some hints as to what identification and characterization of PP1 regulatory subunits will reveal.

The PP2 subgroups are based on differences in subunit composition, divalent cation requirements and substrate specificites. PP2A is a heterotrimer of a catalytic subunit and two distinct regulatory subunits and does not require divalent cations for activity. PP2B, which has not been identified in plants, is a heterodimer with a catalytic subunit and a regulatory subunit which contains an EF-hand motif able to bind calcium for its activation. PP2C is a monomer requiring magnesium for activity.

÷

į

÷

Weiner and Weiner (1997) suggested that plants have more genes for the regulatory and catalytic subunits of PP2A than animals, thus implying that PP2A-type enzymes have some distinct regulatory properties and functions in plants. They demonstrated that the PP2A activity which activates sucrose phosphate synthase (SPS) in spinach leaves after illumination, is regulated by an interconversion between the trimeric (high activity) and dimeric (low activity) forms of the enzyme. Although the exact signal for this interesting mode of regulation was unknown, protein synthesis was shown to be involved in the trimer formation which it is proposed occurs in response to perceived changes in the metabolic phosphate and sugar status in the cytoplasm. PP2A is also the major enzyme responsible for inactivating the cytosolic enzyme quinate dehydrogenase (QDH) (MacKintosh et al., 1991) and is responsible for the dephosphorylation of PEPc, SPS and NR, three key metabolic enzymes, discussed in section 1.8 in the context of coordination of carbon and nitrogen metabolism.

Use of protein phosphatase inhibitors has helped reveal the involvement of protein phosphatases in plant signal transduction. Some of these inhibitors are natural toxins such as cyclosporin A and the marine toxin okadaic acid which are readily taken up by animal and plant cells. Differences in the potency of inhibitors for protein phosphatases has been utilized in several studies. For example, Sheen (1993) used okadaic acid to investigate the role of phosphatase activity in the greening of etiolated maize leaves in response to light. 1 μ M okadaic acid blocked the transcription of light-induced genes from the C₄ *ppdk* (pyruvate phosphate dikinase) promoter and the *rbcS* ((Rubisco) promoter but was ineffective at 100 nM. Okadaic acid inhibits PP2A (I_{0.5}= 0.1-1.0 nM) more potently than PP1 (I_{0.5}=10-100 nM) (MacKintosh and MacKintosh, 1994). Thus, this evidence might suggest that PP1, and not PP2A, activity is necessary for the light-induction of *ppdk* and *rbcS* in etiolated maize leaves. Although physiological substrates for PP2C have not been identified in plants, its activity has been detected in carrot cells, pea leaves and wheat leaves (MacKintosh et al., 1991). An interesting example of a plant PP2C involved in plant signalling is kinase-associated protein



phosphatase (KAPP). Isolated from *A. thaliana* by interactional cloning, KAPP interacts with the autophosphorylated form of the catalytic domain of RLK5 and consists of an N-terminal signal anchor, a kinase interaction domain and a C-terminal PP2C domain (Stone et al., 1994). The structure of KAPP and its interaction with RLK5 suggests that KAPP may control early steps in the RLK5 signalling pathway which participates in self-incompatability, defense responses and plant development.

A paucity of knowledge regarding the physiological substrates and regulatory subunit role of plant protein phosphatases will need to be overcome in order to further knowledge of the biological processes that these enzymes help regulate. Use of inhibitors has so far, and will hopefully continue, to help in this task.

1.3 Phosphoenolpyruvate carboxylase in higher plants

1.3.1 The reaction catalyzed by PEPc

PEPc (EC 4.1.1.31) is found in the cell cytosol of all higher plants and catalyzes the carboxylation of PEP to OAA. The substrate is bicarbonate (HCO₃⁻) rather than CO₂, and a divalent cation is required for activity, Mg^{2+} fulfilling this role *in vivo* (Andreo et al., 1987). The reaction is very exergonic, having a ΔG value of -6 to -8 kcal/mol. Experimental evidence (Andreo et al., 1987 and references therein) indicates that the reaction mechanism occurs in two steps (Figure 1.1). The first step is reversible and consists of the formation of carboxyphosphate and the enolate of pyruvate from the substrates. The second step is thought to be irreversible (Chollet et al., 1996) and combines the CO₂ and enolate with the concomitant release of OAA and P_i. The OAA is then rapidly reduced into malate by NADP-malate dehydrogenase (MDH, EC. 1.1.1.37) or transaminated into aspartate by aspartate aminotransferase (AAT, EC 2.6.1.1).

1.3.2 The regulation of PEPc

The activity of plant PEPc is subject to allosteric control by a variety of positive and negative metabolite effectors, especially when assayed at sub-optimal pH values that approximate that of the cytosol. Malate is a strong inhibitor of PEPc and glucose 6-phosphate activates the activity of the enzyme (Doncaster et al., 1987; Wedding et al., 1990; O'Leary et al., 1982). The sensitivity of PEPc to control by these allosteric effectors is determined by the phosphorylation state of the PEPc protein. When assayed at pH 7.3 and 1 mM pyruvate,

Figure 1.1 The reaction catalyzed by PEPc

Mechanism of the carboxylation of PEP catalyzed by PEPc. The figure was modified from Vidal and Chollet (1997).





i, de

recombinant PEPc from *Sorghum* had an apparent K_i for malate of 0.17 mM and 1.2 mM, and a K_a for glucose 6-phosphate of 1.3 mM and 0.28 mM, for the dephosphorylated and phosphorylated forms of PEPc respectively (Duff et al., 1995). Therefore, although changes in the cytosolic levels of these opposing allosteric effectors and H⁺ likely contribute to the overall regulation of PEPc activity *in vivo*, research over the past decade has focused primarily on the reversible phosphorylation which fine tunes the response of the enzyme to allosteric control.

PEPc has been shown to be phosphorylated at the N-terminal on one or more serine residues in many different plant species (Budde and Chollet, 1986; Nimmo et al., 1986; Jiao and Chollet, 1988; Nimmo et al., 1987; Van Quy et al., 1991; Duff and Chollet, 1995; Zhang et al., 1995). Phosphorylation of the enzyme is catalyzed by a highly regulated protein kinase refered to as PEPc kinase (section 1.8) and dephosphorylation by a typical mammalian-type protein phosphatase type 2A (e.g.; Carter et al., 1990; Echevarria et al., 1990; Jiao et al., 1991; McNaughton et al., 1991; Jiao and Chollet, 1992). PEPc is "activated" in vivo when it is phosphorylated but this activation does not result in an increase in the V_{max} of the enzyme but manifests itself by a decrease in sensitivity to feedback inhibition by malate (Nimmo et al., 1984; Nimmo et al., 1987). The serine residue that becomes phosphorylated resides in a plant-invariant motif SIDAO that is absent in the bacterial and evanobacterial primary structures deduced to date. Thus, this reversible means of fine tuning the activity and allosteric properties of PEPc is apparently unique to the plant enzyme (Chollet et al., 1996). It has been proposed that dimer-tetramer interconversion (Wu and Wedding, 1985; Wu et al., 1990; Willeford and Wedding, 1992) and the redox state of certain critical cysteine residues (Iglesias and Andreo, 1984; Chardot and Wedding, 1992) may be involved in the regulation of PEPc activity of L-malate sensitivity but these are based wholly on in vitro observations and there has been no evidence in vivo to support the involvement of either aggregation state changes or the five plant-invariant cysteines in PEPc regulation.

1.3.3 The function of PEPc

The physiological functions of PEPc in higher plants are diverse. The most extensively studied function of the enzyme is the initial fixation of atmospheric CO_2 during CAM and C_4 photosynthesis described in sections 1.5 and 1.6. This important function of PEPc activity concentrates CO_2 at the stromal location of Rubisco activity and, in so doing, favours the

carboxylase activity of Rubisco over its oxygenase activity thus minimizing the loss of energy through photorespiration. Other specialized and more general functions of PEPc are discussed in section 1.7.1.

ĉ

. .

1

1.3.4 The structure of PEPc

The PEPc protein is thought to be a homotetramer with an approximately 110 kD subunit. Isoforms of the enzyme have been characterized from both photosynthetic and non-photosynthetic tissues of various plants (for a review see Lepiniec et al., 1994 and Toh et al., 1994). These isoforms can be distinguished by their chromatographic, immunological and kinetic properties. Consistent with the enzyme's functional diversity (section 1.3.3), small multigene families have been found which consist of at least three to four members coding for different forms of PEPc. For example, *Sorghum* has three genes for PEPc which encode for the C₄-photosynthetic isoform, the root form and the C₃- like housekeeping form (Lepiniec et al., 1993). PEPc gene families have also been found in plant species performing CAM and C₃ photosynthesis (Lepiniec et al., 1994; Newman et al., 1994).

PEPc amino acid sequences have been deduced from many different plant species. When aligned, these sequences have been found to contain many highly conserved residues and motifs which have been proposed to be involved in the active site and/or regulation of the enzyme (Lepiniec et al., 1994; Toh et al., 1994). Subsequent site-directed mutagenesis and chemical modification studies have confirmed the involvement of many of these identified sites in such roles. In general, the N-terminal half of the PEPc subunit contains the regulatory sites (Jiao and Chollet, 1990; Jiao et al., 1990; Terada et al., 1990; Toh et al., 1994), for example, the plant invariant phosphorylation motif, and the C-terminal half contains most of the residues implicated in the active site (Andreo et al., 1987; Jiao et al., 1990; Rajagopalan et al., 1994; Terada et al., 1992).

Site-directed mutagenesis studies of the phosphorylation motif of a recombinant sorghum C_4 -PEPc demonstrated that the introduction of negative charge into this PEPc domain, by changing the target serine residue to an aspartate (S8D), is sufficient to mimic the regulatory effects of phosphorylation (Wang et al., 1992). Although the S8D form of the enzyme could not be phosphorylated by the C_4 leaf kinase, it was malate-insensitive like the phosphorylated form of the enzyme *in planta* (Duff et al., 1995; Wang et al., 1992). However, when the serine residue was changed to a cysteine (S8C), the enzyme was neither able to be phosphorylated nor insensitive to inhibition by malate (Wang et al., 1992).

Site-directed mutagenesis studies have also suggested that histidine residues play important roles in the regulatory and catalytic functions of the enzyme e.g. the histidine residue of the amino acid motif VfTAHPT (no. 215-221, the residues noted with capital letters being absolutely conserved, Lepiniec et al., 1993) is thought to be essential for the second partial reaction of PEPc (Terada et al., 1992), whilst chemical modification studies have identified an invariant lysine residue in the highly conserved sequence QqVMvGYSDSgKDaG as essential for catalysis (Jiao et al., 1990). Cysteine residues have also been identified in chemical modification studies as important for the catalytic activity of PEPc but no invariant site occupied by cysteine residues has been found in PEPc sequence alignments. Therefore, cysteine residues in PEPcs may not be involved in catalysis although some may be located near the active sites.

The three-dimensional structure of PEPe from *E. colt* has recently been determined (Kai et al., 1999) and this, together with further mutagenesis studies of conserved sites in the PEPc sequence, should advance knowledge of the quaternary structure of plant PEPc and its relationship to the function of the enzyme.

Due to the sufficient number of PEPc amino acid sequences deduced, it has been possible to study the phylogeny of the different PEPc isoforms. The information resulting from these analyses has led to the proposal of a molecular evolutionary pathway of PEPc which seeks to explain the conservation of residues and motifs between the different isoforms. Considering the plants enzymes only, it has been inferred from these phylogenetic trees that all PEPc sequences diverged from a common ancestral gene (Chollet et al., 1996 and references therein). The PEPc sequences from C_4 monocots like *Sorghum* and maize can be grouped and distinguished from the sequences of the other isoforms in these species and various CAM and C_3 isoforms (Lepiniec et al., 1994), whilst the sequence from the C_4 dicot *Flaveria trinervia* is more closely related to C_3 isoforms of the enzyme than to the C_4 monocot sequences. Further accumulation of PEPc sequence data and more elaborate computational analysis will be required before a clearer and more complete explanation of the origin of the different PEPc isoforms can be presented.

1.4 Photosynthesis and photorespiration

1.4.1 The C₃ pathway of photosyntheis

As mentioned in section 1.3.1, PEPc catalyzes the initial fixation of atmospheric CO_2 in C_4 photosynthesis and CAM, additional specialised photosynthetic pathways developed by certain species in adaptation of their environments. This function of PEPc and its regulation in this role will be discussed in more depth in following sections. However, it will first be useful to briefly recap on the reactions involved in photosynthesis by plants .

Photosynthesis is the process by which green plants use the sun's energy to build up carbohydrate reserves. This carbohydrate is then transported around the plant to tissues where it is either stored or broken down to release the energy it contains to fuel various reactions and processes essential for the plants' survival. It is also the means whereby plants acquire CO_2 from the air for the assimilation of carbon into the various molecules found in the plants. The fundamental pathway of photosynthesis is the same in all plants and is referred to as the C_3 pathway. It derives its name from the three-carbon compound 3-phosphoglycerate (3-PGA) that is produced, but is also known as the reductive pentose phosphate (RPP) or Calvin-Benson cycle (after two of its co-discoverers).

The stromal enzyme Rubisco, ribulose 1,5-bisphosphate carboxylase/oxygenase, catalyzes the fixation of atmospheric CO₂ and ribulose 1,5-bisphosphate into a short-lived six-carbon intermediate that is is broken down to yield two molecules of three-carbon 3-PGA. Three turns of the RPP cycle use three molecules of CO₂ to produce six glyceraldehyde 3-phosphate (G3P) molecules, of which five are used to regenerate ribulose 1,5-bisphosphate and the sixth results in the net synthesis of one molecule of the triosc phosphate, dihydoxyacetone-phosphate (DHAP). This DHAP is then converted into starch in the chloroplast or passes out of the chloroplast into the cytosol where it is used to synthesize sucrose. Thus, the overall pathway comprises three phases: carboxylation of ribulose 1,5-bisphosphate, reduction of 3-PGA using ATP and NADPH produced by the light reactions of photosynthesis, and regeneration of ribulose 1,5-bisphosphate. The individual reactions and the enzymes involved in the pathway are detailed in Figure 1.2.

1.4.2 Photorespiration

Photorespiration is the light-dependent evolution of CO_2 and O_2 uptake which occurs in green leaves of C_3 plants performing photosynthesis. It appears not to have an essential function in .
Figure 1.2 The reductive pentose phosphate pathway (RPP) cycle

A simplified diagram of the RPP cycle is outlined below, showing the carboxylation, reduction and regeneration phases and the synthesis of one net molecule of triose phosphate (DHAP) from three molecules of carbon dioxide. For clarity, various intermediate steps have been omitted in the regeneration phase as have the enzymes involved in the cycle. Abbreviations: RuBP, ribulose 1,5-bisphosphate; PGA, 3-phosphoglycerate; DiPGA, 1,3-bisphosphoglycerate; Ru5P, ribulose 5-phosphate; G3P, glyceraldehyde 3-phosphate. Figure is modified from Macdonald, FD and Buchanan, BB (1997).



the life of the plant but is a consequence of the oxygenase activity of Rubisco. CO₂ and O₂ are competitive substrates for the enzyme, their addition to ribulose 1,5-bisphosphate yielding two molecules of 3-PGA via the C₃ pathway, and a molecule of 3-PGA and a molecule of phosphoglycolate via the photosynthetic carbon oxidation (PCO) cycle, respectively (Figure 1.3). The PCO cycle involves the coordinated activity of reactions in the chloroplast, peroxisome and mitochondrion. For every four carbons (two molecules) of phosphoglycolate metabolized in the PCO cycle, one carbon is released as CO₂ from the decarboxylation of glycine by glycine dehydrogenase in the mitochondrion. This reaction also produces ammonia (NH_3) which is reassimilated by glutamine synthetase (GS) followed by glutamate synthese in chloroplast. Glutamate synthase is the also known as glutamine:oxoglutarate aminotransferase (GOGAT) and the system is refered to as the GS/GOGAT system. The other three carbons remaining from the original two molecules of phosphoglycolate are returned to the PCR cycle in the chloroplast as 3-PGA.

Increases in temperature increase the ratio of photorespiration to photosynthesis in C_3 leaves by favouring the oxygenation of ribulose 1,5-bisphosphate by Rubisco and increasing the solubility of O_2 and decreasing the solubility of CO_2 . This means that in C_3 plants, the CO_2 compensation point (the CO_2 concentration at which release of CO_2 by photoresiration is equal to uptake of CO_2 by photosynthesis) increases with temperature so that the loss of CO_2 becomes greater than its assimilation. Therefore, certain plants which grow in hot climates have developed supplementary pathways for fixing atmospheric CO_2 thus enabling them to concentrate CO_2 at the site of Rubisco and minimize photorespiration. These pathways are known as Crassulacean Acid Metabolism (CAM) and the C_4 or Hatch-Slack pathway and the enzyme PEPc is central to their specialized biochemistry.

1.5 PEPc and Crassulacean Acid Metabolism

1.5.1 The role of PEPc in CAM

Crassulacean Acid Metabolism is performed in addition to C_3 photosynthesis in plants that live in arid environments. Three features are central to CAM physiology. Firstly, the stomata open at night for gas exchange thus minimizing water loss through evaporation by keeping the stomata closed during the day. Secondly, atmospheric CO_2 is fixed by PEPc when the stomata are open i.e. at night, and stored as malic acid in the cell vacuole. Thirdly, the malic acid diffuses out of the vacuole the following day and is decarboxylated by malic enzyme or

Figure 1.3 Photosynthetic carbon oxidation (PCO) cycle

Abbreviations: Fd(red or ox), ferredoxin (reduced or oxidized); glu, glutamate; 2-OG, 2-oxoglutarate; 2-PG, 2-phosphoglycolate; PGA, 3-phosphoglycerate; RPP, reductive pentose phosphate; RuBP, ribulose 1,5-bisphosphate; THF, tetrahydrofolic acid. The numbers represent the enzymes which catalyze each step and are as follows:

- 1. Rubisco (ribulose 1,5-bisphosphate carboxylase/oxygenase)
- 2. phosphoglycolate phosphatase
- 3. glycolate oxidase
- 4. catalase
- 5. serine:glyoxylate aminotransferase
- 6. glutamate:glyoxylate aminotransferase
- 7. glycine decarboxylase
- 8. serine hydroxymethyltransferase
- 9. hydroxypyruvate reductase
- 10. glycerate kinase
- 11. glutamine synthetase
- 12. glutamate synthase

Figure modified from Canvin, DT and Salon, C (1997).



PEP carboxykinase (PEPCK), depending on the type of CAM plant. The CO₂ released is then fixed by Rubisco in the chloroplast in the normal C₃ photosynthetic pathway. Therefore, the primary and secondary CO₂ fixation reactions are temporally compartmented, fixation by PEPc occuring at night and fixation by Rubisco occuring during the day respectively. This compartmentation is represented diagrammatically in Figure 1.4. Therefore, CAM concentrates CO₂ in the leaf to a level sufficient to inhibit the oxygenase activity of Rubisco and minimize the loss of energy and assimilation products (gln) that occurs as a consequence of photorespiration. CO₂ fixation during CAM is characterized by four distinct phases: 9 E.

1) a marked increase in CO_2 assimilation catalyzed by PEPc during the night which begins to decline towards the end of the night due to feedback inhibition of PEPc activity by the high levels of malate being accumulated

2) a brief burst of CO_2 fixation into both C_4 and C_3 compounds as illumination activates Rubisco whilst PEPc is still active

3) net CO₂ uptake reaches a minimum during the day as stomata are closed, vacuolar malic acid is decarboxylated and the CO₂ released is refixed by Rubisco

4) atmospheric CO₂ uptake then begins to increase due to the activity of Rubisco at the end of the day as stomata open, and PEPc activity may also be contributing towards the end of this phase. As the day ends Rubisco activity ceases and PEPc activity continues to fix atmospheric CO_2 as the cycle returns to phase 1 at the onset of night.

An interesting feature of CAM is the rhythm of CO₂ assimilation which has been studied extensively in the CAM plant *Kalanchoë (Bryophyllum) fedtschenkoi*. Leaves detached from plants at the end of a day period and kept in continuous darkness and a stream of CO₂-free air at 15°C, exhibited a circadian rhythm of CO₂ assimilation for 3-4 days (Wilkins, 1959; Wilkins, 1962). The rhythm is characterized by peaks in CO₂ output attributed to inactivity of PEPc and consequent loss of respired CO₂, and troughs attributed to PEPc refixing respired CO₂, the only CO₂ available. PEPc activity is inhibited by the accumulation of malate that it synthesizes (peaks) until the malate is pumped into the vacuole and PEPc activity is resumed (trough). The plateau in CO₂ output at the end of 3 or 4 days has been explained by a vacuolar maximum capacity for malate (Wilkins, 1984) which when reached results in accumulation of newly synthesized malate in the cytoplasm and inhibition of any further fixation of respired CO₂ by PEPc. An alternative explanation is that the rhythm dies out because the energy reserves of the leaves have been used up (Nimmo et al., 1987). The

Figure 1.4 Compartmentation of CO₂ fixation reactions in C₄ and CAM species

The diagram below shows the spatial compartmentation and temporal compartmentation of the CO2 fixation reactions catalyzed by PEPc and Rubisco. The C_4 cycle is indicated by red lines (-) and the Calvin cycle is indicated by blue lines (-). Vacuoles are denoted by a V. Figure modified from Leegood et al. (1997).





rhythm can be reinitiated however, by a 4 hour exposure of the leaves to light which causes the malate to diffuse out of the vacuole into the cytoplasm where it is decarboxylated giving rise to a large peak in CO₂ output (Wilkins, 1992).

Detached *K. fedtschenkot* leaves kept in continuous light and a stream of normal air at any temperature, below 31° C, exhibit a circadian rhythm of CO₂ assimilation as opposed to output, which lasts for at least 10 days and which has a shorter time period between peak and trough (Anderson and Wilkins, 1989; Wilkins, 1984; Wilkins, 1992). The persistence of the rhythm can be explained by malate moving readily from the vacuole to the cytoplasm which prevents any accumulation of malate in the cytoplasm when the vacuole reaches its maximum malate capacity. Malate synthesized during the first period of PEPc activity inhibits the enzyme and then is decarboxylated sooner than in continuous darkness (hence the shorter rhythm) thus allowing PEPc to become active again repeating the cycle with a low leaf malate content. CO₂ released from decarboxylation of malate while PEPc is still inactive is either refixed by Rubisco or escapes from the leaf.

1.5.2 The regulation of PEPc in the CAM plant K. fedtschenkoi

Early work on CAM PEPc suggested that it was regulated simply by changes in cytosolic malate concentration. However, further studies indicated that the regulation was more complex. For example, it was reported that PEPc from the CAM species *Mesembryanthemum crystallinum* is more sensitive to inhibition by malate (i.e. less active under physiological conditions) during the day than at night (Winter, 1982). Biochemical investigation of PEPc in *K. fedtschenkoi* revealed that the enzyme was subject to a striking diel regulation *in vivo* that paralleled the changes in CO₂ assimilation. PEPc enzyme assayed in desalted extracts of *K. fedtschenkoi* leaves taken from the middle of a night period had an apparent K_i for L-malate of 3.0 mM and PEPc assayed from desalted day extracts has an apparent K_i for L-malate of 0.3 mM (Nimmo et al., 1984). Thus, a significant change in the sensitivity of PEPc to L-malate and consequently the activity of PEPc, was demonstrated under near-physiological assay conditions but without accompanying changes in V_{max} or PEPc amount (Nimmo et al., 1984). Inactivation of the enzyme during the day would avoid the futile cycling of CO₂ fixation by PEPc.

Immunological confirmation that the amount of PEPc protein did not change throughout the period studied (Nimmo et al., 1986) suggested that PEPc was regulated by some form of

post-translational modification. ³²P_i was fed to detached leaves for 72 hours and then extracts from the middle of the day and night were made and the PEPc immunoprecipitated. The day and night PEPc samples were then analyzed for incorporation of the radioactivity by SDS-PAGE and autoradiography. Only PEPc from the night extract contained ³²P_i (Ninmo et al., 1984) and two-dimensional thin-layer electrophoresis revealed that the phosphate group was bound to a serine residue (Nimmo et al., 1986). As suggested by these results, purification of the day and night forms of *K. fedtschenkoi* PEPc and the ability to alter the apparent K_i for Lmalate of the night form to that of the day form by removal of the phosphate group by alkaline phosphatase *in vitro* (Nimmo et al., 1986), confirmed that the diel variation in the malate sensitivity of PEPc was regulated by post-translational phosphorylation of the PEPc protein.

This result led to two further developments in the understanding of the regulation of PEPc in *K. fedtschenkoi*. First, Nimmo et al. (1987) investigated the malate sensitivity of PEPc extracted from leaves kept in continuous darkness and CO₂-free air at 15°C and found the PEPc to exhibit a persistent circadian rhythm of interconversion between the malate sensitive and malate insensitive forms. This rhythm paralleled that of CO₂ output previously observed by Wilkins (1959; 1962). However, the rhythm was not observed in leaves kept in continuous darkness and normal air at 15°C (Nimmo et al., 1987). Second, a protein phosphatase type-2A (PP2A) activity was identified in *K. fedtschenkoi* and found to be capable of dephosphorylating the phosphorylated "night" form of PEPc (Carter et al., 1990). However, no diurnal variation occurred in the specific activity of the PP2A over a 24 hour period (Carter et al., 1991).

Ŷ

Thus attention began to focus on a kinase activity in *K. fedtschenkoi* leaves capable of phosphorylating PEPc as the major regulatory component of PEPc malate sensitivity. A time course of this PEPc kinase activity showed that its presence at a high level throughout the middle of a 16 hour night period period and at a virtually undetectable level throughout the 8 hour day period correlated with the high and low apparent K_i values of PEPc for L-malate, respectively (Carter et al., 1991). The kinase activity appeared 4-5 hours after the beginning of the night period and disappeared 2-3 hours before the end of the night period, exactly in parallel with the changes in the malate sensitivity of PEPc. Moreover, kinase activity is responsible for the diurnal and circadian rhythm of PEPc phosphorylation. Treatment of

detached *K. fedtschenkoi* leaves with the protein synthesis inhibitors cycloheximide and puromycin blocked the nocturnal appearance of kinase activity. They also blocked both the circadian rhythm in CO₂ output and the interconversion of PEPc between the phosphorylated malate-insensitive and dephosphorylated malate-sensitive forms observed under constant darkness and CO₂-free air (Carter et al., 1991). This suggested that PEPc kinase is regulated directly by protein synthesis, controlled by a circadian oscillator. An alternative explanation was that the inhibitors were preventing the synthesis of another protein that activated PEPc kinase. Further studies using the RNA synthesis inhibitors rifampicin and actinomycin D prevented the appearance of PEPc kinase activity in the middle of the night period following the inhibitor treatment (Nimmo et al., 1993) and the malate sensitivity of PEPc in these night extracts was typical of the dephosphorylated form of the enzyme. This suggested that both protein synthesis and RNA synthesis were involved in the circadian control of PEPc activity by PEPc kinase in *K. fedtschenkoi*.

The development of a combined *in vitro* translation and kinase assay (Hartwell et al., 1996) allowed the quantification of translatable mRNA for the PEPe kinase protein. Total RNA was isolated from leaves taken at various points throughout the normal diurnal cycle of *K. fedtschenkoi* and translated *in vitro*. The translation products were then assayed for PEPe kinase activity. Extracts from leaves taken at the same points were also assayed directly for kinase activity and the apparent K_i of PEPe for L-malate. All three sets of data are shown together in Figure 1.5. The translatable PEPe kinase mRNA, PEPe kinase activity and apparent K_i of PEPe for L-malate exhibit the same diurnal pattern. This excellent correlation between the three makes it unlikely that the effect of protein synthesis inhibitors is to prevent synthesis of a protein that activated the kinase (Carter et al., 1991).

This apparent regulation of PEPc kinase by protein synthesis was an important and exciting finding as protein kinases in animal systems are typically regulated by a second messenger, a metabolite or a phosphorylation cascade (section 1.2.1) without net synthesis/degradation of the kinase. The changes in the level of translatable PEPc kinase mRNA may be due to changes in transcription or mRNA turnover. Also worth considering is the possibility that PEPc kinase synthesis is controlled by variations in the translatability of the kinase mRNA, rather than its amount. However, the nature of the assay for kinase translatable mRNA makes this unlikely but not impossible. Hartwell et al. (1996) also demonstrated that the appearance of kinase translatable mRNA can be prevented by not only the RNA synthesis inhibitor

Figure 1.5 Diurnal variation in PEPc kinase protein, translatable mRNA and PEPc K_i for L-malate.

The data from the experiment are shown in the graph opposite: % PEPc kinase protein (\blacklozenge); % PEPc kinase translatable mRNA (\blacksquare); PEPc K_i for L-malate (\blacktriangle). Figure taken from Hartwell et al. (1996).







K_i (mM) L-malate

time of day

25

cordycepin, but by treatment with the protein synthesis inhibitor puromycin, thus implying that there is a requirement for protein synthesis at the level of appearance of the PEPc kinase translatable mRNA as well as at the translation of the mRNA into kinase protein (Carter et al., 1991).

The correlation between the rhythms of PEPc kinase translatable mRNA, kinase activity and apparent K_i was observed during all inhibitor treatments and incubation conditions already discussed (Nimmo et al., 1987; Carter et al., 1991). In addition, Carter et al. (1995) showed that the circadian disappearance of PEPc kinase activity could be accelerated by high temperature and delayed by low temperature. Temperature also affected the amount of translatable kinase mRNA (Hartweil et al., 1996), the effect paralleling that of the kinase activity (Carter et al., 1995). When applied during troughs of CO_2 output i.e. when PEPc kinase is active and PEPc is phosphorylated, high temperature was also shown capable of resetting the phase of the circadian rhythm of CO_2 metabolism in *K. fedtschenkoi* (Wilkins, 1992).

One final facet of the regulation of PEPc kinase in *K. fedtschenkoi* to be mentioned is its calcium-independence. Some plants contain a CDPK capable of phosphorylating PEPc with low activity, but the role of this is not clear (Ogawa et al., 1998; Nhiri et al., 1998) Neither the PEPc kinase assayed directly from *K. fedtschenkoi* leaf extracts (Carter et al., 1991) nor the *in vitro* translated activity of PEPc kinase require Ca^{2+} (Hartwell et al., 1996) and it is generally accepted that the kinase important for the regulation of PEPc is calcium-independent.

From the work described above, a picture is emerging of a tightly coordinated regulation of PEPc in *K. fedtschenkoi*, the principal features being circadian control of a PEPc kinase at the level of its translatable mRNA and activity. This present knowledge of the regulation of PEPc in *K. fedtschenkoi* is outlined in Table 1.1. The phosphorylation and circadian rhythms of PEPc activity have been observed in several other CAM species as has PEPc kinase activity (for a review see Nimmo, 1998).

1.6 PEPc and C₄ photosynthesis

1.6.1 The role of PEPc in C₄ photosynthesis

Many plants that grow in climates with high temperature and light intensity perform C_4 photosynthesis. Like CAM, C_4 photosynthesis serves to concentrate CO_2 at the stromal site

 Table 1.1
 Comparison of the regulation of PEPc in CAM, C₄ and C₃ species

The table below is a summary and comparison of the regulation of PEPc in the three photosynthetic classes, CAM, C₄ and C₃. The different components involved are listed on the left-hand column along with some of the relevant references. For brevity, not all the significant experimental data is given but a more complete coverage of research developments is given in the text of Chapter 1.

	CAM	C4	C,
malate inhibition (Doncaster and Leegood, 1987; Wedding et al., 1990)	yes	yes	yes
malate sensitivity altered by reversible phosphorylation by Ca ²⁺ - independent PEPc kinase (Carter et al., 1991; Jiao and Chollet, 1989; Duff and Chollet, 1995)	yes; under circadian control	yes; phosphorylation state increases in response to illumination	yes; phosphorylation state increases in response to illumination
<i>de novo</i> protein synthesis involved (Hartwell et al., 1996; Bakrim et al., 1992)	yes; two protein synthesis steps involved - production of PEPc kinase mRNA and translation of PEPc kinase protein	yes; one protein synthesis step involved in the production of PEPc kinase activity	yes; one protein synthesis step involved in the production of PEPc kinase activity

transcriptional regulation involved	yes; RNA synthesis inhibitors block	yes; conflicting data on the effect of	yes; RNA synthesis inhibitors prevent
(Cartcr ct al., 1991; Hartwell et al.,	circadian increase in kinase activity;	RNA synthesis inhibitors on the	increase in PEPc kinase activity;
1996; Nimmo et al., 1993; Gigliolo-	increase in translatable PEPc kinase	increase in kinase activity; increase in	increase in translatable PEPc kinase
Guivarc'h et al., 1996)	mRNA levels parallels that of kinase	translatable PEPc kinase mRNA levels	mRNA levels parallels that of kinase
	activity	parallels that of kinase activity	activity
Ca ²⁺ /calmodulin interaction	yes; e.g. Ca2+/calmodulin antagonist	yes; e.g. $Ca^{2+}/calmodulin$ antagonist	yes; addition of Ca ²⁺ inhibits PEPc
(Hartwell et al., 1996; Pierre et al.,	prevented increase in PEPc kinase	prevented increase in PEPc kinase	kinase activity but depletion increased
1992; Giglioli-Guivare'h et al., 1996;	translatable mRNA and kinase activity	translatable mRNA and kinase activity	Ki of PEPc for malate without
Smith et al., 1996)			increasing PEPc kinase activity
N-metabolism	٤	2	yes; nitrate enhances PEPc kinase
(Van Quy et al., 1991; Duff and			activity
Chollet, 1995)			
cytosolic pH	è	yes; weak bases increase cytosolic pH	L
(Pierre et al., 1992; Giglioli-Guivarc'h		and phosphorylation state of PEPc	
et al., 1996)			
photosynthetic electron transport	6	yes; pre-freatment with inhibitors	yes; pre-treatment with inhibitors
(McNaughton et al., 1991; Giglioli-		blocked light-induction of PEPc	blocked light-induction of PEPc
Guivarc'h et al., 1996; Duff and		kinase activity	kinase activity
Chollet, 1995)			

of Rubisco activity. In contrast to CAM, the two CO₂ fixation events catalyzed by PEPc and Rubisco are separated spatially as opposed to temporally. This spatial compartmentation is facilitated by the specialized "Kranz" leaf anatomy displayed by C4 plants (Figure 1.4). The CO₂ fixation by phosphoenolpyruvate carboxylase (PEPc) occurs in the mesophyll cells of the C_4 leaf. The cytosolic PEPc combines atmospheric CO_2 with phosphoenolpyruvate (PEP) to form the four-carbon compound oxaloacetate (OAA), hence the name C_4 photosynthesis. The OAA is converted into either malate by malate dehydrogenase (MDH) or aspartate by aspartate aminotransferase (AAT) and passes into the bundle sheath cells which form the inner ring of the characteristic Kranz anatomy. In these cells the malate and aspartate is decarboxylated, depending on the plant, by NADP- or NAD- malic enzyme or PEP carboxykinase to provide CO₂ for Rubisco. The three-carbon acid remaining (pyruvate or PEP) returns to the mesophyll cells where it is used to capture another molecule of atmospheric CO₂. The CO₂ released in the bundle sheath cells raises the local concentration of the molecule to a level that out-competes the oxygen. The Rubisco located in the stroma of the bundle sheath cell chloroplasts is thus able to carboxylate ribulose 1,5-bisphosphate. The C_4 pathway which concentrates CO_2 thus reducing oxygenation of ribulose 1.5-bisphosphate and subsequent production of phosphoglycolate, costs energy. However, at high light intensity this extra energy cost is not a serious disadvantage because energy availability is not limiting. It is therefore seen that the C_4 pathway, by means of concentrating CO_2 at the site of Rubisco activity, is able to minimize the wasteful process of photorespiration.

1.6.2 The regulation of PEPc in C₄ plants

PEPc activity in C₄ plants is regulated by metabolites as it is in CAM species (O'Leary et al., 1982; Doncaster and Leegood, 1987). Nimmo et al. (1987) showed changes in the phosphorylation state and kinetic properties of PEPc in C₄ maize leaf in response to light and dark. The apparent K₄ of PEPc for L-malate was found to be higher in extracts of leaves which had been illuminated than of leaves which had been kept in darkness. The conversion between the two forms of the enzyme took about 1 hour from light/dark transition. Phosphorylation of PEPc on one or more serine residues was shown by pre-labelling leaves with ³²P_i and the sample with the highest apparent K_i for malate also had the greatest ³²P incorporation. Phosphorylation of PEPc on a serine residue *in vitro* was then performed using

purified dark dephosphorylated PEPc from maize, a partially purified protein kinase from maize and $[\gamma^{-32}P]ATP$ (Jiao and Chollet, 1989). Phosphorylation of PEPc was accompanied by a decrease in its malate sensitivity thus establishing a direct correlation between the kinetic properties and phosphorylation of the enzyme. Illumination of maize leaves with white, blue or red light (600 μ E/m²/s) for 1 hour gave similar, approximately 2.5 fold, increases in the apparent K_i of PEPc for malate (McNaughton et al., 1991). This suggested that the activation of PEPc kinase was not controlled by phytochrome or the blue-light photoreceptor. The appearance of kinase activity correlated well with the timing of changes in the kinetic properties of PEPc observed on illumination of intact maize leaves but protein phosphatase type-2A activity from maize leaves was not affected by light. Thus, as with studies investigating the regulation of PEPc in CAM, the focus of regulatory control of C₄ PEPc was PEPc kinase.

Cycloheximide, a cytoplasmic protein synthesis inhibitor, prevented the *in vivo* light activation of PEPc kinase activity in the C₄ species maize, *Sorghum* and *Digitaria sanguinalis* (Jiao et al., 1991; Bakrim et al., 1992; Pierre et al., 1992). This finding implicated *de novo* protein synthesis of PEPc kinase or a secondary component in the reversible light activation of PEPc in C₄ plants by phosphorylation, as had been previously shown in CAM plants (section 1.5.2) and kinase assays of *in vitro* translated RNA from maize leaves showed that the PEPc kinase mRNA increased in response to light (Hartwell et al., 1996).

The ability of the photosynthesis inhibitor 3(3,4-dichlorophenyl)-1,1-dimethylurea (DCMU) to block the light-induced increase in the apparent K₄ of PEPc for malate in maize (McNaughton et al., 1991) was consistent with other studies which suggested that photosynthetic electron flow was involved, either directly or indirectly, in the mechanism of light-activation *in vivo* (Samaras et al., 1988; Jiao and Chollet, 1992). Detached maize leaves were treated with various photosynthesis inhibitors before being given a light or dark treatment. Both the light activation of PEPc kinase activity and the subsequent phosphorylation of PEPc were prevented by inhibiting photosystem I and photosystem II (with methylviologen and DCMU or isocil, repectively)-directed electron transport, and the Calvin Cycle (with DL-glyceraldehyde) (Jiao and Chollet, 1992). The same effect was observed in *Sorghum* and the use of the ATP synthesis inhibitor gramicidin demonstrated a requirement of ATP synthesis for the light induction of the C₄ PEPc kinase in *Sorghum* (Bakrim et al., 1992). These inhibitor studies led to the proposal that the Calvin Cycle

supplies the mesophyll cells with a putative signal (e.g. phosphorylated metabolite, amino acid) that interacts with the cytoplasmic protein synthesis event (Jiao et al., 1991) to effect the light activation of PEPc kinase and the concomitant phosphorylation of PEPc. The Calvin Cycle would also supply high levels of known positive effectors of PEPc e.g. triose phosphate and glucose 6-phosphate to the mesophyll cells. The effect of such regulation would be desensitization of PEPc in the mesophyll cells to the high levels of malate which are required to sustain rapid diffusion from the mesophyll to the bundle sheath cells during high rates of C_4 photosynthesis. Development of this idea by treating mesophyll cells from the C_4 species *Digitaria sanguinalis* with 3-phosphogiycerate or pyruvate in the dark led to significant increases in PEPc kinase activity and decreases in the malate sensitivity of PEPc, thus providing direct evidence that bundle sheath-derived C_4 pathway intermediates are involved in the activation of PEPc in the C_4 mesophyll cytosol (Duff et al., 1996).

Studies using Sorghum mesophyll cell protoplasts (Pierre et al., 1992) demonstrated that the malate sensitivity of PEPc was decreased by increasing the cytosolic pH using the weak bases NH4Cl and methylamine. This suggested that cytosolic alkalinization mediated the light induction of PEPc phosphorylation. Consistent with this hypothesis is the finding of Yin et al. (1990) that cytosolic alkalinization occurs during photosynthesis in C₃ plants. This was attributed to two components: protonated phosphoglycerate transferred into chloroplasts and pumping of protons into the vacuole. Studies using the D. sanguinalis mesophyll cell protoplasts also indicated an involvement for calcium in the light-transduction pathway. The in situ light-dependent stimulation of PEPc kinase activity and phosphorylation of PEPc was inhibited by incubation of protoplasts using the Ca2+-chelating agent EGTA and the Ca2+selective ionophore A23187 (Pierre et al., 1992). These results are internally consistent as cytosolic alkalinization would mobilize Ca2+ from intracellular stores, but other reports showed that PEPc kinase activity assayed from C4 species was calcium-independent (Jiao et al., 1989; Hartwell et al., 1996). These results therefore suggested that Ca²⁺ was involved at a step upstream of the PEPc kinase in the C₄ light signal transduction pathway. In contrast, the findings of another group were that the PEPc kinase from maize leaves is calcium-dependent (Ogawa et al., 1992). Two types of protein kinase, one requiring and one not requiring Ca²⁺ for activity, were then shown to phosphorylate a recombinant C4-type Sorghum PEPc at the specific serine residue (Li and Chollet, 1993). The calcium-independent kinase has been highly purified from maize leaves and its activity in vivo is light-dependent and sensitive to

inhibitors of photosynthesis and cytosolic protein synthesis (Li and Chollet, 1993; Wang and Chollet, 1993). These properties make the calcium-independent kinase the most physiologically likely candidate for regulating PEPc. Giglioli-Guivarc'h et al. (1996) built upon the previous work using D. sanguinalis protoplasts (Pierre et al., 1992) and proposed that a highly regulated cross-talk occurs between the two photosynthetic cell types of the C_4 leaf. Fluoresence imaging confirmed the involvement of cytosolic pH in the control of PEPc kinase activity and PEPc phosphorylation via light and calcium mobilization from vacuoles. The photosynthesis-related metabolite 3-PGA was able to mimic the effect of weak bases in contributing to cytosolic alkalinization suggesting that 3-PGA may be transported from the bundle-sheath cells to the mesophyll cells in an intact C4 leaf as part of the light-induction of PEPc kinase. The various inhibitor treatments used in previous studies were repeated and confirmed the involvement of electron transport (DCMU), ATP synthesis (gramicidin) and cytosolic protein synthesis (cycloheximide) in the C_4 PEPc signal transduction pathway (Jiao et al., 1991; Bakrim et al., 1992). However, the RNA polymerase II inhibitors α -amanitin and actinomycin D had no effect on the light plus weak base-dependent phosphorylation of PEPc in situ. They proposed that the light induction of PEPe kinase did not require RNA synthesis. This is in direct contrast to the findings of Hartwell et al. (1999) who showed a lightdependent increase in maize PEPc kinase mRNA which was blocked by RNA synthesis inhibitors. Calcium/calmodulin antagonists were also investigated and from the results obtained, Giglioi-Guivarc'h et al. (1996) hypothesized that the signalling pathway is perhaps multicyclic, involving an upstream calcium/calmodulin-dependent kinase and the calciumindependent, light-activated PEPc kinase (Figure 1.6).

1.7 PEPc and C₃ photosynthesis

1.7.1 The role of PEPc in C₃ plants

In the leaves of C₃ plants and non-photosynthetic tissues the role of PEPc is anaplerotic, topping up intermediates from the TCA cycle that are removed for biosynthesis, particularly amino acid biosynthesis and nitrogen assimilation (Melzer and O'Leary, 1987; Van Quy et al., 1991a). In general, this necessitates stimulation of PEPc in processes that require intensive plant cell division and organogenesis e.g. germination (Lepiniec et al., 1994).

Figure 1.6 C₄ PEPc transduction pathway

Abbreviations: PS, photosystem; $[Ca^{2+}]_c$, cytosolic calcium concentration. Figure modified from Giglioli-Guivarc'h et al. (1996).



33

There are also a variety of what are termed "housekeeping" functions in which PEPc activity is involved (Lepiniec et al., 1993). These include maintenance of pH, electroneutrality and ionic balance, recapture of respiratory CO_2 and transfer of reducing power (NADH).

Specialized functions of PEPc other than those described in section 1.5.1 and 1.6.1 include N_2 fixation in legume root nodules (Deroche and Carrayol, 1988; Schuller et al., 1990; Schuller and Werner, 1993) and an involvement in stomatal guard cell movements (Tarczynski and Outlaw, 1993; Zhang et al., 1994). In symbiotic dinitrogen fixation, the plant cytosolic PEPc provides both energy, in the form of C_4 acids, for bacteroid metabolism and carbon skeletons for ammonia assimilation.

1.7.2 The regulation of PEPc in C₃ plants

By comparison with the regulation of the C_4 and CAM isoforms of PEPc, relatively little is known about the regulation of the non-photosynthetic isoforms of PEPc. However, many aspects of PEPc regulation in C_4 and CAM species is common to C_3 species.

The light activation of PEPc activity in a C_3 species was first demonstrated in wheat (Van Quy et al., 1991). Comparison studies showed that the light activation of PEPc activity in C₃ species is only 30 % of that found in C₄ species (Rajagopalan et al., 1993). The K_i of PEPc for malate increased about 3-fold upon illumination in C_4 species, but increased only slightly in C_3 species (Gupta et al., 1994). The reversible phosphorylation of PEPc responsible for the change of malate sensitivity of PEPc in C_1 species was shown to affect the activity and allosteric properties of the enzyme to a much lesser extent than in C₄ species (Gupta et al., 1994). Consistent with these findings, the level of translatable PEPc kinase mRNA from the C_3 species barley increased only about two-fold in response to light compared with an approximately 7-fold light increase in C₄ maize (Hartwell et al., 1996). Interestingly, whilst Matsuoka and Hata (1987) reported increased levels of PEPc protein in C_3 rice in response to light, they also observed a reduction in levels of a second type of PEPc protein in response to the same light signal. Rajagopalan et al. (1993) proposed that the low level of light-induced aikalinization of cell cytoplasm in C_3 species may limit light activation of PEPc. They reported that the light-dependent alkalinization of cell sap is three times higher in C₄ plants than in C₃ plants, but information on changes in the cytosol of mesophyll cells in vivo cannot be derived from crude leaf extract.

The first increase in the phosphorylation state of PEPc from a C₃ species in response to light was demonstrated using ³²P labelling studies and okadaic acid and was enhanced by feeding the nitrogen-deficient leaves 40 mM nitrate before illumination (Van Quy et al., 1991). This treatment also altered the malate sensitivity of PEPc such that the $I_{0.5}$ (malate) was 1.45 mM, 1.7 mM and 3.4 mM for the enzyme from dark control, illuminated low nitrate and illuminated high nitrate leaves respectively. In contrast, Duff and Chollet (1995) reported little short-term effect of nitrate feeding on the activity and malate sensitivity of C₃ PEPc in either illuminated or darkened wheat leaves detached from low-N-grown plants. However, the lightactivation of PEPc kinase activity was enhanced by nitrate, or more specifically glutamine levels (Wang and Chollet, 1993; Duff and Chollet, 1995). Nitrate was suggested to modulate the activity of cytosolic protein kinases (PEPc kinase and SPS kinase) resulting in the diversion of photosynthetic carbon flow from sucrose synthesis to amino acid biosynthesis (Champigny and Foyer, 1992; Huber et al., 1994): the resupply of nitrate to N-deficient wheat leaves caused an increase in the production of amino acids and a short-term inhibition of sucrose synthesis (Champigny et al., 1992; Van Quy et al., 1991b). Treatment of leaves with mannose depletes the pool of inorganic phosphate through the production of mannose 6phosphate and thereby inhibits ATP synthesis. Thus by blocking kinase reactions, mannose was able to inhibit the NO3-activation of PEPc and inactivation of sucrose phosphate synthase (SPS) (Van Quy and Champigny, 1992). Treatment of leaves with the phosphatase inhibitor okadaic acid increased the PEPc activation and SPS inactivation by light and nitrate (Van Quy and Champigny, 1992) suggesting that as with CAM and C_4 species, the phosphorylation state of PEPc is controlled by the activity of PEPc kinase rather than phosphatase activity.

2

The feeding of detached N-depleted wheat leaves with various nitrogen-containing compounds and with nitrate plus various inhibitors of enzymes involved in nitrogen metabolism (NR, GS, GOGAT, AAT- see page x for abbreviations) led to the proposal that glutamine and not nitrate was the most likely effector controlling the nitrogen-dependent PEPc activation, via the modulation of PEPc kinase and PEPc gene expression by the glutamine:glutamate ratio (Manh et al., 1993). A similar conclusion was drawn from inhibition of GS in barley leaves using phosphoinothricine (Diaz et al., 1996; Diaz et al., 1995). Specific inhibitors of glutamine synthetase have also been shown to block the light activation of tobacco leaf PEPc kinase under both photorespiratory and non-photorespiratory

conditions (Li et al., 1996). This striking effect is partially and specifically reversed by exogenous glutamine. Doubt was later cast as to the regulatory role of glutamine proposed by Manh et al.(1993) when no direct effect of glutamine was observed on the *in vitro* activity of the wheat leaf PEPc kinase activity assayed using a $[\gamma^{-32}P]$ ATP-based *in vitro* kinase assay (Duff and Chollet, 1995). However, the wheat PEPc kinase activity assayed by Duff and Chollet (1995) was calcium-independent (Wang and Chollet, 1993; Li et al., 1996) whereas the PEPc kinase activity assay used by Manh et al. (1993) had not included the Ca²⁺-chelating agent EGTA and so it is possible that the glutamine-enhanced PEPc kinase activity is different from the calcium-independent PEPc kinase activity detected by others. It is equally possible that glutamine might facilitate expression of the calcium-independent PEPc kinase.

Protein synthesis inhibitors and photosynthesis inhibitors were able to prevent the increase in PEPc kinase activity in response to light plus nitrate and light alone in C_3 species (Duff and Chollet, 1995; Li et al., 1996; Smith et al., 1996), similar to the effects of these compounds observed in C_4 illuminated plants without nitrate feeding (Jiao et al., 1991; Bakrim et al., 1992; Pierre et al., 1992; McNaughton et al., 1991; Jiao and Chollet, 1992). Studies using barley mesophyll protoplasts have confirmed the presence of a light-activated, cycloheximide-sensitive kinase but also indicated the presence of a constitutive, cycloheximide-insensitive kinase (Smith et al., 1996).

In contrast to the general trend in results indicating the central importance of phosphorylation in the regulation of PEPc, a few studies using barley and pea tissues have raised the possibility that metabolite regulation of C_3 -leaf PEPc is more important than post-translational regulation. Kromer et al. (1996a; 1996b) demonstrated the light-activation of barley leaf PEPc measured as an increase in phosphorylation state and a decrease in malate sensitivity but showed that physiological concentrations of glucose 6-phosphate were able to overcome the malate inhibition. They then proposed that feedback inhibition by glutamate and aspartate was perhaps more physiologically significant because these metabolites were to be found at higher concentrations *in vivo* than malate. Leport et al. (1996) then used protein kinase and phosphatase inhibitors to show that PEPc from pea leaf and root is controlled allosterically by changes in the relative concentrations of PEP, malate and glucose 6-phosphate and not by phosphorylation. However, the validity of any such interpretation is questionable because no attempts were made to determine the effect of light or nitrate on the *in vivo* phosphorylation status of PEPc or PEPc kinase activity. Studies using barley leaf mesophyll protoplasts can be seen as the C3 PEPc equivalent to the D. sanguinalis protoplasts studies of C_4 PEPc carried out by Giglioli-Guivarc'h et al. (1996) which brought together and confirmed much of the previous data in an attempt to construct a model of the C₄ PEPc signal transduction pathway. Smith et al. (1996) carried out a series of experiments using barley leaf protoplasts and different inhibitors and chemicals used previously in other studies and were able to piece together the following biochemical scenario of PEPc regulation in the C₃ species. Two different PEPc kinase activities are present in the barley protoplasts: one is a basal activity, slowly turned over and present in the dark and the other is light-activated and rapidly turned over, similar to the PEPc kinase activity described in other C_3 species. The light-activation of PEPc kinase activity occurred in the presence or absence of nitrogen and effected the typical changes in allosteric properties of PEPc in response to activation e.g. decrease in malate sensitivity. The light-activated kinase activity was not prevented by treatment with DCMU (although light-activation of nitrate reductase in the same system was affected (Lillo et al., 1996), indicating that the light-activation of barley leaf protoplast PEPc kinase does not require photosynthesis. Reduction in the intracellular calcium concentration using EGTA and the calcium ionophore A23187 enhanced the lightmediated reduction in the malate sensitivity of PEPc but did not affect PEPc kinase activity assayed in vitro.

÷,

÷

Reversible phosphorylation of various other nonphotosynthetic isoforms of PEPc have been reported. Schuller et al. (1990) found PEPc from root nodules to be regulated by various metabolites including malate, glucose 6-phosphate and aspartate and *in vitro* studies showed the presence of PEPc kinase activity in soybean root nodules which is capable of effecting a change in the malate sensitivity of PEPc by phosphorylating it (Schuller and Werner, 1993). Studies with ³²P_i demonstrated the reversible phosphorylation of PEPc in soybean root nodules *in vivo* (Zhang et al., 1995) and complementary changes in PEPc activity and/or malate sensitivity under near-physiological assay conditions (i.e. low pFI, low [PEP] relative to K_m). It would seem that translocation of sucrose from the leaves to the nodule is the signal regulating PEPc activity. In the light, when the supply of photosynthate from the phloem is available to the nodule, PEPc kinase activity is bigh and the malate sensitivity of PEPc is low. If this supply of photosynthate is disrupted by exposure to darkness, decapitation (shoot removal) or stem girdling, PEPc kinase activity decreases and the PEPc is sensitive to malate inhibition (Wadham et al., 1996; Zhang and Chollet, 1997; Zhang et al., 1995). PEPc kinase

has been partially purified from soybean root nodules and is calcium-independent (Zhang and Chollet., 1997).

PEPc is a key enzyme in stomatal movements in most plants by providing malate (malate²) which is accumulated by the stomatal guard cells to maintain intracellular pH and ion balance during the proton extrusion and K⁻ uptake associated with the changes in stomatal aperture (Du et al., 1997). PEPc from Vicia faba guard cells microdissected from stomata was significantly inhibited by 400 µM malate in the dark but not during opening (Zhang et al., 1994). Further investigation of guard cell PEPc regulation demonstrated that this change in the malate sensitivity of PEPc is correlated with an increase in PEPc phosphorylation state which can be promoted by guard cell opening resulting from treatment with fusicoccin, or suppressed by guard cell closure resulting from treatment with abscisic acid (Du et al., 1997). Osuna et al. (1996) demonstrated that the components required for the covalent regulation of PEPc are present in wheat seeds and that phosphorylation occurs in vivo during germination. There was a significant (50%) increase in PEPc specific activity and indication of the enzyme's stability during the germinative and post-germinative periods in wheat. Antiphosphorylation site antibodies which cross-reacted with both the phosphorylated and nonphosphorylated forms of the enzyme were shown to enhance PEPc activity and cause an increase in the IC_{50} for L-malate. They also facilitated the identification of phosphorylatable seed PEPc subunits by specifically inhibiting their phosphorylation in vitro. A partial inhibitory effect of EGTA on PEPc phosphorylation in desalted crude wheat seed extracts was observed suggestive of calcium-dependency.

Various *in vitro* studies have established that PEPc kinase activity is present in many different C_3 tissues (Duff and Chollet, 1995; Wang and Chollet, 1993; Pacquit et al., 1993), and have demonstrated this kinase's similarity to the C_4 and CAM enzymes with respect to its Ca^{2+} independency, chromatographic properties, and catalytic subunit(s). However, the results of many studies have also begun to reveal the existence of differentiated light-signal transduction pathways in illuminated C_3 and C_4 leaves which could be beneficial for the post-translational regulation of PEPc to fulfil the distinct anaplerotic and photosynthetic functions of the respective isoforms. The present knowledge of PEPc regulation in C_3 species is summarized and compared with that of CAM and C_4 isoforms of PEPc in Table 1.1.

1.8 Molecular properties of phosphoenolpyruvate carboxylase (PEPc) kinase

The regulation of PEPc kinase activity has been described in the previous sections. The molecular properties of the enzyme will now be described.

McNaughton et al. (1991) reported that PEPc kinase activity from maize leaves was inhibited by malate but that Ca^{2+} , calmodulin, EGTA and trifluoperazine had no significant effect on the activity of PEPc kinase, nor did PEP, glucose 6-phosphate, glycerol 3-phosphate and fructose 2,6-bisphosphate. Another study reported inhibition by glucose 6-phosphate in addition to malate (Jiao and Chollet, 1991).

The extremely low abundance PEPc kinase has been partially purified from a few species (Jiao et al., 1989; Carter et al., 1991; Wang et al., 1993). Maize PEPc kinase was purified about 4000-fold and, as isolated, catalyzes neither autophosphorylation nor the phosphorylation of heterologous substrates e.g. casein (Jiao and Chollet, 1989; Li and Chollet, 1993; Li and Chollet, 1994; Wang and Chollet, 1993). In contrast, all plant PEPc isoforms examined serve as substrates for the kinase *in vitro* (Li and Chollet, 1994; Wang and Chollet, 1993; Wang et al., 1992) and although some preference for the corresponding PEPc kinase has been reported (Li, B, Zhang, XG and Chollet, R, unpublished data quoted in Chollet et al., 1996), *K fedtschenkoi* PEPc has been used as a substrate for the PEPc kinases of *K. fedtschenkoi*, maize, barley and sugarcane (Nimmo, HG, personal communication).

SDS-PAGE and a subsequent *in situ* renaturation method was used to identify two Ca²⁺independent protein kinase polypeptides (~30/32 kDa and ~37/39 kDa) capable of phosphorylating PEPc from C₄ maize, C₃ tobacco and the CAM species *Mesembryanthemum crystallinum in vitro* exclusively at the target serine residue (Li and Choflet, 1993; Li et al., 1996; Li and Choflet, 1994). One Ca²⁺-dependent (~57 kDa) protein kinase polypeptide capable of phosphorylating maize PEPc was also identified (Li and Choflet, 1993). Only the Ca²⁺-independent, 30- to 39-kD PEPc kinase has been shown to be light-dark (C₄) or daynight (CAM) regulated *in vivo* (Li and Choflet, 1993; Li and Choflet, 1994).

1.9 The coordinate regulation of carbon and nitrogen metabolism in plants by protein phosphorylation

As mentioned previously (section 1.7.2), nitrate has been proposed to regulate the activity of cytosolic kinases thus controlling the flow of photosynthetic carbon between sucrose synthesis and amino acid biosynthesis (Champigny and Foyer, 1992) As PEPc kinase and its

errore indiana UCP-plurore - fruction 6-phonoisel - - + sucrove 6-phonophale + UDP suprove break down as sucrose - ITP = improvide + proctore substrate has already been discussed in this context, this section will focus on the other two enzymes refered to, nitrate reductase (NR) kinase and sucrose phosphate synthase (SPS) kinase, and their substrates.

2

1

Nitrate reductase (NR) catalyzes the reduction of nitrate to nitrite in the assimilation of nitrogen in plants. Sucrose phosphate synthase (SPS) is involved in sucrose synthesis and catalyzes the conversion of UDP-glucose and fructose 6-phosphate to sucrose 6-phosphate which is then rapidly dephosphorylated by sucrose synthase to form sucrose. Both enzymes are light-activated and regulated by reversible phosphorylation, phosphorylation causing their deactivation. Phosphorylation of SPS by SPS kinase occurs at multiple sites although phosphorylation of a specific serine residue is thought to be essential for deactivating the enzyme (Huber et al., 1995). Okadaic acid (a phosphatase inhibitor) has been shown to decrease SPS activity and mannose (which sequesters P_i) has been shown to prevent SPS deactivation (Van Quy et al., 1992). Phosphorylation of SPS increases its sensitivity to inhibition by P_i and decreases it sensitivity to activation by glucose 6-phosphate. Phosphorylation of NR is slightly more complex as it has no effect on the activity of NR but rather phosphorylation of a specific serine residue enables NR to interact with an inhibitor protein called NIP, with a conformational change in the N-terminal region of NR induced by phosphorylation suggested to be the mechanism involved (Douglas et al., 1995; Huber et al., 1995; MacKintosh et al., 1995). Purification of NIP revealed that it was a member of the 14-3-3 family of proteins (Moorhead et al., 1997) which have been shown to participate in many diverse signalling pathways. Their exact mode of action is unknown although they have been shown to function as adapter molecules bringing specific target proteins together or to particular locations, and as solubility factors and enzyme inhibitors (Ferl, 1996). No allosteric effectors are known for NR.

Three peaks of SPS and/or NR kinase activity have been identified after anion-exchange chromatography of spinach leaf extracts (McMichael et al., 1995). The first peak of kinase activity is capable of phosphorylating both SPS and NR and is itself regulated by phosphorylation (Pauline Douglas and Carol MacKintosh, unpublished, quoted in MacKintosh, 1997) and a divalent cation. The second peak of kinase activity is calcium dependent and acts only on NR and the third peak of kinase activity acts only on SPS. Like activity 1, activity 3 is regulated by phosphorylation and was demonstrated to be a CDPK (Douglas et al., 1997). SnRK1 kinase activity (plant homolog of SNF1) has been reported to

inactivate NR and SPS activity *in vitro* (Hardie et al., 1997), suggesting that NR kinase and SPS kinase are closely related members of the SNF1 related protein kinase family.

The "dual specificity" of kinase activity 1 (McMichael, 1995) which phosphorylates both NR and SPS would provide a mechanism for coordinately regulating NR and SPS, whilst kinase activity 2 and 3 enable NR and SPS to be regulated independently where their response to a particular signal is different. Huber et al. (1995) reported that nitrate enhances the progressive inactivation of SPS (phosphorylation) and activation of NR (dephosphorylation) in response to accumulation of the end-products of photosynthesis. This therefore would be an example of where kinase activity 3 would act alone.

Ì

į

3

When the effects of nitrate on PEPc, NR and SPS are considered together, the results are consistent with the hypothesis suggested by Champigny and Foyer (1992). The common requirements of nitrate and carbon dioxide metabolism in terms of reducing power, ATP and carbon skeletons also make the coordinate regulation of PEPc, NR and SPS quite logical.

A simplified scheme of the regulation of PEPc, NR and SPS in C_3 leaves is shown in Figure 1.7. In addition to the regulatory phosphorylation and allosteric effectors discussed already, several other factors known to be involved in the regulation of these enzymes are shown in the diagram thus indicating the complexity of the signalling pathways involved.

1.10 Project objectives

The preceding discussion has provided a brief overview of the function and regulation of PEPc in higher plants. A summary and comparison of the emerging PEPc regulatory pathways in the three photosynthetic classes is given in Table 1.1. The overall objective of this research was to investigate the physiological roles of the phosphorylation of PEPc and the signalling pathways that bring this about.

The regulation of PEPc in CAM and C_4 species have certain common features like the calcium-independency of the principal kinase activity phosphorylating PEPc and the involvement of *de novo* protein synthesis of PEPc kinase. However, a major component of the C_4 PEPc signal transduction pathway is cytosolic alkalinization. Little was known about the regulatory involvement of pH in the CAM PEPc signalling pathway so the objective of the first part of this research was to investigate the involvement of cytosolic pH in the regulation of PEPc in the CAM species *K.fedtschenkoi* (Figure 1.8).

Figure 1.7 Coordinate regulation of PEPc, SPS and NR in C₃ leaves

In the diagram opposite, the factors enclosed in blue shaded areas e.g. anoxia, NO₃⁻ override the photosynthesis signals to promote the activation or inactivation of the enzymes as indicated. Abbreviations: PK, protein kinase; PP, protein phosphatase; NIP, nitrate reductase inhibitor protein protein; C, carbon; N, nitrogen. Figure modified from MacKintosh, C (1997).





The objective of the second part of this research was to discover to what extent the mechanisms and signalling involved in regulating PEPc in CAM and C_4 species are the same for C_3 species. Plant tissue (Figure 1.9) and a photomixotrophic cell suspension culture of *Arabidopsis thaliana* (Figure 1.10) were used as experimental systems for this purpose.

ملي ا



Figure 1.10 Photograph of *Arabidopsis thaliana* photomixotrophic cell suspension culture



Chapter Two

MATERIALS AND METHODS

2.1 MATERIALS AND MATERIAL SOURCES

 $[\gamma^{-32}P]ATP$ (triethylammonium salt, 3000 Ci/mmol), $[\alpha^{-32}P]dCTP$ (3000 Ci/mmol), HybondTM-N nylon membrane, rabbit reticulocyte lysate *in vitro* translation kit, T7 Sequenase version 2.0 DNA sequencing kit, the Rediprime and Decaprime DNA labelling kits and Redivue^{TM 35}S-methionine (1000 Ci/mmol) were obtained from Amersham International, Bucks., U.K.

Antipain hydrochloride, benzamidine hydrochloride, Bis-Tris, bovine serum albumin (BSA), bromophenol blue, cantharidin, casein acid hydrolysate C-9386, casein enzymatic hydrolysate C-0626, chymostatin, Coomassie brilliant blue G250, diethyl pyrocarbonate (DEPC), formaldehyde, formamide, hexadecyltrimethylammonium bromide (CTAB), leupeptin (hemisulphate salt), lithium chloride, L-malate (disodium salt), octanol, mineral oil, M_r marker proteins for SDS/polyacrylamide gel electrophoresis, MSMO salts, polyvinylpolypyrrolidone (PVPP or insoluble PVP), propionic acid, silver nitrate, sodium azide, N,N,N',N'-tetramethylethylenediane (TEMED) and Tween 20 were obtained from the Sigma Chemicai Co. (London), Poole, Dorset, UK.

.;;

ATP (disodium salt), dithiothreitol (DTT), nicotinamide adenine dinucleotide reduced form (NADH, disodium salt), phosphoenolpyruvate (monosodium salt), and pig heart malate dehydrogenase were from Boehringer Corp. (London) Ltd., Lewes, Sussex, UK. Sephadex G-25 (medium) and Sephadex G50 were from Pharmacia, Milton Keynes, Bucks., U.K.

Acrylamide monomer, ammonium persulphate, ethanol, glacial acetic acid, glycine, hydrogen peroxide, 2-mercaptoethanol, methanol, N',N' methylenebisacrylamide, potassium dihydrogen orthophosphate, sodium dodecyl sulphate (SDS) and trichloracetic acid were 'AnalaR' grade materials from BDH Chemicals, Poole, Dorset, U.K.

37.5:1 acrylamide:bis-acrylamide solution was from Bio-Rad Laboratories (England) Ltd., Bramley, Kent, U.K.

Acetone, ammonium sulphate, boric acid, chloroform, disodium hydrogen phosphate, ethylene diamine tetra-acetic acid (EDTA, disodium salt), Hepes, hydrochloric acid,

magnesium chloride, magnesium sulphate, Optiflow Safe 1 scintillation fluid, potassium chloride, sodium acetate, sodium chloride, sodium dodecyl sulphate, sodium hydrogen carbonate, sodium hydroxide and sucrose were from Fisher Scientific UK Ltd., Loughborough, Leics., LE11 5RG, UK.

Agarose (electrophoresis grade) and 1 kb DNA ladder were from Gibco BRL Life Technologies Ltd., Paisley, Scotland.

Agar and yeast extract were from Merck, Darmstadt, Germany.

Okadaic acid was from Moana Bioproducts.

AMV reverse transcriptase, Poly Atract mRNA isolation system, RNA size markers (0.28 - 6.6 kb), Taq DNA polymerase and all DNA restriction enzymes were from Promega (U.K.) Ltd., Southampton, U.K.

T7 mMessage Machine *in vitro* transcription kit was from Ambion Inc., AMS Biotechnology (U.K.) Ltd., U.K.

QIA-quick gel extraction kit and QIA-prep spin plasmid miniprep kit were from Qiagen Ltd., Crawley, West Sussex, U.K.

All other reagents used were of the highest grade commercially available.

2.2 EXPERIMENTAL TISSUES

2.2.1 Culture material

An Arabidopsis thaliana photomixotrophic suspension culture was established (May and Leaver, 1993). Cells were grown under continuous low intensity white light (about 25 μ mol/m²/s) at 20°C with constant shaking at 110 rpm. Subculturing was performed after 7 days by transferring 20 ml of culture to 180 ml fresh sterile medium in a 500 ml conical flask. The culture medium consisted of:

3% sucrose; 0.5 mg/L Naphthalene Acetic Acid; 0.05 mg/L Kinetin; 1x Murashige & Skoog Minimal Organic (MSMO) salts.

MSMO salts contain:

1650.0 mg/L ammonium nitrate; 6.2 mg/L boric acid; 332.2 mg/L calcium chloride anhydrous; 0.025 mg/L cobalt chloride. $6H_2O$; 0.025 mg/L cupric sulphate. $5H_2O$; 27.8 mg/L ferrous sulphate. $7H_2O$; 180.7 mg/L magnesium sulphate; 16.9 mg/L manganese sulphate. H_2O ; 0.25 mg/L molybdic acid (sodium salt). $2H_2O$; 0.83 mg/l potassium iodide;
1900.0 mg/L potassium nitrate; 170.0 mg/L potassium phosphate monobasic; 8.6 mg/L zinc sulphate.7H₂O; 100.0 mg/L myo-inositol; 0.4 mg/L thiamine/HCl.

The medium was then adjusted to pH 5.8 with 0.2 M KOH and sterilised by autoclaving.

2.2.2 Plant material

(i) Seeds of wild type *A. thaliana* var. Columbia or Landsberg erecta were spread out onto watered soil in plant pots, covered with elingfilm to prevent moisture loss and left to cold-imbibe at 4°C in the dark for 3 days. The pots were then transferred to 16 hour light/8 hour dark conditions at approximately 70 μ moles/m²/s white light and 21°C and 18°C in the light and dark respectively. Other pots were put under constant light conditions. Plants were watered every 2 to 3 days.

Rosette leaves, bracts, bolts, flowers and buds were taken from 5-week old plants for the Northern blot analysis of PEPc and PEPc kinase gene expression in mature tissue. The rosette leaf tissue included all rosette leaves removed from a few individuals of approximately the same size and development, whereas only the first bracts from a group of similarly sized plants were taken. Only primay bolts of approximately 15 cm were harvested. These were excised approximately I cm from the base and the bracts, secondary bolts, and top of the bolt 1 cm below the first silique were removed. Completely green siliques of no less than 1.5 cm were excised from plants 6-8 weeks old. Suitable root tissue was obtained by spreading sterilized seed on 1.5% agar plates containing 0.5x MS salts and 20 mM sucrose. The seeds were cold-imbibed at 4°C in the dark for 3 days then grown for 10 days under constant illumination ($\sim 20 \,\mu \text{moles/m}^2/\text{s}$) at 20°C. The seedlings were then transferred into glass conical flasks (six seedlings per flask) containing sterile A. thaliana root culture medium (see section 2.4) and shaken at 80 rpm under the same growth conditions as previously. After 14 days the seedlings had grown extensive root systems (Czako, M. and Marton, L., in ATCG Arabidopsis: The Compleat Guide. AFRC PMB Arabidopsis Programme).

(ii) *Kalanchoë (Bryophyllum) fedtschenkoi* Hamet et Perrier was propagated by cuttings from the original stock used in previous studies (Wilkins, 1959; Wilkins, 1960). Cuttings were grown in a glasshouse under a 16 hour photoperiod, maintained throughout the year using mercury-vapour lamps. Four to six month old plants were transferred to either of two controlled environment growth chambers. In one chamber the 8 hour photoperiod

was from 0800 h - 1600 h and in the other the photoperiod was 1600 h - 2400 h (reverse-phase). Light was provided by white fluorescent tubes and twelve 100W tungsten lamps, giving a radiant fluence rate of 20 W/m²/s. The temperature was 28°C during the photoperiod and 15°C in the dark. Plants were watered every 4 to 5 days. Plants were allowed to adjust to growth chamber conditions for at least 7 days prior to use. All experiments were carried out using leaf material from between nodes six and ten.

200 A.V.

÷

10.00

ALC GAME I

2

2.3 GENERAL BIOCHEMICAL METHODS

2.3.1 pH calibrations

All pH calibarations were performed using a Russel pH probe connected to an EDT Instruments RE 357 Microprocessor pH meter. The pH of all buffers and solutions was adjusted at their working temperature (i.e. 4°C or room temperature).

2.3.2 Glassware and plastics

Glassware and plastics were washed in hot water containing the detergent Pyroneg, rinsed in clean water and dried in ovens.

2.3.3 Spectrophotometric assays

All spectrophotometric assays were carried out in semi-micro quartz cuvettes or plastic disposable cuvettes (1 cm path length, 1 ml volume) using either a Gilford Instrument 2600 spectrophotometer or a Philips PU 8700 series UV/Vis scanning spectrophotometer for the determination of DNA/RNA solution concentrations and enzyme assays.

2.3.4 Centrifugation

Beckman J2-HS and J2-21 centrifuges were used with the appropriate Beckman centrifuge rotors for all centrifuge steps. Centrufugation of small volume samples (<1.5 ml) was carried out in an MSE Microcentaur microfuge.

2.3.5 Micropipetting

 $(0.5 \ \mu l - 1 \ m l)$ All micropipetting was performed using adjustable Finnpipette micropipettes.

2.3.6 Chromatographic materials

Sephadex G25 and G50 were swollen and packed according to the manufacturer's protocols. Sephadex G25 was stored in 0.05 % (w/v) azide.

1

ł,

2.4 **BUFFERS AND SOLUTIONS**

A. *thaliana* extraction buffer at pH 8.0 consisted of: 100 mM Tris/HCl; 2 mM EDTA; 10 mM malate; 2% (w/v) polyethylene glycol 20,000; 1 mM DTT; 1 mM benzamidine hydrochloride.

A. thaliana root culture medium at pH 5.8 contained per litre: 4.3 g MS salts; 3 ml 6% (w/v) KH₂PO₄; 200 mg myo-inositol; 2 ml Vitamix stock; 30 g sucrose.

A. thaliana RNA extraction buffer consisted of: 25 mM Tris/HCl, pH 8.0; 25 mM EDTA; 75 mM NaCl; 1% SDS; 7.8% β-mercaptoethanol.

Bradford reagent consisted of: 100 mg Coomassie Brilliant Blue (G250); 50 ml 95% ethanol; 100 ml 85% (w/v) phosphoric acid.

CTAB RNA extraction buffer consisted of: 2% CTAB; 2% PVP 40; 100 mM Tris/HCl, pH 8.0; 25 mM EDTA; 2 mM NaCl; 0.5 g/l spermidine/HCl; 2% β -mercaptoethanol. The prepared solution was autoclaved and the 2% β -mercaptoethanol then added.

6x DNA loading buffer consisted of: 30% (v/v) glycerol; 0.25% (w/v) xylene cyanol; 0.25% (w/v) bromophenol blue.

K. fedtschenkoi desalting buffer at pH 7.5 consisted of: 100 mM Tris/HCl; 1 mM benzamidine hydrochloride; 1 mM DTT; 2 mM EDTA; 10 mM L-malate.

Laemmli sample buffer consisted of: 50 mM Tris/HCl; 1% (w/v) SDS; 10% (v/v) glycerol, 0.01% (w/v) bromophenol blue; 1% (v/v) 2-mercaptoethanol.

Lauri Bertrani (LB) medium contained per litre: 10 g Bactotryptone; 5 g bacto yeast extract; 10 g NaCl adjusted to pH 7.0 with 5 M NaOH.

10x Mops buffer at pH 7.0 and 8.0 consisted of: 0.2 M Mops; 50 mM sodium acetate; 10 mM EDTA.

MS salts consisted of the same ingredients as MSMO (section 2.2.1) without ammonium nitrate or potassium nitrate and with 2.0 mg/L glycine, 0.5 mg/L nicotinic acid, 0.5 mg/L pyridoxine-HCl and 0.1 mg/L thiamine.

NZYCM agar contained per litre: 1 g Casamino acids (casein hydrolysate Σ type C9386); 5 g NaCl; 2 g MgSO₄.7H₂O; 5 g yeast extract; 10 g NZ amine (casein hydrolysate Σ type C0626); 15 g agar (7 g in top agar).

Phosphorylation buffer at pH 8.0 consisted of: 100 mM Tris/HCl; 5 mM MgCl₂; 1 mM EDTA; 5% glycerol; 1 mM DTT.

RNA sample buffer consisted of: 72 μ g/ml ethidium bromide; 2x Mops at pH 8.0; 6 % formaldehyde; 70 % formamide. All manipulations using the buffer were carried out in a fume cupboard.

SDS-PAGE running buffer consisted of: 3 g/L Tris; 14.4 g/L glycine; 0.1% SDS.

SM buffer contained per litre: 5.8 g NaCl; 2 g MgSO₄.7H₂O; 5 ml 1 M Tris/HCl, pH 7.5; 5 ml 2% gelatin.

Sonication buffer at pH 7.4 consisted of: 50 mM Hepes/KOH; 0.45 M sorbitol and 1 mM CaCl₂.

20x SSC at pH 7.0 consisted of: 3 M NaCl, 0.3 M trisodium citrate, pH 7.0.

0.5x TBE buffer at pH 8.0 consisted of: 45 mM Tris-borate; 1 mM EDTA.

TE at pH 8.0 consisted of: 10 mM Tris/HCl, 1 mM EDTA.

Vitamix (500x stock solution) contained per 100 ml: 500 mg thiamine pyrophosphate (vitamin B1); 50 mg pyridoxal phosphate (vitamin B6); 100 mg glycine; 50 mg nicotinic acid; 25 mg folic acid; 50 mg biotin (vitamin H).

2.5 METHODS FOR THE EXTRACTION OF ENZYME ACTIVITIES

2.5.1 A. thaliana cell culture

20 ml of cell culture was harvested and centrifuged in a Beckman for 5 min and 1,000 rpm. The pelleted cells were then resuspended in 1 ml of sonication buffer.

Aliquots (1 ml) of the cell suspension were sonicated for 30 s at an amplitude of 5 microns using the Sanyo Soniprep and spun for 2 min at 13,000 rpm in a microfuge. The supernatant was then spun for a further 20 min at 30,000 rpm using a Sorvall OTD ultracentrifuge. The resulting supernatant was precipitated with 75% ammonium sulphate on ice for 30 min. The material was then spun for a further 2 min at 13,000 rpm and the pellet resuspended in 50 μ l of sonication buffer.

2.5.2 A. thaliana leaf extracts

1 g of *A. thaliana* leaves was collected from plants which had flowered and then had their bolts removed. The leaves were ground up in a mortar and pestle at 4°C with some sand, a few drops of octanol, 1 ml/g extraction buffer, and 5% (w/v) NaHCO₃.

i.

Extracts were desalted as described in section 2.5.4.

A 1 ml volume of desalted extract was then taken to 75% saturation with ammonium sulphate and left on ice for 30 min to precipitate. Finally the extract was spun at 13,000 rpm for 2 min and the pellet resuspended in 100 μ l of phosphorylation buffer (see section 2.4).

2.5.3 K. fedtschenkoi leaves and leaf disks

K. fedtschenkoi leaves and leaf disks were ground up as described in section 2.5.2. The extracts were desalted as described in section 2.5.4 and assayed directly.

2.5.4 Desalting crude extracts

Crude extracts from *A. thaliana* leaves and *K. fedtschenkoi* leaves and disks were spun at 4° C, 13,000 rpm for 5 min. The resulting supernatant was then loaded onto a 10 ml G25 Sephadex column using a glass pasteur pipette and desalted by pumping buffer through the column using a peristaltic pump. Phosphorylation buffer was used to pre-equilibrate and desalt the *A. thaliana* extracts (section 2.4).

Fractions of the desalted extract were collected as approximate 500 μ l volumes in 1.5 ml microcentrifuge tubes. Fractions were assayed for phosphoenolpyruvate carboxylase (PEPc) activity (section 2.6.1) and the fractions containing the peak of activity pooled.

K. fedtschenkoi leaves and disks were desalted in the K. fedtschenkoi desalting buffer (see section 2.4).

Fractions of the desalted extract were collected as 1 ml volumes in glass test tubes using a BIORAD model 2110 fraction collector, assayed for PEPc activity and the fractions containing the peak of activity pooled.

2.6 ASSAY PROCEDURES

All assays were repeated, in duplicate at least, unless otherwise stated. PEPc activities given are the mean of two assays that agree to within 15% and PEPc kinase activity data is representative of duplicate experiments.

2.6.1 Determination of PEPc activity and the apparent K_i of PEPc for L-malate

PEPc assay reaction mix was added to the volume of cell extract being assayed to give a total reaction volume of 1 ml in a 1 ml plastic cuvette. The cuvette was then inverted five times to mix. The oxidation of nicotinamide adenine dinucleotide (NADH) by the coupling enzyme malate dehydrogenase (MDH) was monitored as the decrease in the A_{340} at 25°C. This decrease was proportional to the PEPc concentration in the enzyme sample (Nimmo et al., 1984). The difference in rate of change of absorbance was attributed to PEPc activity. Contaminating NADH oxidase activity was accounted for by measuring the rate of change of absorbance in the absence of PEP.

PEPc activity assay mixture for A. thaliana at pH 7.6 (adapted from Van Quy et al., 1991) consisted of:

50 mM Tris/HCl; 10 mM MgCl₂; 0.3 mM PEP; 20 mM NaHCO₃; 0.2 mM NADH; 5 mM DTT; 5 units NAD-linked MDH.

PEPc activity assay mixture for K. fedtschenkoi at pH 7.8 consisted of:

50 mM Tris/HCl; 10 mM MgCl₂; 2 mM PEP; 10 mM NaHCO₃; 0.2 mM NADH; 5 units NAD-linked MDH.

The L-malate sensitivity of PEPc was determined by measuring the activity of the enzyme in the presence of different concentrations of L-malate. The PEPc activity was plotted against L-malate concentration as a percentage of the activity in the absence of any inhibitor. The apparent K_i for malate of PEPc could then be read from this graph.

2.6.2 PEPc kinase activity

A volume of desaited extract equivalent to 0.001-0.003U PEPc was used in reactions to assay extracts for PEPc kinase activity. PEPc kinase reaction mixture at pH 7.8 consisted of:

50 mM Tris/HCl; 1 mM benzamidine hydrochloride; 10 μ g antipain/ml; 10 μ g leupeptin/ml; 5 nM okadaic acid; 0.03 units purified, dephosphorylated PEPc from *K*. *fedtschenkoi*; 5 mM MgCl₂; 1 μ Ci [γ -³²P] ATP; 0.1 mM ATP.

The reaction mixture was assembled in 1.5 ml microcentrifuge tubes on ice and the ATP mix (0.4 μ Ci/nmol) was added last, initiating the reaction. The total reaction volume was 25 μ i. All reactions were then transferred to a 30°C water bath and incubated for 30 min. The reactions were stopped by removing all the tubes to an ice bath and adding 10 μ i Laemmli sample buffer (section 2.4) before boiling for 4 min to denature the samples. The denatured samples were then run on an 8% SDS/polyacrylamide gel, stained, destained overnight and the gel dried down (see section 2.7). The gel was then either phosphorimaged or autoradiographed as preferred (see section 2.7.3) to determine the incorporation of ³²P into the major PEPc band for each sample. (*K. fedtschenkoi* PEPc contains two related subunits with molecular mass values of 112 000 and 123 000 in a ratio of some 10:1 (Nimmo et al., 1986)). This incorporation is a measure of the PEPc kinase activity for each sample.

3

ę

ĵ

For kinase assays on products of *in vitro* translation, 5 μ l of each translation reaction was used. The kinase reactions were stopped by addition of 10 μ l of rabbit anti-*K*. *fedtschenkoi* PEPc antiserum. The PEPc was immunoprecipitated on ice for 1 hour. The samples were spun for 5 min at 13,000 rpm in a microfuge at room temperature and the supernatant removed. The pellet was washed in 200 μ l of 2 M NaCl / 2 mM EDTA and spun for another 5 min. The pellet was then resuspended in 10 μ l of Laemmli sample buffer and boiled for 4 min to denature protein.

Partially purified maize PEPc kinase (a gift from Dr. GA Nimmo) was used as a control in all PEPc kinase activity assays.

2.6.3 Malate Dehydrogenase (MDH) activity

MDH activity of samples was measured by the rate of decrease in the A_{340} at 25°C due to oxidation of NADH. The total reaction volume of assay reaction mix and sample was 1 ml. The cuvette was inverted five times to mix. The assay mix, pH 7.0, consisted of: 0.15 mM NADH; 1.0 mM OAA; 50 mM NaH₂PO₄/Na₂HPO₄.

2.7 GEL ELECTROPHORESIS TECHNIQUES

2.7.1 SDS/polyacrylamide gel electrophoresis

Proteins were separated by discontinuous SDS-PAGE according to the method originally outlined by Laemmli (1970). All SDS-PAGE gels contained an 8% polyacrylamide

separating gel and a 3% stacking gel. Samples were denatured by the addition of one fifth volume of Laemmli sample buffer (section 2.4) to achieve a final concentration of 1x, and heating in a boiling water bath for 4 minutes. Samples were then loaded on to the gel and electrophoresed at 60 - 70 mA for 2 - 2.5 hours until the tracking dye had left the bottom of the separating gel.

2.7.2 Coomassie Staining of SDS/polyacrylamide gels

Gels were stained for protein in the following solution for 30 min at 37°C:

0.1% Coomassie Brilliant Blue G250; 50% (v/v) methanol; 10% (v/v) acetic acid.

The gels were then destained at 37°C for 30 min, the destain changed and left overnight. The destain consisted of:

10% (v/v) methanol; 10% (v/v) acetic acid.

2.7.3 Drying, autoradiography and phosphorimaging of gels

Destained gels were dried onto Whatman 3 MM chromatography paper using a Biorad Laboratories Slab Gel Drier model 1125 connected to an Aquavac Junior multi-purpose vacuum unit (Uniscience Ltd., 12-14 Ann's Crescent, London SW18 2LS, UK.). Gels which had been dried were autoradiographed using Fuji RX X-ray film with an intensifying screen. Autoradiography, which although less rapid than phosphorimaging results in a higher resolution image, was carried out at -80°C before developing the film using a Kodak X-OMAT Processor Model ME-3. The period of exposure was usually between 3 hours and five days. Radioactive gels were also routinely phosphorimaged using a Fuji Bio-Imaging Analyser (Fuji Photo Film Co. Ltd., Japan). Exposed plates were developed automatically by the Fuji Bio-Imaging Analyser and the images were captured onto a Macintosh Quadra 650 computer running Mac-Bas software (Fuji Photo Film Co. Ltd., Japan). Phosphorimages were analyzed by quantifying the intensity of each band. These intensities were then expressed as a percentage or relative intensity of the most intense band.

2.7.4 Agarose gel electrophoresis of nucleic acids

A 1% agarose get was prepared by adding 1 g of agarose to 100 ml of 0.5x TBE buffer (see section 2.4).

The gel solution was boiled until all the agarose had dissolved. When the solution was sufficiently cool to pour, 1 μ l of 10 mg/ml ethidium bromide was added and the solution poured into the gel apparatus. Approximately 20 μ l of sample was loaded into each well with an appropriate amount of 6x DNA loading buffer (see section 2.4).

Marker DNA was run on agarose gels to estimate the size of the DNA that was being analyzed. The 1 kb ladder was most frequently used. This contained marker bands ranging in size between 75 bp and 12 kb.

Gels were run at 80 mA until the required resolution of DNA fragments was achieved. Ethidium bromide stained bands were visualised on a UV transilluminator and recorded with a polaroid camera.

2.7.5 Denaturing agarose gel electrophoresis of RNA

This method ensures accurate separation of RNA according to size as the denaturing conditions remove secondary structures that can cause RNA to run at aberrant molecular weights in non-denaturing agarose gels.

Denaturing agarose gels for the electrophoresis of RNA were made by dissolving 1 g of agarose in 100 ml of water in the microwave. When the agarose had cooled to about 60° C, 10% (v/v) formaldehyde and 1x Mops buffer at pH 8.0 (see section 2.4) were added to the cooling solution. The gel was then poured and allowed to set for 45 min. All manipulations were carried out in a fume cupboard.

Equal volumes of RNA sample and RNA sample buffer (see section 2.4) were mixed and heated to 65°C for 2.5 min and then snap cooled on ice. This denatured any secondary structures.

The sample was then loaded onto the gel and run out at 100 mV for 2-4 hours in 1x Mops at pH 7.0.

2.8 NORTHERN AND SOUTHERN BLOTTING.

2.8.1 Capillary transfer of nucleic acids onto nylon membranes

DNA and RNA fragments were separated by gel electrophoresis as described in sections 2.7.4 and 2.7.5 respectively.

Prior to blotting of DNA, the nucleic acid was denatured by soaking the agarose gel in 3 volumes of denaturing solution (0.5 M NaOH, 1.5 M NaCl) for 45 min and then

neutralised by soaking in neutralising solution (1 M Tris/HCl (pH 8.0), 1.5 M NaCl) for 45 min. The gel was rinsed in 2x SSC (see section 2.4) before blotting. A support was prepared onto which was placed a wick of Whatman 3 MM paper which was soaked in 20x SSC (see section 2.4). The gel was placed on top of the wick with the well side of the gel face down, ensuring that there were no air bubbles between the wick and the gel. A piece of Hybond-NTM nylon membrane and two pieces of Whatman 3 MM paper were cut to the size of the gel, soaked in 2x SSC and placed on top of the gel in the order given. The immediate area of the wick surrounding the gel was covered with parafilm to prevent a short circuit and a large quantity of paper towels folded to the approximate size of the gel was then placed on top of the 3 MM paper. A glass plate was placed on top. The blot was left overnight for the capillary transfer of the DNA from the agarose gel onto the nylon membrane to occur. The blot was then disassembled and the membrane dipped briefly in distilled water to remove any excess salt deposits. DNA was then fixed to the membrane using the UVP CL-1000 ultraviolet crosslinker at 120,000 μ J/cm².

<u>, </u>

۲. . .

÷

Capillary transfer of RNA onto a nitrocellulose membrane was performed as for Southern blotting but the denaturation step prior to setting up the transfer was unnecessary.

2.8.2 Random priming of cDNAs to synthesize radioactive probes

eDNA fragments were radiolabelled for use as probes on blots by the method of random priming using the Klenow fragment of DNA polymerase I. Approximately 25 ng of DNA was labelled with the Decaprime or *redi*prime DNA labelling system (Amersham, UK) according to the manufacturer's protocol, using 50 μ Ci of Redivue [α -³²P]dCTP (Amersham, UK). Once synthesized, probes were denatured by heating to 95-100°C for 5 min followed by chilling on ice before adding to the hybridization solution as described in section 2.8.4.

2.8.3 Separation of labelled DNA from un-incorporated radionucleotides using spin columns

This was achieved by constructing a mini Sephadex G50 column in a 0.5 ml microcentrifuge tube sitting inside a larger 1.5 ml microcentrifuge tube. A small hole was

made in the bottom of the smaller eppendorf using a small bore needle and some sterile washed glass beads packed into the bottom of the tube. G50 Sephadex in TE, pH 8.0 (see section 2.4) was then added to the top of the tube and spun until dry. The probe was then loaded onto the column and spun through. The un-incorporated radionucleotides became trapped in the Sephadex and the larger molecules of radiolabelled DNA passed through the column and were collected in the bigger 1.5 ml tube in which the spin column was located.

The incorporation into and the specific radioactivity of cDNA probes were determined by measuring the ³²P-radioactivity of the probe before and after passing through the spin column. 1 μ l samples were diluted in 3 ml of scintillation fluid and counted in a scintillation counter.

2.8.4 Hybridization of radiolabelled DNA probes to Northern and Southern blots

All hybridization steps were performed in an hybridization oven (Techne Hybridizer HB-1D). Blots were incubated at 55°C for 1-2 hours in 20 ml pre-hybridization solution which consisted of:

0.5 M Na₂HPO₄, pH 7.2; 7 % SDS; 10 mg/ml BSA.

The radioactive DNA probe was then added to the pre-hybridization solution and the blot left to hybridize overnight at 55°C except where otherwise stated.

The hybridization solution was then poured off and stored at 4°C for future use. Blots were routinely washed in 2x SSC/1% SDS at 55°C for 10 min. This was followed by a number of washes at higher stringency (depending on the radioactivity of the blot) until the signal to background ratio was acceptable.

Washed filters were heat sealed in plastic and autoradiographed or phosphorimaged as preferred.

2.8.5 Stripping Hybond-N membranes of radioactive nucleotides

After probing, Hybond-N membranes were stripped of radioactive nucleotides by washing in boiling 0.1% SDS until radioactivity could no longer be detected by a Geiger counter.

2.9 DNA EXPERIMENTS

All experiments using DNA were carried out only once unless otherwise stated.

2.9.1 Prevention of degradation of DNA samples by DNase contamination

All solutions used in the preparation or manipulation of DNA samples were autoclaved. Tips and plasticware were autoclaved and only handled with gloved hands. On occasion sterile filter tips were used as extra protection against contamination of samples and reagents.

2.9.2 Isolation of plasmid DNA

Plasmid DNA was isolated using the QIAGEN QIAprep Spin Plasmid Miniprep kit (Qiagen, U.K.) following the manufacturer's instructions.

2.9.3 Determination of nucleic acid concentration and quality

100 μ l of water in a quartz cuvette was used to record a baseline scan on a Philips spectrophotometer between 240 nm and 300 nm. 1-2 μ l of nucleic acid sample was added to the cuvette and the sample scanned in the range mentioned. The ratio of absorbance at 260 nm to 280 nm indicates the purity of the nucleic acid (a ratio of 1.8 or greater is desirable). An absorbance at 260 nm (A₂₆₀) of 1 was taken to indicate the following concentrations:

Form of nucleic acid	Concentration (µg/ml)
Double stranded DNA	50
Single stranded DNA/RNA	40
Oligonucleotides	20

2.9.4 Digestion of DNA with restriction endonucleases

The DNA to be digested was prepared in a solution as follows:

1x reaction buffer

approximately 2 units of restriction enzyme

1x BSA

2 µg DNA

sterile distilled water.

The reactions were incubated at 37°C for 1-24 hours and the reactions were monitored where sufficient DNA was available by running an aliquot of the reaction on an agarose gel against uncut DNA.

2.9.5 Polymerase Chain Reaction (PCR)

The reactions were assembled on ice in a total volume of 25 μ l and overlaid with some mineral oil. Reactions contained template DNA (approximately 20-200 ng) and the following components:

1x reaction buffer; 2.5 mM MgCl₂; 0.5 mM dNTP mix; 0.5 μ M primer 1; 0.5 μ M primer 2: 1 unit Tag polymerase.

The annealing temperature for each PCR was decided upon from the melting temperature of the primers calculated using the following equation:

Annealing temperature $(^{\circ}C) = n[G + C] + n[A + T] - 5$ where n is the total number of each purine/pyrimidine base pair.

The PCR protocol used was as follows:

Denaturation - 95°C for 1 min

- X°C for 30 s 30 cycles Annealing

- 72°C for 2 min ----Extension

PCR products were separated on a 1% agarose gel and the ethidium bromide stained bands visualised using an UV-illuminator.

2.9.6 Sequencing of DNA

The nucleotide sequence of both cDNA strands was determined by the dideoxynucleotide chain-termination method. The first approximately 300 bp of 5'- and 3'- sequence were determined using the T7 Sequenase Version 2.0 DNA Polymerase Sequencing kit (see protocol for instructions). Subsequent rounds of sequencing were carried out in collaboration with the Molecular Biology Sequencing Unit, IBLS, University of Glasgow, Glasgow G12 8QQ using automated sequencing.

2.9.7 Design of primers

Specific primers for PEPc and PEPc kinase from *A. thaliana* and all primers used for the sequencing of the *A. thaliana* PEPc clone were designed using GeneJockey II software. The conserved kinase primers were degenerate primers designed to the conserved domains I and VIII of *K. fedtschenkoi* and *A. thaliana* PEPc kinase sequences. The sequences for the conserved PEPc primers and *A. thaliana* α -tubulin primers were obtained from Honda et al., (1996) and Jackson et al., (1995), respectively. All primers were synthesized by Cruachem Ltd., Todd Campus, West of Scotland Science Park, Acre Road, Glasgow G20 0UA.

The alignment of the different primers to the appropriate cDNA is shown diagrammatically in Figure 2.1. The fragment size amplified and the annealing temperatue used are also given in this figure.

2.10 RNA EXPERIMENTS

All experiments using RNA were carried out only once unless otherwise stated.

2.10.1 Prevention of RNA sample degradation by RNase contamination

All solutions used in the preparation or manipulation of RNA samples were made with DEPC-treated water. Tips and plasticware were autoclaved and handled only with gloved hands. On occasion sterile filter tips were used as extra protection against contamination of samples and reagents.

2.10.2 Isolation of RNA from plant tissue and cell culture

(i) *A. thaliana* cell culture and plant tissue. Cell culture samples were filtered through Whatman 3 MM paper using Buchner apparatus under vacuum to remove liquid medium from the samples. The cells were then weighed and packaged in aluminium foil before being snap-frozen and stored at -80°C until required.

63

Figure 2.1 Alignment of primers for A. thaliana PEPc, PEPc kinase and α -tubulin

The primer sequences for each cDNA sequence are given below along with the size of fragment amplified and the annealing temperature. The alignment site for each primer is shown diagrammatically opposite as a red arrow (sense primer \rightarrow ; antisense primer \leftarrow).

A. thaliana PEPc-specific primers:

fragment size 174 bp; annealing temperature 55°C sense 5'- GGCAAGCAGGAGGTCATGATCGG -3' antisense 5'- GAAGTACTCGACAAATCGAGGCTC -3'

Conserved PEPc primers:

fragment size 404 bp; annealing temperature 55°C 5'- GGCAAGCAGGAGGTCATGATCGG -3' 5'- GAAGTACTCGACAAATCGAGGCTC -3'

A. thaliana α-tubulin-specific primers:

fragment size 501 bp; annealing temperature 55°C 5'- TTTCTTTAGCGACTGGTGC -3' 5'- TCAACCTGTTCAAGTTTGTG -3'

Conserved PEPc kinase primers:

fragment size 468 bp; annealing temperature 55°C 5'- TGCGAGGAGATCGGCGKG -3' 5'- ACCTCCGGCGCCACGTARTAC -3'

A. thaliana PEPc kinase-specific primers:

fragment size 501 bp; annealing temperature 55°C 5'- ATGGCTCTCTTGTCCTATCACC -3' 5'- CTTTAGCCATAGATGAAACCCC -3'



A. thaliana PEPc with A. thaliana PEPc-specific primers

65

Cell culture and *A. thaliana* tissue were ground up in a chilled mortar and pestle using liquid nitrogen. 900 μ l of RNA extraction buffer was then added (see section 2.4).

Each sample was ground in the buffer until it was like fine sand. 900 μ l of phenol:chloroform:isoamyl alcohol (PIC) in a ratio of 25:24:1 was then added and the sample ground very fast. Tissue was kept frozen throughout the extraction by repeated additions of liquid nitrogen to the mortar and pestle. Each sample was then transferred to a microfuge tube, allowed to thaw at room temperature and spun for 10 min at 13,000 rpm at 4°C. The aqueous layer was then removed and put into a new eppendorf. 1 volume of PIC was added, the sample was vortexed and then spun again. This was repeated until a clean interface between the organic and aqueous phase was obtained. The aqueous layer was removed to a new tube, 1 volume of chloroform was added and the sample was spun. The aqueous phase was again removed to a new tube and 10 M LiCl added to a final concentration of 2 M. The sample was mixed very quickly by inversion to prevent DNA precipitation and then left overnight at 4°C.

Precipitated RNA was collected by centrifugation at 13,000 rpm for 15 min at 4°C and the supernatant was carefully removed. The pellet was washed with 500 μ l 2 M LiCl and then vortexed to lift the pellet. The sample was spun again and the supernatant was removed completely. The pellet was then resuspended in 500 μ l DEPC-treated water. 0.1 volume of 3 M sodium acetate, pH 5.5 was added and the sample was vortexed. 2.5 volumes of ethanol was then added and vortexed and the sample was left on ice for 15 min. The sample was then spun for 15 min and the supernatant was removed. The pellet was washed with 70% ethanol, the sample was spun for 5 min and the supernatant was removed. The pellet was air-dried by inverting the tube on paper towel for 10-15 min. The dried pellet was resuspended in DEPC-treated water and its purity and concentration determined by measuring the absorbance at 260-280 nm.

Samples were frozen at -80°C until required.

(ii) K. fedtschenkoi leaves and disks. 2-3 g of tissue was ground up in a mortar and pestle using liquid nitrogen. The ground tissue was then added to a tube containing 10 ml of CTAB extraction buffer (see section 2.4) in a water bath at 65° C and the tube was inverted to mix. The lid of the tube was previously punctured using a syringe needle. An equal volume of chloroform (i.e. 10ml) was then added to the tube which was spun at

9000 rpm for 10 min at 4°C in a Beckman centrifuge to extract the RNA. The supernatant was then removed and another 10 ml of chloroform was added to the tube which was centrifuged at 5000 rpm for 10 min. The supernatant was again removed and 3 ml of 10 M LiCl was added. The RNA was then left to precipate overnight at 4°C.

After centrifugation at 9000 rpm for 20 min at 4°C the resulting pellet was dissolved in 500 μ l sterile TE, pH 7.5 (see section 2.4) and transferred to a microfuge tube. The RNA was extracted with 500 μ l of chloroform and microfuged at 13,000 rpm for 10 min at 4°C. The top aqueous layer was then removed to a clean tube and 50 μ l of 3 M sodium acetate added. The tube was then filled with ice-cold 100% ethanol and left to precipate on ice for 1 hour. The RNA was then pelleted by spinning at 13,000 rpm for 20 min in a microfuge at 4°C and the pellet dried using a Savant Speed Vac® Plus SC110A. The pellet was then allowed to dissolve in a small volume of DEPC-treated water (e.g. 25 μ l).

2.10.3 Isolation of poly A⁺RNA

Poly A⁺ RNA was obtained from isolated total RNA using Promega's Poly Atract system (Promega, U.K.) following the manufacturer's instructions.

2.10.4 In vitro translation of RNA samples

The rabbit reticulocyte lysate system (Amersham, U.K.) possessed no detectable PEPc kinase activity and was used for the *in vitro* translation of RNA samples. 100-200 μ g/ml of total RNA were translated according to the manufacturer's protocols using RedivueTM [³⁵S] - methionine as the labelled amino acid. Incubations were for 45 min at 30°C. The incorporation of [³⁵S] - methionine into protein was measured by precipitation with trichloroacetic acid according to the maufacturer's instructions to allow standardisation of the subsequent PEPc kinase assays.

RNA isolated from *K. fedtschenkoi* leaves in the middle of the night by Dr. J Hartwell was used as a control in all *in vitro* translation-kinase assays.

2.10.5 RT-PCR

RNA samples were reverse-transcribed using Promega's AMV reverse transcriptase. 0.5 μ g primer / μ g RNA in a total volume of $\leq 15 \mu$ l in water was heated to 70°C for 5 minutes to denature any secondary structure and then chilled on ice for 5 minutes. The reverse transcription reaction components were then added to the annealed primer/template in the order below:

AMV RT 1x reaction buffer; 10 mM dNTP mix; 40 units RNasin® ribonuclease inhibitor; 4 mM sodium pyrophosphatase (prewarmed to 42°C); 30 units AMV RT.

The reaction mix was incubated at 42° C for 60 minutes and then put on ice. 5 µl of reverse transcription products were then put into a 50 µl PCR containing:

1x reaction buffer; 2.5 mM MgCl₂; 0.5 mM dNTP mix; 0.5 μ M primer 1; 0.5 μ M primer 2; 1 unit Taq polymerase.

The reactions were assembled on ice and overlayed with some mineral oil. The reaction conditions used were as follows:

95°C for 2 minutes;

55°C for 30 s, 72°C for 1 minute and 95°C for 30 s for 30 cycles;

55°C for 1 minute;

72°C for 5 minutes.

The PCR products were then run out on a 1% agarose gel and the ethidium bromidestained bands visualised using an UV-illuminator.

2.10.6 In vitro transcription of A. thaliana PEPc kinase cDNA

In vitro transcription of A. thaliana PEPc kinase cDNA was carried out using T7 mMessage Machine according to the manufacturer's instructions. The kit requires that linear DNA be used as the template for the transcription reaction and so plasmid pZL1 that contained the kinase cDNA was linearized by digestion with *Hind* III (section 2.9.4).

2.11 CLONING TECHNIQUES

2.11.1 Preparation of plating bacteria

A stab of plating cells *E.coli* strain Y1090ZL was obtained from the Arabidopsis Biological Resource Center (ABRC), Ohio, USA and streaked out on NZYCM agar (see section 2.4).

The solution was calibrated to pH 7.5 before dissolving the agar. A single colony was then used to inoculate 5 ml of sterile Lauri Bertrani (LB) medium (see section 2.4) in a sterile universal and grown at 37°C and 250 rpm in a rotary shaker.

When this "starter culture" had developed a cloudy appearance after approximately 2-3 hours, 100 μ I was removed and added to 100 ml of sterile LB medium supplemented with 0.2% maltose in a sterile 250 ml glass conical flask. This was grown up under the same conditions until a A₆₀₀ of about 0.4 was reached. The cells were then centrifuged at 4000 g for 10 min at room temperature. The supernatant was discarded and the pellet was resuspended and washed in sterile 0.01 M MgSO₄ by centrifuging at 4000 g for 10 min at room temperature. The pelleted cells were again resuspended in 0.01 M MgSO₄ to a A₆₀₀ of 2.0. The cells were stored at 4°C and used within 3 days. Approximately 200 μ I of cells were added to 4 ml top NZY agar for plating out the bacteriophage in which the cDNA of the λ PRL2 library is packaged.

2.11.2 Titration of λ PRL2 cDNA library and λ ZIPlox recombinants

An *A. thaliana* cDNA library (λ PRL2) was obtained from the Arabidopsis Biological Resource Center (ABRC), Ohio, USA. The given titre of the λ PRL2 library was 10⁸/ml. In order to confirm this, a few microlitres of the library were taken and serial dilutions were made in SM buffer (see section 2.4).

These dilutions were then incubated at 37° C for 20 min with 100 µl of Y1090ZL plating cells before being added to 4 ml of top agar (48°C) in a sterile 15 ml Falcon tube, vortexed briefly and immediately poured onto 20 ml of hardened, pre-warmed NZYCM bottom agar supplemented with 0.1 M MgSO₄ and 0.2% maltose in a 90 mm petri dish. The plate was swirled gently to allow the formation of a level top agarose layer and then left to stand on a flat surface at room temperature. When the top agar had hardened, the

Petri dishes were inverted and incubated at 37°C for 12-16 hours. The number of plaques that formed in the bacterial lawn were counted and found to be much lower than the value estimated from the titre quoted above.

2.11.3 Preparation of A. thaliana ESTs

A PEPc EST (identified as in Chapter 4) was obtained from ABRC as a stab. A loopful of the stab was streaked onto an LB plate containing 100 μ g/ml ampicillin and 10 mM MgCl₂. A colony was then used to inoculate 5 ml of LB broth containing the required ampicillin selection and the broth was grown overnight at 37°C and 250 rpm. Plasmid was isolated from the bacteria as described in section 2.9.2.

2.11.4 Immobilization of λ ZIPlox recombinants outo nitrocellulose membrane

The method used was based on that of Benton & Davies (1977). Plates were prepared as outlined above (2.11.2). As the titre of the library was much lower than that specified, the entire library was plated out on six NZY agar plates in 140 mm diameter petri dishes with the Y1090ZL cells. 90 mm diameter plates were used for all subsequent rounds of screening. When the Y1090ZL bacteria had formed a lawn on the NZY agar the plates were chilled for 2 hours at 4°C. Using forceps and wearing gloves, 90 mm diameter nitrocellulose circles were then placed carefully onto the top agar of each plate for 2 min. A needle and some ink were used to prick through the filters and the agar for orientation. Duplicate membranes were allowed to transfer for 4 min. The nitrocellulose membranes were then neutralized for 5 min in 1.5 M NaCl and 0.5 M NaOH for 2 min. The membranes were then neutralized for 5 min in 1.5 M NaCl, 0.5 M Tris/HCl (pH 8.0). Finally the membranes were rinsed for 30 s maximum in 0.2 M Tris/HCl (pH 7.5) and 2x SSC buffer solution. After briefly blotting the membranes on Whatman 3 MM paper the DNA was crosslinked to the membranes using a UVP CL-1000 Ultraviolet Crosslinker to deliver 120,000 μ J/cm² of UV energy for about 30 s.

2.11.5 Hybridization using radiolabelled PEPc probe

Hybridization using radiolabelled PEPc probe was carried out in a shaking water bath at 63°C, the calculated hybridization temperature for the 500 bp PEPc probe. Just enough

pre-hybridisation solution was used to cover the filter(s) in the bottom of a plastic box. The pre-hybridization solution contained:

6x SSC (see section 2.4); 20 mM Na_2HPO_4 ; 0.4% SDS; 5x Denhardt's reagent; 500 μ g/ml denatured, sonicated salmon sperm DNA.

After hybridisation, the filters were washed at low stringency in 6x SSC and 0.1% SDS and shaken carefully at room temperature for 5 min. The wash was then repeated with fresh solution for a further 20 min at 63° C.

2.11.6 Isolation of putative positive phage

The plates were aligned with the autoradiographs and plaques containing putative positives were identified. The plaques were then cored out of the agar plates using the end of either a sterile pasteur pipette or yellow tip and put into a sterile eppendorf containing 500 μ l of SM (see section 2.4) buffer. The phage were allowed to elute into the buffer overnight at 4°C.

2.11.7 In vivo excision of recombinant pKP42

Putative positive phage were eluted into SM buffer as described (2.11.6). The phage were then vortexed vigorously for 10 s and incubated at room temperature for 5 min. 50 μ l, 150 μ l and 300 μ l of the supernatant from the eluted phage was mixed with 100 μ l of DH10B(ZIP) cells in a sterile eppendorf and incubated at room temperature for a further 5 min. (The DH10B(ZIP) cells had been prepared as described in 2.11.1 for the plating cells Y1090ZL). The entire mixture was spread onto a NZYCM plate containing 10 mM MgCl₂ and 100 μ g/ml ampicillin. The plate was then inverted and incubated overnight at 37°C. Any colonies that appeared contained recombinant pKP42 plasmid which conferred resistance to ampicillin.

2.12 BRADFORD ASSAY FOR PROTEIN DETERMINATION

The A_{595} of duplicate BSA standards of 2, 4, 8, 12, 16 and 25 µg of protein were used to construct a calibration curve for determining the protein concentration of samples. BSA standards and protein samples were made up to a volume of 25 µl in plastic cuvettes and

975 μ l of dilute Bradford reagent (see section 2.4) was then added to each cuvette which was subsequently inverted five times to mix the contents.

3 parts of reagent were diluted with 17 parts of water immediately before use and filtered through Whatman 3 MM paper.

2.13 GAS EXCHANGE MEASUREMENTS

For experiments on *K. fedtschenkoi*, rates of net CO_2 assimilation were continuously on the same leaf over 24-72 hours. The leaves were enclosed in porometer heads which tracked the environmental conditions in the growth chamber with gas exchange parameters measured using an open infra-red gas exchange system (H. Walz, GmbH Effeitrich, Germany) with Binos gas analyser. The temperature, light and air-supply conditions are as described in Chapter 3. Gas exchange parameters were calculated using DIAGAS software supplied by Walz.

2.14 MEASUREMENT OF TOTAL LEAF MALATE CONTENT

A crude extract of *K. fedtschenkoi* leaf disks was prepared by crushing 7 disks in a garlic press. The malate content of the leaf disk extract was determined enzymatically using malate dehydrogenase in a spectrophotometric assay at 340 nm by monitoring the oxidation of NADH. All leaf malate content measurements were carried out only once.

Chapter Three

THE REGULATION OF PHOSPHOENOLPYRUVATE CARBOXYLASE FROM *KALANCHOË FEDTSCHENKOI* BY PHOSPHORYLATION

3.1 INTRODUCTION

The malate sensitivity of PEPc from the CAM species *Kalanchoë (Bryophyllum) fedtschenkoi* is controlled by a circadian oscillator (Nimmo et al., 1984; Nimmo et al., 1987). Rhythmic changes are also seen in PEPc kinase activity and translatable mRNA (Carter et al., 1990; Hartwell et al., 1996). Protein synthesis inhibitors block the circadian appearance of PEPc kinase activity and translatable mRNA in *K. fedtschenkoi* (Carter et al., 1991; Hartwell et al., 1996) implying that two protein synthesis steps exist between the circadian clock and the appearance of the kinase protein. The RNA synthesis inhibitor, cordycepin, also blocked the nocturnal increase in PEPc kinase mRNA in *K. fedtschenkoi* (Hartwell et al., 1996). However, at present nothing else is known about the transduction pathway in CAM species between the circadian oscillator and the PEPc kinase RNA.

In both C₄ and C₃ species, light activates PEPc kinase activity via increasing the levels of the kinase mRNA (Hartwell et al., 1996). In C₄ plants, the light-induced change in PEPc activity has been linked to the light-mediated cytosolic alkalinization and subsequent release of Ca²⁺ from the vacuole (Giglioli-Guivarc'h et al., 1996; Pierre et al., 1992). In C₃ barley leaf protoplasts kept in darkness, PEPc was activated by treatment with both acid and a weak base (Lillo et al., 1996).

In consideration of the evident involvement of pH in the regulation of PEPc from C_4 and C_3 species, the effect of cytosolic pH on CAM PEPc was investigated in an attempt to elucidate more of the CAM signal transduction pathway affecting PEPc activity. The pH work in C_4 and C_3 species had used protoplasts but previous work had shown that protoplasts from the CAM species *K. fedtschenkoi* were too fragile for use (P.J. Carter, unpublished), presumably due to the large size of the CAM leaf vacuole. Disks cut from the leaves of *K. fedtschenkoi* were therefore used as the experimental system for this particular research.

3.2 RESULTS

3.2.1 Cytosolic alkalinization has no affect on the malate sensitivity of PEPc and PEPc kinase in *K. fedtschenkoi* leaf disks

Disks of diameter 1.5 cm were cut from *K. fedtschenkoi* leaves in the middle of the day of an 8-hour day $(28^{\circ}C)/16$ -hour night $(15^{\circ}C)$ growth regime (section 2.2.2(ii)). The disks were incubated in 90 mm petri dishes containing distilled water and transferred to fresh water after 1 and 2 hours incubation. A previous experiment had shown that when *K. fedtschenkoi* disks were incubated in distilled water over a period of 2 hours, the conductivity of the water increased suggesting that malate and other cell contents had diffused out of the damaged cells into the incubation solution. Transferring the disks into a petri dish containing fresh distilled water ("washing") after 15, 45 75 and 120 minutes helped to reduce the conductivity of the incubation solution. All control disks in subsequent experiments were therefore washed for 2 x 1 hours. As judged by enzymatic assay, no malate is released after these washes.

At the beginning of the dark period the disks were transferred either to fresh water or to ammonium chloride (NH₄Cl) (10 mM, 20 mM or 50 mM). Samples were then taken after 0, 3, 6 and 8 hours of darkness. Fourteen disks were used for each NH₄Cl concentration and incubation time. Disks were blotted dry, and desalted extracts made (section 2.5). The PEPc extracted from disks incubated in water had an apparent K_i for L-malate of 0.4 mM (section 2.6.1) at the beginning of the dark treatment which rose steadily to a value of 1.4 mM at the end of the 8 hour period. The NH₄Cl did not have any affect on the increase of the apparent K_{of} PEPc for malate compared with controls.

Thus when assessed by the malate sensitivity of PEPc, there is no evidence that NH_4Cl increases the level of PEPc kinase activity in CAM species as it has been reported to do in C₄ species. Therefore, the effect o*lowering* the cytosolic pH was investigated.

3.2.2 Cytosolic acidification prevents the nocturnal increase in the apparent K_i of PEPc for L-malate and PEPc kinase activity in *K. fedtschenkoi* leaf disks

Leaf disks were prepared from *K. fedtschenkoi* just before the light phase began at which time PEPc is dephosphorylated (Nimmo et al., 1984; Carter et al., 1991). Disks were incubated in 90 mm petri dishes containing water, 0.1 mM acetic acid, 1.0 mM acetic acid and 3.0 mM acetic acid for 16 hours, that is, until the middle of the following dark

phase. Fourteen disks were again used for each condition of this experiment and all subsequent experiments using leaf disks. PEPe activity was then extracted from the disks, assayed and the malate sensitivity determined. The activity of PEPe kinase was measured in the same extracts.

194

Increasing the concentration of acetic acid had the effect of lowering the apparent K_i of PEPc in the *K. fedtschenkoi* leaf disk extracts (Figure 3.1A). This suggested that the PEPc was in a low-phosphorylated state after this treatment i.e. that the acid treatment prevented the usual nocturnal phosphorylation of PEPc. In almost all other situations tested to date, the phosphorylation state of PEPc has reflected the activity of PEPc kinase. Disk extracts were therefore assayed for PEPc kinase activity (section 2.6.2). Figure 3.1B demonstrates that there was indeed a decrease in kinase activity in the acid-treated disks, consistent with the increased sensitivity of the enzyme to feedback inhibition by malate.

This preliminary experiment showed that acctic acid treatment reduced the rise in PEPc kinase activity in *K. fedtschenkoi* disks at night and consequently reduced the apparent K_i of PEPc for malate. The effects of acid treatment of disks were therefore examined in more detail.

3.2.3 Optimisation of the incubation conditions

The apparent K_i values observed for PEPc extracted from the *K. fedtschenkoi* disks in Figure 3.1 were low compared to those previously seen in leaves at similar time points, e.g. 3 mM in leaves in the middle of the night (Nimmo et al., 1984) compared to about 1.5 mM in disks. It was therefore decided to compare leaf and disk samples from the same plants.

Intact leaves were detached at the start of the photoperiod and allowed to take up water through the petiole. Disks were cut from the same plants at the same time and incubated in water. Samples were taken every 2 hours for 10 hours into the dark period. The apparent K_i values obtained for the leaves and disks at 10 hours were approximately 2.35 mM and 1.43 mM respectively (Figure 3.2A). This result confirms that the PEPc in disks docs become phosphorylated in the dark but presumably to a lower extent than in leaves.

Figure 3.1 The effect of acetic acid on the malate sensitivity of PEPc and PEPc

kinase activity of K. fedtschenkoi leaf disks

Leaf disks of *K. fedtschenkoi* were incubated in different concentrations of acetic acid from the beginning of the day period to the middle of the night period, a total of 16 hours. Desalted extracts of disks were made and assayed for the malate sensitivity of PEPc and PEPc kinase activity.

(A) Graph used to determine the apparent K_i of PEPc for L-malate (i.e. L-malate concentration at 50% maximum PEPc activity).

The different treatments are represented thus: water (\blacklozenge); 0.1 mM acetic acid (\blacksquare); 1.0 mM acetic acid (\blacklozenge); 3.0 mM acetic acid (\blacklozenge).

(B) Phosphorimage showing the PEPc kinase activity in desalted extracts from different disk samples, expressed as a percentage of the value obtained with the kinase control: Lane 1 - water

Lane 2 - 0.1 mM acetic acid

Lane 3 - 1.0 mM acetic acid

- Lane 4 3.0 mM acetic acid
- Lane 5 kinase control (partially purified maize PEPc kinase)







B

%

Α



77

Figure 3.2 The malate sensitivity of PEPc and PEPc kinase activity of K. *fedtschenkoi* leaves and leaf disks at night

Intact leaves and leaf disks were taken from K. *fedtschenkoi* plants at the beginning of the day. Leaves were allowed to take up water through the petiole and disks were incubated in water. Samples were taken at different times during the night. Desalted extracts of leaves and disks were assayed for the malate sensitivity of PEPc and PEPc kinase activity.

(A) Graph showing the apparent K_i of PEPc for L-malate in disks (\blacklozenge) and leaves (\blacksquare) throughout a night period time course.

(B) Phosphorimage showing the PEPc kinase activity in desalted extracts from leaves(i) and disks (ii), expressed as a percentage of the value obtained with kinase control:

- Lane 1 0 hours
- Lane 2 3 hours
- Lane 3 6 hours
- Lane 4 8 hours
- Lane 5 10 hours

Lane 6 - kinase control (partially purified maize PEPc kinase)



B(i)



B(ii)



%

%

The PEPc kinase activity in leaves and disks increased in a manner consistent with the observed changes in malate sensitivity (Figure 3.2B).

Thus there appeared to be a significant difference between *K. fedtschenkoi* leaves and disks in the apparent K_i values of PEPc for L-malate and PEPc kinase activities. In a further experiment disks were shaken gently (approximately 75 rpm) during incubation with no lids covering the petri dishes to ensure adequate aeration of the incubation solution. Leaves and disks were again taken at the beginning of the light period. Samples were taken for the first 8 hours of the dark period. Although there was a slight delay in the increase of apparent K_i from the disk samples, by 8 hours darkness the disks shaken gently in water and the detached leaves had apparent K_i values for PEPc of 2.2 mM and 2.65 mM L-malate respectively (Figure 3.3). Shaking the disks during incubation appeared to allow the apparent K_i of PEPc for L-malate to increase quite significantly, to a level comparable to that of leaves. All subsequent experiments therefore involved the aeration of disk incubation medium in this way.

3.2.4 Acetic acid affects PEPc via the acidification of cytosolic pH

It is possible that the effect seen with acetic acid is actually metabolic i.e. derived from the metabolism of acetate, rather than a consequence of reduction in cytosolic pH. Lipid soluble weak acids/bases can be used to manipulate cytosolic pH whereas strong acids, which are fully dissociated, are not lipid permeable and so are unable to alter the cytosolic pH. To investigate the nature of the effect of acetic acid on PEPc, disks were cut at the beginning of the day and put into water, 3 mM acetic acid, 3 mM propionic acid or approximately 1 mM HCl. Acetic acid and propionic acid are both weak acids but their uptake would affect metabolites in different ways. HCl is a strong acid. The pH of each was approximately 3.70. The samples were incubated until the middle of the following night when desalted extracts were made and the malate sensitivity of PEPc assayed. Figure 3.4A shows the malate sensitivitics obtained.

It can be seen that 3 mM propionic acid and 3 mM acetic acid both decrease the dark apparent Ki value as previously observed with acetic acid. However, 1 mM HCl has no affect on the malate sensitivity of PEPc when compared to the control.

Figure 3.3 The effect of shaking on the malate sensitivity of PEPc from *K. fedtschenkoi* disks and leaves at night

Intact leaves and leaf disks were taken from *K. fedtschenkoi* plants at the beginning of the day. Leaves were allowed to take up water through the petiole and disks were shaken gently in water and left uncovered to facilitate aeration. Samples were taken at different times during the night and desalted extracts assayed for the malate sensitivity of PEPc.

The graph below shows the apparent K_i of PEPc for L-malate in disks (\blacklozenge) and leaves (\blacksquare) throughout a night period time course.



hours darkness

Figure 3.4 The effect of different acids on the malate sensitivity of PEPc from *K. fedtschenkoi* disks

Leaf disks were cut from *K. fedtschenkoi* plants at the beginning of the day period and incubated in different acid solutions until the middle of the following night period. Desalted extracts of the disks were made and assayed for the malate sensitivity of PEPc and PEPc kinase activity.

(A) Graph used to determine the apparent K_i of PEPc for L-malate i.e. L-malate concentration at 50% PEPc activity. The different treatments are represented thus: water (\blacklozenge); 3 mM acetic acid (\blacksquare); 3 mM propionic acid (-); 1 mM hydrochloric acid (\blacklozenge).

(B) Phosphorimage showing the PEPc kinase activity in desalted extracts from different disk samples, expressed as a percentage of the value obtained with the kinase control:

Lane 1 - water

Lane 2 - 3 mM acetic acid

Lane 3 - 3 mM propionic acid

Lane 4 - 1 mM hydrochloric acid

Lane 5 - kinase control (partially purified maize PEPc kinase)





L-malate concentration (mM)

B

A

PEPc activity (% of maximum)

%



83

The samples were also assayed for PEPc kinase activity (Figure 3.4B). Although in this experiment the PEPc kinase activities are low, the results suggest that PEPc kinase is still present in the control and HCl- treated disks but reduced in the acetic acid- and propionic acid-treated disks. These results therefore suggest that the effect of acetic acid is not metabolic but rather is due to cytosolic acidification.

3.2.5 Time course of the effect of 3 mM acetic acid

In the work described above, treatment of leaf disks with 3 mM acetic acid over the course of a normal day prevented the usual rise in apparent K_i for malate (i.e. phosphorylation) of PEPc in the following night. To assess the duration of treatment with acetic acid that was required to give this effect, a time course experiment was performed. Disks were cut as the light period was beginning and put into either water or 3 mM acetic acid. At midday, 3 pm and 6 pm disks were transferred from the water into 3 mM acetic acid, All samples were shaken until the middle of the night when disks were removed from their incubation medium, blotted dry and snap frozen. The PEPc activity was then extracted from the disks, assayed and the malate sensitivity determined. Figure 3.5A shows that the apparent K_i value decreased with increasing length of 3 mM acetic acid treatment. However, even treatment with acetic acid from midday onwards had much less effect on the malate sensitivity of PEPc than treatment from 8 am onwards. Extracts were assayed for PEPc kinase activity and the results were consistent with the apparent K_i values obtained (Figure 3.5B). It can be seen that when disks were transferred to acetic acid at 6 pm, i.e. 2 hours after the dark period had begun, the kinase activity extracted is actually greater than that extracted from the control (Figure 3.5B; lanes 1 and 9), RNA was also extracted from the disks (section 2.10.2(ii)) and this was translated in vitro (section 2.10.4). Translation products were then assayed for kinase activity and results corrected for the efficiency of each in vitro translation as determined by [³⁵S]methionine incorporation. The control RNA (Figure 3.5 IC lane 1) translated much less efficiently than the other samples which explains the apparent incongruity between the value given for the % kinase activity and the actual intensity of the band apparent from the phosphorimage (compare with Figure 3.5 IIC). The RNA isolation from the intact leaf at midnight (Figure 3.5 IC lane 7) was carried out by Dr J. Hartwell

Figure 3.5 The effect of acetic acid treatment length on the malate sensitivity of PEPc, PEPc kinase activity and level of PEPc kinase translatable mRNA

Leaf disks were cut from *K. fedtschenkoi* plants at the beginning of the day period and put into either water or 3 mM acetic acid. At midday, 3 pm and 6 pm some disks were transferred from the control (water) incubation into 3 mM acetic acid. All samples were incubated until the middle of the night. Desalted extracts of the disks were assayed for the malate sensitivity of PEPc and PEPc kinase activity. RNA extracted from the disks was translated *in vitro* and assayed for PEPc kinase activity. Data from two separate experiments are shown in Figure 3.5I and 3.5II.

(A) Graph used to determine the apparent K_i of PEPc for L-malate i.e. L-malate concentration at 50% PEPc activity. The different treatments are represented thus: control 8 am (\blacklozenge); acetic acid 8 am (\blacksquare); acetic acid midday (\blacktriangle); acetic acid 3 pm (\blacklozenge); acetic acid 6 pm (\blacklozenge).

(B) Autoradiograph showing the PEPc kinase activity in desalted extracts from different disk samples, expressed as a percentage of the highest activity (lane 9) disregarding the control samples (lanes 11 - 13):

Lane 1 - control 8 am

Lane 2 - " without PEPc

Lane 3 - acetic acid 8 am

Lane 4 - " without PEPc

Lane 5 - acetic acid midday

Lane 6 - " without PEPc

- Lane 7 acetic acid 3 pm
- Lane 8 " without PEPc
- Lane 9 acetic acid 6 pm
- Lane 10 " without PEPc
- Lane 11 mid-dark leaf
- Lane 12 " without PEPc

Lane 13 - kinase control (partially purified maize PEPc kinase)

(C) Autoradiograph showing the PEPc kinase activity of *in vitro* translated mRNA from different disk samples, expressed as a percentage of the highest activity (lane 4) disregarding the control samples (lanes 6 - 8). The intensities of the bands have been corrected for efficiency of [³⁵S] methionine incorporation:

Lane 1 - control 8 am

- Lane 2 acetic acid 8 am
- Lane 3 acetic acid midday
- Lane 4 acetic acid 3 pm
- Lane 5 acetic acid 6 pm
- Lane 6 no RNA
- Lane 7 RNA from K. fedtschenkoi leaves in middle of night
- Lane 8 kinase control (partially purified maize PEPc kinase)
The effect of acetic acid treatment length on the malate sensitivity of PEPc from K. fedtschenkoi leaf disks





PEPc activity (% of maximum)

IB



IC

%



IA



kinase activity and the actual intensity of the band apparent from the phosphorimage (compare with Figure 3.5 IIC). The RNA isolation from the intact leaf at midnight (Figure 3.5 IC lane 7) was carried out by Dr. J. Hartwell using different plants which explains the apparently much greater level of mRNA. The level of PEPc kinase translatable mRNA was found to decrease with length of acetic acid treatment (Figure 3.5C). It can therefore be seen that 3 mM acetic acid affects the level of the PEPc kinase translatable RNA to decrease the kinase activity and the consequent sensitivity of PEPc to malate inbibition determined by the phosphorylation state of the PEPc protein. However, it would appear that disks have to be treated for at least 9-12 hours before the effect is noticeable i.e. disks transferred to acetic acid at 8 am are much more affected by the treatment than those transferred at 3 pm or later. In one experiment disks transferred at midday were similar to those transferred at 3 pm; in the other they resembled disks transferred at 8 am.

3.2.6 Cytosolic acidification prevents decarboxylation of intracellular malate

The total concentration of malate in the disks was also determined throughout the time course (see section 2.14). Figure 3.6 shows that just before the lights came on the malate concentration was high due to its accumulation in the cell vacuole throughout the night. By the end of the day malate levels in the control sample were very low as the malate had been decarboxylated by the activity of malic enzyme. Throughout the following night period the malate levels rose again. However, in the samples treated with acetic acid from 8 am the decline in malate by 4 pm was reduced and the subsequent rise in malate throughout the following dark period was blocked. With the later acetic acid treatments i.e. from midday, 3 pm or 6 pm, when malate had already fallen (nearly as far as the control), the treatment with acetic acid slightly reduced but did not prevent the subsequent rise in malate content.

It appears that one effect of acctic acid is to reduce the decarboxylation during the day of the malate that accumulates in the cell vacuole throughout the night. This raised the question of whether treatment with acetic acid would affect the apparent K_i of PEPc for L-malate in *K. fedtschenkoi* disks if there was very little malate at the beginning of the time course. This was investigated by detaching leaves from plants at the beginning of the night and maintaining them in CO₂-free air throughout the night

Figure 3.6 The effect of acetic acid treatment length on the intracellular malate concentration of *K. fedtschenkoi* leaf disks

Leaf disks were cut from *K. fedtschenkoi* plants at the night/day transition. Disks were put into either water or 3 mM acetic acid. At 12 noon, 3 pm and 6 pm some disks were transferred from the control (water) incubation into 3 mM acetic acid. Samples were taken at 8 am (the beginning of the incubations), 4 pm, 12 midnight and 7 am the following morning and the tissue malate concentration determined.

The graph opposite shows the malate concentration at the different points of the time course thus: 8 am (\blacksquare); 4 pm (\blacksquare); 12 midnight (\blacksquare); 7 am (\blacksquare).

The effect of acetic acid treatment length on the malate concentration in K. fedtschenkoi leaf disks







06

L-malate concentration (MM) (section 2.13). As only respiratory CO_2 is available for primary fixation by PEPc throughout the night there should be much less malate present in the leaves the following morning. Disks were then cut from these CO2-deprived leaves and from control leaves the next morning and incubated throughout the day in water or 3 mM acetic acid. The malate content of the disks was determined at 8 am (i.e. at the end of the night incubation in CO₂-free air), 4 pm and 12 midnight. In the middle of the following night period, desalted extracts of the disks were also made and assayed for the malate sensitivity of PEPc. The CO2-free air treatment reduced but did not completely eliminate malate accumulation during the night (Figure 3.7). By the end of the following day the malate levels in disks cut from these CO₂-deprived leaves and incubated in either water or 3 mM acetic acid had dropped further to a level similar to or just greater than the control disks and then rose, although by a lesser extent than the control disks, by the middle of the following night. Therefore, nocturnal treatment with the CO₂-free air seemed to eliminate the effect of acetic acid on intracellular malate levels. However, when the malate sensitivity of the PEPc in the disks was determined, it was seen that the CO_2 -free air treatment did not alter the effect of the 3 mM acetic acid on the dark apparent K_i of PEPc for malate (Figure 3.8). Hence it appears that the disks can accumulate at least some malate even though PEPc remains dephosphorylated.

3.2.7 PEPc inhibitors fail to prevent dark CO₂ fixation

The PEPc inhibitors DCDP and DMDP were employed in an attempt to further reduce the levels of intracellular malate at the end of a night by using them in conjunction with CO $_2$ -free air.

To test the effects of these compounds on PEPc itself, purified *K. fedtschenkoi* PEPc was diluted 30-fold in 50 mM Tris-IICl, pH 7.8 and 1 mM DTT and assayed in the presence of increasing concentrations of each inhibitor. From Figure 3.9 it can be seen that DCDP is over 10 times more potent than DMDP in the inhibition of PEPc, with apparent apparent K_i values of approximately 40 μ M and 575 μ M respectively at 2 mM PEP.

The inhibition of purified PEPc by DMDP in the presence of 0.2 mM PEP was also studied. The inhibition of PEPc by DMDP was approximately 10 times greater at the

Figure 3.7 The effect of CO₂-free air and 3 mM acctic acid on the intracellular malate concentration of *K. fedtschenkoi* leaf disks

Intact leaves were incubated in CO_2 -free air throughout the night. Disks were cut from these leaves and from control leaves next morning and incubated in water or 3 mM acetic acid. Samples were taken at different time points and the tissue malate concentrations determined.

The graph opposite shows the malate concentration of disks sampled: 8 am (\blacksquare); 4 pm (\blacksquare) and 12 midnight (\blacksquare).

The effect of CO₂-free air and 3 mM acetic acid on the malate content of K. fedtschenkoi disks



L-malate concentration (mM) water + CO₂-free air acetic acid control acetic acid + CO₂-free air water control

treatment

Figure 3.8 The effect of CO₂-free air and 3 mM acetic acid on the malate sensitivity of PEPc from *K. fedtschenkoi* leaf disks

Intact leaves were incubated in CO_2 -free air throughout the night. Disks were cut from these leaves and control leaves next morning and incubated in water or 3 mM acetic acid until the middle of the following night period. Desalted extracts of the disks were made and assayed for the malate sensitivity of PEPc.

The graph below shows the sensitivity of PEPc to malate in the desalted extracts from the different disk treatments: water control (\blacklozenge); water + CO₂-free (\blacksquare); acetic acid control (\blacklozenge); acetic acid + CO₂-free (\blacksquare).



PEPc activity (% of maximum)

L-malate concentration (mM)

Figure 3.9 The effect of DCDP and DMDP on the activity of purified K. *fedtschenkoi* PEPc

Purified PEPc from *K. fedtschenkoi* was diluted 30-fold in 50 mM Tris-HCl, pH 7.8 and 1 mM DTT and assayed in the presence of different concentrations of the PEPc inhibitors DCDP and DMDP.

The graph below was used to determine the apparent K_i of PEPc for each of the inhibitors i.e. the inhibitor concentration at 50 % PEPc activity: DCDP (\blacklozenge); DMDP (\blacksquare).



PEPc inhibitor concentration (µM)

lower substrate concentration of 0.2 mM giving an apparent K_i of 40 - 50 μ M compared with 575 μ M at 2 mM PEP (results not shown).

A desalted extract of *K. fedtschenkoi* leaves harvested in the middle of the day was also prepared and assayed for PEPc activity in the presence of increasing concentrations of each inhibitor. PEPc in the desalted extract had an apparent K_i for DCDP (40 - 50 μ M) similar to that seen with purified PEPc and an apparent K_i of approximately 300 μ M for DMDP, slightly lower than the value for purified PEPc (results not shown).

K. fedtschenkot leaves were detached from plants at the end of a day period and allowed to take up either 2 mM DCDP or 10 mM DMDP in constant darkness, 15° C and a stream of CO₂-free air. CO₂ output by the leaves was monitored by infra red gas analysis (section 2.13). As shown by Wilkins (1959; 1962), control leaves show a persistent rhythm of CO₂ output that lasts for several days. This phenomenon has been attributed to oscillations in flux through PEPc. The troughs in output correspond to fixation of respired CO₂ by PEPc. The peaks in output correspond to escape of respired CO₂ from the leaf when PEPc is inactive.

The results (Figure 3.10) show that neither inhibitor reduced CO_2 fixation by the leaves, even though the leaves did take up the inhibitors. Hence it was not possible to prevent CO_2 assimulation and malate accumulation by the use of these inhibitors.

3.2.8 The involvement of light in the effect of 3 mM acetic acid on the malate sensitivity of PEPc and malate content in *K. fedtschenkoi* disks

The present work has demonstrated that cytosolic acidification, in response to acetic acid treatment in *K. fedtschenkoi* disks, leads to a reduction in both malate content and the malate sensitivity of PEPc. This latter effect is a consequence of the reduction in the level of PEPc kinase translatable mRNA and PEPc kinase activity.

Light-induced cytosolic alkalinization has been shown to be involved in the lightinduced change in PEPc activity observed in C_4 plants (Giglioli-Guivarc'h et al., 1996). Is it therefore possible that the effect of acetic acid observed thus far might be dampened by the counteraction of light-induced cytosolic alkalinization in the *K*. *fedtschenkoi* disks? To investigate this possibility, disks were cut from leaves just before the day began and shaken in water or 3 mM acetic acid in the normal

Figure 3.10 The effect of the PEPc inhibitors DCDP and DMDP on the circadian rhythm of CO_2 output in *K. fedtschenkoi* leaves maintained in CO_2 -free air and constant darkness at 15[°]C

Detached K. fedtschenkoi leaves were placed in CO_2 -free air and constant darkness at 15°C and the CO_2 output of the leaves was monitored using an infra-red gas analyser. The data from the experiment are shown in the graph opposite: 10 mM DMDP (\blacklozenge); water (\blacksquare); 2 mM DCDP (\blacktriangle).





CO₂ output (μg/h/g fresh weight)

time (hours)

86

photoperiod either uncovered or in a light-tight box (continuous darkness and 28°C/15°C as normal growth regime, see section 3.2.1) until the middle of the following dark period. The malate content of the disks was determined at different points in the incubation and desalted extracts of the disks in the middle of the night were made and assayed for malate sensitivity of PEPc. From Figure 3.11 it can be seen that the disks incubated in water and continuous darkness gave a malate sensitivity greater than the light control and similar to the light/acetic acid-treated disks and the dark/acetic acid-treated disks.

The malate concentration of the disks at different points in the incubation is shown in Figure 3.12. The disks incubated under normal light conditions, and in water and acetic acid, behaved as previously. The malate concentration of disks incubated in continuous darkness in both water and acetic acid decreased between the start of the incubation at 8 am and 4 pm, although to a lesser extent than both light samples, and less in the dark acetic acid-treated disks than the dark control disks. In both dark samples the malate levels then continued to decrease until 12 midnight when they had reached a similar concentration. Incubation of disks in continuous darkness therefore seems to reduce the decarboxylation of malate even further than treatment with acetic acid alone.

3.3 DISCUSSION

Cytosolic pH affects the activity of both PEPc and PEPc kinase (Andreo et al., 1987; Rajagopalan et al., 1993; Wang and Chollet, 1993; Echevarria et al., 1994) and has been shown to be an important factor during the light activation of PEPc in mesophyll cells of C₄ plants (Devi and Raghavendra, 1992; Pierre et al., 1992; Raghavendra et al., 1993; Yin et al., 1993). *In situ* evidence for the involvement of cytosolic alkalinization in the light-activation of PEPc has also implicated increases in $[Ca^{2+}]$ which may modulate an upstream protein kinase (Pierre et al., 1992; Giglioli-Guivare'h et al., 1996; Vidal et al., 1996). The pH optimum of PEPc is approximately pH 8 (O'Leary, 1982 and references therein) and so an increase in cytosolic pH could directly enhance PEPc activity or lead to an increase in PEPc kinase activity and the subsequent phosphorylation state of PEPc. As comparatively little was known about

Figure 3.11 The effect of light and 3 mM acetic acid on the malate sensitivity of PEPc in *K. fedtschenkoi* disks

Leaf disks were cut from *K. fedtschenkoi* plants at the beginning of the day period and put into either water or 3 mM acetic acid and then incubated in the normal photoperiod and left uncovered or put in a light-tight box (continuous darkness) until the middle of the following night. Desalted extracts of the disks were assayed for the malate sensitivity of PEPc.

The graph below shows the sensitivity of PEPc to malate in the desalted extracts from the different disk treatments: light/control (\blacklozenge); light/acetic (\blacksquare); dark/acetic (\blacktriangle); dark/control (\blacklozenge).



L-malate concentration (mM)

Figure 3.12 The effect of light and 3 mM acetic acid on the intracellular malate concentration of *K. fedtschenkoi* leaf disks

Leaf disks were cut from *K. fedtschenkoi* plants at the beginning of the day period and put into either water or 3 mM acetic acid. The disks were then incubated in the normal photoperiod and left uncovered or put in a light-tight box (continuous darkness) until the middle of the following night. Samples were taken at different time points and the malate concentration determined.

The graph opposite shows the malate concentration of disks sampled at 8 am (\blacksquare) ; 4 pm (\blacksquare) and 12 midnight ().

The effect of light and 3 mM acetic acid on the malate concentration in K. fedtschenkoi leaf disks



L-malate concentration (mM) treatment

102

the involvement of pH in the regulation of PEPc in CAM species, attempts were made to manipulate the cytosolic pH of *K. fedtschenkoi* disks and the effects studied.

Treatment of K. fedtschenkoi disks with NH4Cl had no affect on the malate sensitivity of PEPc (section 3.2.1). Cytosolic alkalinization therefore does not appear to affect PEPc kinase in the CAM species as it has been reported to do in C₄ species (Pierre et al., 1992; Giglioli-Guivarc'h et al., 1996). In leaf protoplasts of the C₃ species barley, treatment with the weak base methylamine in darkness resulted in activation of PEPc (Lillo et al., 1995). Suprisingly, treatment with the acid DMO had a similar effect. When K. fedtschenkoi disks were incubated in 3 mM acetic acid, the nocturnal decrease in the malate sensitivity of PEPc was prevented (Figure 3.1A). A time course of the effect of acetic acid on the malate sensitivity of PEPc showed the effect to be slow, requiring at least 9-12 hours treatment (Figure 3.5A). This is in contrast to the activation of barley leaf protoplast PEPc by methylamine and DMO which was apparent after 40 minutes. Although the effect of acetic acid is too slow to be presumed significant in signal transduction affecting CAM PEPc, it caused a reduction in the level of translatable PEPc kinase mRNA (Figure 3.5C), a decrease in PEPc kinase activity (Figure 3.5B) and ultimately a reduced phosphorylation state of PEPc detectable as a low K (of PEPc for L-malate (Figure 3.5A).

The comparison of treatment with acetic acid, propionic acid and HCl on the malate sensitivity of PEPc (Figure 3.4A) suggests that the effects of acetic and propionic acid arc modulated by changes in intracellular pH as opposed to metabolic effects. Even though the H⁺ concentration of the three acids is the same in the incubations no effect is seen with HCl. This can be explained by the pK₀ values of the acids. Acetic acid and propionic acid have pK₀ values of 4.75 and 4.87 respectively and so are both weak acids. This means that in their protonated forms the acid molecules are able to cross the plasma membrane lipid bilayer and dissociate in the cell cytosol, thus affecting intracellular pH. However, HCl is a strong acid and in its fully dissociated state is unable to cross the cell plasma membrane to affect intracellular pH. More direct evidence that the acetic acid effect is a result of pH changes could be provided by fluorescent dyes which could confirm the changes in cytosolic pH. However, this would require the use of protoplasts and the difficulty of obtaining protoplasts from *K. fedtschenkoi* has already been referred to in section 3.1.

2

ò

Physiologically significant increases in PEPc activity (e.g. during the night in CAM plants) are thought to increase the cytosolic malate pool. The importance of phosphorylation of PEPc is to reduce the enzyme's sensitivity to inhibition by malate and allow CO_2 fixation in these conditions. However, it is known that malate inhibits PEPc kinase activity (Carter et al., 1991) and more recently (Borland et al., 1999), malate has been suggested to be a feedback inhibitor of PEPc kinase transcription. It was therefore of interest to measure the intracellular concentration of malate in *K*. *fedtschenkoi* disks treated with acetic acid.

The data suggested that acetic acid prevents the decarboxylation of malate (Figure 3.6). Disks put into 3 mM acetic acid at the end of a night period (i.e. 8 am) have a greater concentration of malate at the end of the day (4 pm) than the control disks incubated in water. The accumulation of malate in the subsequent dark period attributed to PEPc activity is prevented in the acid-treated disks. The malate sensitivity of PEPc in the middle of the dark period (12 midnight) increases with the length of acetic acid treatment i.e. the apparent K_i for malate decreases (Figure 3.5A). This seems to be due to a reduction in the expression of PEPc kinase as judged both by kinase activity (Figure 3.5B) and by kinase mRNA levels (Figure 3.5C). One possible explanation of these results is inhibition of PEPc kinase transcription by the high levels of malate present in the acid-treated disks at the end of the day. The PEPc would therefore not become phosphorylated during the dark period and would also be inhibited by the levels of malate present and unable to catalyze any further production of malate. The malate that accumulates at night during CAM is pumped into the cell vacuole. Presumably this doesn't occur in the acid-treated disks. It is conceivable that the cytosolic pH is so low that the leaves are unable to generate a big enough pH gradient across the tonoplast to pump malate; the low pH might also prevent the H⁺ symport necessary for the activity of the vacuole H⁺ATPase that pumps the malate into the vacuole. In addition, it seems that the activities of NAD-malic enzyme and/or NADP-malic enzyme are reduced when the cytosolic pH is low, thus inhibiting decarboxylation of malate. Further studies of these two enzymes would be valuable. This explanation implies that the failures to decarboxylate malate and induce PEPc kinase are related. However it is also possible that a decline in cytosolic pH reduces expression of PEPe kinase irrespective of malate. It is not possible to distinguish between these at the present time.

Incubating leaves in CO₂-free air reduced the amount of malate accumulated throughout the night (Figure 3.7). The apparent mechanism by which acetic acid affects PEPc activity was thus made ineffective and the malate was able to accumulate during the following dark period to a similar extent as the control. This result nicely indicates that the effect of acetic acid is not a consequence of damage caused to the disks by the treatment. However, even with disks from leaves incubated in CO₂-free air, acetic acid prevented the nocturnal increase in the apparent K_i of PEPc for malate (Figure 3.8). One explanation of this result is that there is sufficient cytosolic malate left at the end of the day in these disks to inhibit PEPc kinase transcription (Borland et al., submitted) and thus the PEPc remains dephosphorylated and malate sensitive, but there is not enough malate to inhibit PEPc activity and therefore malate can be made in the following night (Figure 3.7). However, it seems suprising that PEPc could still function in the face of low pH and high cytosolic malate. In comparison acetic acid treatment alone results in higher levels of malate at the end of the day but the acid prevents re-uptake of the malate into the vacuole, and thereby blocks induction of PEPc kinase and phosphorylation of PEPc. Anderson and Wilkins (1989) reported the inhibition of the circadian rhythm of CO₂ fixation and a low malate content of K. fedtschenkoi leaves in response to high temperature (40°C) in continuous light. However, the leaves appeared to have a basal level of about 10 mM malate which apparently cannot be metabolized. It is possible that there is a threshold concentration of malate above which the transcription of PEPc kinase and PEPc activity will be inhibited. This could explain why in Figure 3.7 the malate level rises in the night following treatment of the K.fedtschenkoi disks with acetic acid and CO_2 -free air but not after treatment with acetic acid alone. Likewise in Figure 3.6 malate accumulates when acetic acid treatment of disks begins at or after noon but not before. From close inspection of Figure 3.6 and 3.7 it would appear that the proposed threshold malate concentration is about 20 mM. A significant part of this is presumably not metabolizable perhaps due to compartmentation.

The reduction in malate accumulated overnight was not complete using CO_2 -free air. However, even though the inhibition potential of DCDP and DMDP could be demonstrated *in vitro* (Figure 3.9), neither PEPc inhibitor was able to reduce the fixation of respired CO₂ further than CO₂-free air alone (Figure 3.10). The incubation of leaves in N_2 might have been more effective in lowering the malate levels than CO₂-free air but the extent to which the leaves would be perturbed by this treatment eliminated this as an attractive experimental alternative.

Another possible factor in the effect of acetic acid treatment was light. The acid and base treatment of barley leaf protoplasts mentioned above (Lillo et al., 1995) had been given during darkness. From Figure 3.11 it would seem that the only condition that leaves PEPc phosphorylated in the middle of the night is the light control. With the other three treatments the malate level is a lot higher at 4 pm (Figure 3.12) so presumably the kinase is not induced thus accounting for the greater sensitivity to malate of PEPc from these disks. The failure of disks in continuous darkness to accumulate malate in the following "night" (Figure 3.12) is exactly as predicted from previous unpublished work (G.A. Nimmo and H.G. Nimmo, personal communication). When K. fedtschenkoi leaves were placed in continuous darkness as the end of a night, i.e. when malate is high, there is no rhythm in CO_2 fixation, malate does not decline and PEPc does not become phosphorylated in the following "night". A rhythm is only observed if the constant conditions (continuous darkness, 15°C, CO_2 -free air) commence at the end of a day. This is presumably because malate is low under these conditions and the PEPc kinase can be induced. These findings all support the view that metabolites can over-ride circadian control. They are also consistent with the view that the circadian clock actually controls PEPc kinase via altering metabolites e.g. the primary target of circadian control could be vacuolar malate transport/permeability. However, the issue central to this discussion is the cytosolic malate concentration i.e. is there sufficient malate in the cytosol to inhibit PEPc kinase transcription and PEPc activity? It is difficult to measure cytosolic malate with any degree of accuracy because such a high proportion of total malate is in the vacuole. However, a rough estimate might be possible using ¹³C nuclear magnetic resonance (vacuolar and cytosolic malate would give separate signals because of the pH difference) or non-aqueous fractionation to give physical separation of the various compartments.

In summary then, acetic acid reduced the dark apparent K_i of PEPc for malate in *K*. *fedtschenkoi* disks incubated in a normal 8 hour photoperiod and reduced the decarboxylation of malate during the day. However, the effect of acetic acid was sufficiently slow to suggest that cytosolic acidification is *not* involved in the CAM PEPc signal transduction pathway as such but instead has a less direct role to play in its regulation through preventing the decarboxylation of malate.

In consequence, while further experiments could have been done to ascertain that the acid treatments did indeed affect cytosolic pH, it was felt that these should not be attempted since the change in pH did not seem to be part of a physiologically important signalling process.

Chapter Four

CLONING AND SEQUENCING OF FIRST FULL-LENGTH PHOSPHOENOLPYRUVATE CARBOXYLASE cDNA FROM ARABIDOPSIS THALIANA

4.1 INTRODUCTION

For the purposes of investigating the regulation of PEPc in C_3 species it was decided that it would be useful to obtain a full-length cDNA of the enzyme from *Arabidopsis thaliana*. Cloning of the cDNAs for different PEPc isoforms that are known to exist is essential for investigation of the tissue specificity of their expression. It is also possible in principle to use a clone to overexpress a protein and facilitate studies of the enzyme. The purification of enough *A. thaliana* PEPc protein to investigate its regulation would be difficult using conventional sources of the enzyme. A cloning approach was therefore employed in preference to protein purification in the isolation of the C_3 isoform of PEPc. This chapter describes the results obtained. Data on the expression of PEPc using a probe resulting from the work described here will be presented in Chapter 5.

4.2 RESULTS

4.2.1 Cloning of a full-length putative PEPc from an A. thaliana cDNA library

The *A. thaliana* var. Columbia cDNA library to be screened for the PEPc gene was λ PRL2, a library constructed in the vector λ ZipLox (see D'Alessio et al., 1992) and containing the plasmid pZL1. The cDNA was made by reverse transcription of an mRNA pool from four tissue types: i) roots ii) tissue culture iii) leaves and iv) germinating seedlings. The library was obtained from the Arabidopsis Biological Resource Center, Ohio State University. A search of the *A. thaliana* sequence database (http://genome-www.stanford-edu/Arabidopsis) using a *Z. mays* PEPc sequence found an EST (expressed sequence tag), 129G6T7, with good identity to the *Z. mays* PEPc sequence. The EST was a clone of the λ PRL2 cDNA library and was therefore obtained for use as a probe to screen the library.

A pair of PEPc-specific primers were synthesized from a sequence given by Honda et al. (1996):

e r

5' - GGCAAGCAGGAGGTCATGATCGG - 3'

5' - GAAGTACTCGACAAATCGAGGCTC - 3'

The region of the PEPc gene that these primers anneal to is shown diagrammatically in Figure 2.1 (section 2.9.7). Use of these primers had led to the amplification of a 0.5kb cDNA fragment from the RNA of *Aloe aborescens* leaves (a monocot CAM plant). The nucleotide sequence of this fragment showed a high degree of similarity to the conserved region of the ice plant, Sorghum and maize PEPc genes that the primer design was based upon (Cushman and Bohnert, 1989; Cretin et al., 1991; Izui et al., 1986). The *A. thaliana* PEPc EST 129G6T7 was approximately 1.6 kb and therefore because it was rather long to be used as a probe, the above primers were used in a PCR (section 2.9.5) with the cDNA (section 2.11.3) to amplify a 500 bp fragment of PEPc. The corresponding band was extracted from the gel using the QIAGEN gel extraction kit. The 500bp sequence from clone 129G6T7 was then radiolabelled using Amersham's Decaprime DNA labelling kit (section 2.8.2) and subsequently used to probe the λ PRL2 library for PEPc (section 2.11.5).

The λ PRL2 library had a lower titre than specified (section 2.11.2). The full library (approximately 150,000 pfu) was therefore screened by nucleic acid hybridization analysis using the 500 bp ³²P-dCTP labelled PEPc probe (section 2.11.4 - 2.11.7). Plaques resulting from each round of screening were immobilized onto nitrocellulose membrane (section 2.8.1) and probed with the radiolabelled PEPc fragment. Autoradiographs (section 2.7.3) of the membranes allowed identification of phage plaques with putative PEPc cDNA inserts. These were cored out and the phage eluted into SM buffer. A PCR was then performed on these phage using the phage primers Sp6 and T7 and the PCR products run out on a 1% agarose gel in TAE buffer (section 2.7.4). The DNA in the agarose gel was then transferred by capillary action onto another nitrocellulose membrane and this was also probed with the radiolabelled PEPc fragment and autoradiographed. This combined PCR/Southern blotting approach facilitated the identification of phage containing the putative full-length PEPc cDNA inserts i.e. approximately 3 - 3.5 kb. These phage were then titred and plated out as described above and the screening procedure repeated until a cDNA of the desired size had been isolated from a single plaque. Figure 4.1 shows the PCR products of the third round screen run on an agarose gel and then blotted and probed with the radiolabelled 500 bp

Figure 4.1 Hybridization of radiolabelled PEPc cDNA to PCR products from λ PRL2 cDNA library screen

A combined PCR/Southern hybridization analysis of putative positives from each round of the library screening facilitated the identification of phage containing putative full-length PEPc cDNA inserts i.e. approximately 3-3.5 kb.

(A) A photograph of the PCR products amplified from third round screen putative positives using the phage primers T7 and Sp6 and separated on a 1% agarose gel. The ethidium bromide-stained DNA fragments are visualised under UV light. The lanes are as follows:

M - 1 kb DNA ladder

Lanes 1-16 - third round screen putative positives 3.1-3.16 labelled 1-16 C - water control

(B) Autoradiograph of Southern hybridization of PCR products as shown in (A) with radiolabelled PEPc cDNA. The lanes are as follows:

Lanes 1-16 - third round screen putative positives 3.1-3.16 labelled 1-16

C - water control

P - 500 bp PEPc control



PEPc probe. In lanes 1 - 3 of Figure 4.1A (PEPc clones 3.1 - 3.3 respectively) a faint band of approximately 3 - 4 kb can be seen. The radiolabelled PEPc fragment then appeared to hybridize to these bands suggesting that they were PEPc fragments of the desired size (Figure 4.1B lanes 1 - 3). The radiolabelled PEPc probe hybridized to two bands in lanes 1 - 16 of Figure 4.1B. This indicates that the plaques from the third round of screening contain more than one colony of phage. After a further round of PCR/Southern blotting, a PEPc cDNA insert of approximately 3.5 kb was obtained in a plaque purified form (Figure 4.2 lane 2). An in vivo excision of the recombinant plasmid pKP42 (pZL1 containing PEPc clone 4.2) from the bacteriophage vector was then performed using E.coli strain DH10B and selection on NZYCM plates containing 100µg/ml ampicillin and 10mM MgCl₂. 5ml LB broth containing the antibiotic and magnesium were then inoculated with one of the colonies and grown to an approximate density of $\Lambda_{600} = 0.5$. A small scale plasmid preparation was performed on the culture using the QIAGEN mini-prep kit and some of the purified plasmid checked by agarose gel electrophoresis and a restriction digest (section 2.9.4) using Sall to linearize the recombinant plasmid (Figure 4.3). The restriction digest produced an approximately 8kb fragment on an agarose gel consistent with an approximately 3.5kb cDNA insert and a 4.08kb pZL1 plasmid.

4.2.2 Sequencing of the putative full length PEPC cDNA

Some of the recombinant plasmid pKP42 was denatured and then the single stranded DNA was sequenced (section 2.9.6) using the T7 Sequenase version 2.0 kit and the bacteriophage primers T7 and Sp6 (section 2.9.6). This allowed the sequencing of approximately 250 bases from both the 5'- and 3'- ends of the PEPc cDNA. The nucleotide sequence was read from the autoradiograph of the sequencing gel and translated in each of the three reading frames. A BLAST search was then performed on the DNA sequence databases available via the National Centre for Biotechnology Information website (http://www.ncbi.nlm.nih.gov). The search showed that the 5' end of the clone was very similar in sequence to PEPc, contained a possible start codon and the putative PEPc phosphorylation motif which is close to the N-terminal end of the protein. The 3' end contained a polyA tail but no similarity to other PEPc sequences. The size of the clone suggested it might contain a significant 3' untranslated region and

Figure 4.2 Purified phage containing putative full-length PEPc cDNA inserts

Four rounds of combined PCR/Southern hybridization analysis of putative positives from each round of the library screening resulted in the isolation of three purified phage containing putative full-length PEPc cDNA inserts. Below is an autoradiograph of the Southern hybridization of radiolabelled PEPc cDNA with PCR products amplified from fourth round screen putative positives using the phage primers T7 and Sp6. The lanes are as follows:

Lane 1-4 - fourth round screen putative positives 4.1 - 4.4 labelled 1- 4 C - water control

P - 500 bp PEPc control



Figure 4.3 Sal I digest of purified recombinant plasmid pKP42

An *in vivo* excision of the recombinant plasmid pKP42 containing PEPc cDNA clone 4.2 was performed and the plasmid purified. The integrity of the plasmid was checked by agarose gel electrophoresis and the plasmid then linearized by *Sal* I to size the putative full-length PEPc cDNA insert.

(A) Photograph of undigested pKP42 separated on a 1% agarose gel. The ethidium bromide stained DNA fragments were visualised under UV light. The lanes are as follows:

M - 1 kb DNA ladder P - undigested pKP42

(B) Photograph of *Sal* I digested pKP42 separated on a 1% agarose gel. The ethidium stained DNA fragments were visualised under UV light. The lanes are as follows:

M - 1 kb DNA ladder

D - Sal I digested pKP42



so it was possible that the 3' sequence read had not reached the PEPc coding region. The presence of a possible start codon and a polyA tail are indicative of a full length functional clone and so sequencing was continued.

As the clone to be sequenced was greater than 3 kb the remainder of the sequencing was carried out in collaboration with the Molecular Biology Sequencing Unit (MBSU) using an automated sequencer. Oligonucleotides primers internal to the 5' and 3' sequences were designed and synthesized and used to obtain further sequence. This procedure was repeated until both the sense and anti-sense strands of the cDNA had been completely sequenced twice.

Multiple alignments of all sequences were performed using GeneJockey II and finally used to produce a sequence contig of the full clone. When analysed the contig had an open reading frame (ORF) of 968 amino acid residues in frame +2. The multiple cloning site sequence was identified and this revealed the beginning and end of the 5' and 3' untranslated regions respectively. The interleaved format of the nucleotide and amino acid sequence for the PEPc clone from the λ PRL2 *A. thaliana* cDNA library is shown in Figure 4.4.

4.2.3 Sequence analysis of full-length *A. thaliana* PEPc cDNA

The PEPe clone has an open reading frame of 968 amino acid residues (M_r 124, 890 and an isoelectric point of 7.31) beginning at base number 116 of the full length sequence including 5' untranslated sequence and finishing at base number 3022. There are two ATG triplets in the PEPc encoding ORF that are candidates for translation initiation (Figure 4.4 nucleotide nos. 116 and 140). Two possible start codons have been found in the genomic and cDNA sequences of other plant PEPcs (Relie and Wild, 1996), many of which initiated translation from the first ATG. However, until the protein encoded by the *A. thaliana* PEPc is sequenced it is not possible to say with any certainty which site initiates translation. A sequence corresponding to the typical polyadenylation signal, ATAA, described for higher plants (Joshi, 1987) and reported in other PEPc cDNA clones (e.g. Honda et al., 1996), is found at nucleotide number 3107, 142 bp upstream of the poly A sequence. The conserved PEPc phosphorylation motif SIDAQ is highlighted in Figure 4.4, indicating serine¹¹ as the phosphorylated residue. The *A. thaliana* PEPc cDNA sequence was aligned with other known PEPc sequences using the Genejockey II

Figure 4.4 Interleaved nucleotide and amino acid sequence of a full-length PEPc cDNA from *A. thaliana*

The numbers down the left hand side of the sequence correspond to nucleotide number. The open reading frame starts at nucleotide number 116 and ends at nucleotide number 3022. The shaded box is the PEPc phosphorylation motif conserved among all known PEPc sequences to date. \bullet is the stop codon.

1					(CIGO	AGGI	ACCO	GTCC	GGAA	TTCC	CGGG	TCGA	CCCA	CGCG	TCCG	AAT
49	CCGC	CCCCAGAGATTTCTTCTTTTCAGAAGAAGTAAGAGGGTGGCGAAGAAGATTTGATTGA							ATA								
	Met	Ala	Gly	Arg	Asn	Ile	Glu	Lys	Met	Ala	Ser	Ile-	Asp	Ala,	Gľn	Leu	Λrg
116	ATC	GCG	\mathbf{GGT}	CGG	AAC	ATA	GAG	AAG	ATG	GCA	TCT	ATT.	GAT	GĊT	CAG	Ċтт	CGG
	Gln	Leu	Val	Pro	Ala	Lys	Val	Ser	Glu	Asp	Азр	$_{ m Lys}$	Leu	Val	Glu	Tyr	Asp
167	CAA	CTC	G'I'T	CCT	GCT	VVV	GTC	AGT	GAA	GAC	GAT	AAG	CTT	GTT	GAG	TAC	GAT
	Ala	Leu	Leu	Leu	Λsp	Arg	Phe	Leu	Asp	Ile	Lęu	Gin	Азр	Leu	His	$\mathrm{G}_{\mathbb{Z}} y$	Glu
218	GCT	CTT	CTC	CTT	GAT	CGC	$\mathbf{T}\mathbf{T}\mathbf{T}$	CTC	GAC	ATT	CTC	CAG	GAT	TTA	CAC	GGC	GAG
	Asp	Leu	Arg	Glu	Thr	Val	Gln	Glu	Leu	Tyr	Glu	Ľеu	Ser	Ala	Glu	Tyr	Glu
269	GAT	CTC	CGT	GAA	ACG	GTT	CVV	GΛG	TTA	TAC	GAG	CTT	TCT	GCT	GAC	TAT	GAA
	Gly	Lys	Arg	Glu	\Pr	Ser	Lys	Leu	Glu	Glu	Leu	Gly	Ser	Val	Leu	Thr	Ser
320	GGG	AAG	CGT	GAG	CCT	AGC	ΑΛG	CTT	GAG	GAG	CTA	GGG	AGT	GTC	CTA	ACG	ACT
	Leu	Asp	Pro	Gly	Asp	Ser	Ile	Val	Ilę	Ser	Lys	Лlа	Phe	Ser	His	Mel	Leu
371	1'TG	GAT	CCT	$\mathbf{G}\mathbf{G}\mathbf{T}$	GAC	TCA	TTA	\mathbf{GTT}	ATC	TÇC	AAG	GCT	TTC	TCT	CAC	ATC	CTT
	Аэл	Leu	A⊥a	Asn	Leu	Ala	Glu	Glu	Val	Gln	Ile	Ala	His	Arg	Arg	Arg	Ilc.
422	AAC	$\Box TA$	GCC	AAT	TTG	GCT	GAG	GAG	G‴G	CAG	ATT	GCT	CAC	CGT	CGC	AGG	ATC
	Гла	1'A2	Leu	Lys	Lys	Gly	Asp	Phe	Val	Asp	Glu	Ser	Ser	Ala	ግስ <u></u> ድ	Thr	Glu
473	AAG	AAG	ÇTG	AAG	AAA	GGT	GAT	ITC	GTT	GAT	GAG	AGT	TCT	GCA	ACT	ACT	GAA
	Ser	Asb	Ile	Glu	Glu	Thr	Phe	Lys	Arg	Leu	Va]	Ser	Asp	Leu	Gly	Lys	Ser
524	TCC	GAT	TTA	CAA	GAG	ACT	TTT	AAG	AGG	CTC	GT T	"CG	GAT	CTT	GGT	AAG	TCT
	Pro	Glu	Glu	Ile	Phe	Asp	Ala	Leu	Lys	Asn	Glr	Thr	Val	Asp	Leu	Val	Ъeu
575	CCT	GAA	GAG	ATC	TTT	CAT	GCC	TLC	AAC	AA'l'	CAG	ACT	GTG	GAT	CTG	GTT	TTG
	Thr	Ala	His	Pro	Thr	Gln	Ser	Val.	Arg	Arg	Ser	Leu	Leu	Gln	Lys	His	GLY
626	ACT	GCT	CAT	CCT	ACT	CAG	TCT	GTG	CGT	NGA	TCA	TTG	CTT	CAG	AAG	CAT	GGG
0.01.0	Arg	e 11	Arg	Asp	Cys	beu amm	A⊥a	GIN	Leu	Tyr	Ala	Lys	Asp	ile	Thr	Pro	Asp
677	AGG	ATA	AGG	GAC	TGT	CTT	GCT	CAA	CTC T	TAC	GCA	AAG	GAC	ATT -	ACT	CCT	CAT
700	Asp	Lys	GLD	GIU	Leu	Asp	GIU	ser	Leu	GIN	Arg	GIU	TTG	GIN	ALA	ALA	Phe
128	GAC	AAG	CAG	GAG	CTA	GAT	GAG	TCT	Dre	CAA	AGA mba	GAG	ATT CIN	Dan	GUL	GCA.	TTU
775	ALU	THE	ASP CBM	GIU		ALG	Arg	TUL	DCT.	CCR	1111 3.00	CCD	Can	Asp Chm	CDD	NUC	ALG
119	OGA Also	ACA	GAT	GAG	MII Tura	Dbo	AGA	ACA Clu	The	TLA	HCC Trans	LUG	CAA	viAL Vol	Dro	Tura	Dho
020	Ard	GUN	net	Ser acer	түт	Pile	CBC	GLU	THE DCD	TTE	TTP	ыу з ъъъ	GGTY	vа. CTC	PLO.	лүз рус	D.L.C.
030	LOU	bra	Arg	AG1 (7⇒1	LAL	Thr		Len	LVE	Aco	100	Clv	116	Asn	Sla	Ara	Val
8.91	TTTC.	CCC	CCT	dro.	GAC	aca	CCT	CT3	272 272	220	አብር አብርም	G2G	<u>አ</u> ጥጥ	CAT	CAA	CGT	GTT
001	Pro	1000	Asn	Ala	Pro	Len	TIA	Gln	Phe	Ser	Ser	Tro.	Met	Glv	Glv	λap	Ara
932	CCT	TAC	AAT	GCC	CCA	TTG	Aምም	CAA	TTC	TCT	TCG	TGG	ATG	GGC	GGT	GAT	CGT
202	Asp	Glv	Asn	Pro	Ara	Val	Thr	Pro	Glu	Val	Thr	Ara	Asp	Val	Cvs	Leu	Leu
983	GAT	GGT	AAT	CCG	AGG	GTC	ACA	CCT	GAG	GTC	ACT	AGA	GAT	GIG	TGC	TTG	TTG
200	Ala	Ara	Met	Met	Ala	Ala	Asn	Leu	Tvr	Tvr	Asn	Gln	Ile	Glu	Asn	Leu	Met
1034	GCT	λGA	ATG	ATG	GCT	GCC	AAT	CTC	TAC	TAT	AAC	CAA	ATC	GAG	AAT	CTG	ATG
1001	Pine	Glu	Leu	Ser	Met	מיניוי	Ara	Cvs	Thr	Asp	Glu	Phe	Arg	Val	Ara	Ala	Asp
1085	ΤΞΤ	GΛG	TTA	TCT	ATG	TGG	CGT	TGC	ACT	GAT	GAA	TTC	CGT	GTG	CGG	GCG	GAT
	Glu	Leu	Uis	Arg	Asn	Ser	Arg	Lys	Asp	A⊥a	Ala	Lys	His	Tyr	Ile	$\operatorname{GL}\mathbf{u}$	⊋he
1136	GAA	CTG	CAC	AGG	AAC	TCA	AGG	AAA	GAT	GCT	GCA	AAA	CAT	TAC	ATA	GAA	TTC
-	Trp	Lys	Thr	Ile	Pro	Pro	Thr	Glu	Pro	Tyr	Arg	Val	Ile	Leu	Gly	Asp	Val
1187	TGG	AAG	ACA	ATT	CCT	CCA	ACT	GAG	CCA	TAC	CGT	GTG	ATT:	CTT	GGT	GAT	GTG
	Arg	Asp	Lys	Leu	Tyr	His	Thr	Arg	Glu	Arg	Ser	Arg	Gln	Leu	Leu	Ser	Asn
1238	AGG	GAT	AAG	CTG	TAT	CAC	ACA	CGT	GAG	CGT	TCC	CGC	CAA	$\mathbf{T}\mathbf{T}\mathbf{G}$	CTG	AGT	AAT
									115								

115

Gly Ile Ser Asp Ile Pro Glu Glu Ala Thr Phe Thr Asn Val Glu Gln Pho 1289 GGA ATC TCG GAT ATT CCT GAA GAA GCT ACC TTC ACT AAT GTG GAA CAG TTC Leu Glu Pro Leu Glu Leu Cys Tyr Arg Ser Leu Cys Ser Cys Gly Asp Scr 1340 TTG GAG CCT CIT GAG CTC TGT TAC CGA TCA CTA TGT ICA TGT GGI GAC AGC Pro Lie Ala Asp Giy Ser Leu Leu Asp Phe Leu Arg Gln Val Ser Thr Phe 1391 CCG ATA GCT GAT GGA AGC CTT CTT GAT TTC TTG AGG CAA GTC TCT ACC TTT Gly Leu Ser Leu Val Arg Leu Asp Ile Arg Gln Glu Ser Glu Arg His Thr 1442 GGA CTC ICC CTT GTG AGA CTT GAC ATC AGG CAA GAG TCT GAA CGC CAC ACA Asp Val Leu Asp Ala Ile Thr Lys His Leu Asp Ile Gly Ser Ser Tyr Arg 1493 GAT GTC TTG GAT GCT ATC ACC AAC CAC TTG GAC ATC GGT TCC TCC TAT AGA Asp Trp Ser Glu Glu Gly Arg Gln Clu Trp Leu Leu Ala Glu Leu Ser Gly 1544 GAC TGG TCT GAA GAA GGC CGA CAG GAA TGG CTT CTT GCT GAA CTA AGC GGC Lys Arg Pro Leu Phe Cly Pro Asp Leu Pro Lys Thr Glu Clu Ile Ser Asp 1595 AAA CGT CCA CTT TTC CGA CCT GAT CTT CCC AAA ACC GAA GAA AFT TCT GAT Val Leu Asp Thr Phe Lys Val Ile Ser Clu Leu Pro Ser Asp Cys Phe Gly 1646 GTC CTG GAC ACA TTC AAA GTC ATA TCT GAG CTG CCT TCA GAT TGT TTT GGA Ala Tyr Ile Ile Ser Met Ala Thr Ser Pro Ser Asp Val Leu Ala Val Clu 1697 GCT TAT ATT ATC TCT ATG GCA ACT TCA CCT AGT GAT GTG CTT CCG CTT GAG Leu Leu Gln Arg Glu Cys His Val Lys Asn Pro Leu Arg Val Val Pro Leu 1748 CET TTA CAG CGC GAA TGC CAT GTG AAA AAT CCA CTT AGA GIT GTT CCA CTC Phe Glu Lys Leu Ala Asp Leu Glu Ala Ala Pro Ala Ala Val Ala Arg Leu 1799 TIT GAG AAG CTA GCT GAT CIT GAA GCA GCT CCT SCC GCT GTT GCA AGA CTC The Ser Ile Asp Trp Tyr Lys Asn Arg Ile Asn Gly Lys Gln Glu Val Met 1850 TIT TCT ATA GAC TGG TAC AAA AAC CGT ATT AAC GGT AAA CAA GAG GTT ATG Ile Gly Tyr Ser Asp Ser Gly Lys Asp Ala Gly Arg Leu Ser Ala Ala Trp 1901 ATT GGT TAC TCA GAT TCA GGC AAA GAT GCA GGG CGT CIC TCA GCT GGC TGG Glu Leu Tyr Tys Ala Cln Clu Glu Leu Val Lys Val Ala Lys Lys Tyr Gly 1952 GAG CTA TAC ANA GCT CAA GAA GAG CTT GTG AAG GTT GCT AAG AAA TAT GGA Val Lys Leu Thr Met Phe His Gly Arg Gly Cly Thr Val Gly Arg Gly Gly 2003 GTS AAG CTA ACT ATG TIC CAI GGC CGT CGI GCC ACA GTC GGA AGA GGA GGT Gly Pro Thr His Leu Ala Ile Leu Ser Gln Pro Pro Asp Thr Val Asn Gly 2054 GGT CCT ACT CAT CTT GCT ATA TTG TCT CAG CCA CCA GAT ACA GTT AAT GGC Ser Leu Arg Val Thr Val Gin Gly Glu Val Ile Glu Gin Ser Phe Gly Glu 2105 TCT CTT CGA GTC ACG GTT CAG GGT GAA GTC ATT GAG CAA TCA TTT GGG GAG Ala His Leu Cys Phe Arg Thr Leu Gln Arg Phe Thr Ala Ala Thr Leu Glu 2156 GCA CAC TYA TGC TTY AGA ACA CTT CAA CGT TTC ACA GCA GCT ACT CTA GAG His Gly Met Asn Pro Fro Ile Ser Pro Lys Pro Glu Trp Arc Ala Leu Leu 2207 CAC GGA ATG AAC CCT CCG ATT TCA CCA AAA CCC GAG TGG CGT GCT TTG CTT Asp Clu Mot Ala Val Val Ala Thr Glu Glu Tyr Arg Ser Val Val Phe Gin 2258 GAT CAA ATG GCG GTT GFT GCA ACT GAG GAA TAC CGA TCT GTC GTT TTC CAA Glu Pro Arg Phe Val Glu Tyr Phe Arg Leu Ala Thr Pro Glu Leu Glu Tyr 2309 GAA CCT CGA FTC GIC GAG TAT TTC CGC CTC GCT ACT CCG GAG CIG GAG TAT Gly Arg Met Asn Ile Gly Ser Arg Pro Ser Lys Arg Lys Pro Ser Gly Gly 2360 GGA CGT ATG AAT ATT GGA AGT AGA CCT TCA AAG CGA AAA CCA AGC GGT GGG The Glu Ser Leu Arg Ala Ile Pro Trp Ile Phe Ala Trp Thr Glr. Thr Arg 2411 ATC GAA TCT CTC CGT GCA ATC CCA TGG ATC TTT GCT TGG ACG CAA ACA AGA Phe His Leu Pro Val Trp Leu Gly Phe Gly Ala Ala Phe Arg Tyr Ala Ile 2462 TTC CAT CTT CCT GTA TGG TTA GGT TTC GGA GCA GCA TTT AGG TAT GCG ATC Lys Lys Asp Val Arg Asn Leu Nis Met Leu Gin Asp Met Tyr Lys Gln Trp. 2513 AAG AAG GAT ETG AGA AAC CTT CAC ATG CTG CAA GAT ATG TAT AAA CAA TGE Pro Phe Phe Arg Val Thr Ile Asp Leu Ile Glu Met Val Phe Ala Lys Gly 2564 CCC TTT TTC CGA GTC ACC ATC GAT CTA ATT GAA ATC GTG TTC GCC AAG GGA Asp Pro Gly Ile Ala Ala Leu Tyr Asp Lys Leu Leu Val Ser Glu Asp Leu 2615 GAC CCC GGG ATC GCT GCT TIG TAC GAC AAA CTT CTT GTC TCA GAA GAT TTA Trp Ala Phe Gly Glu Lys Leu Arg Ala Asn Phe Asp Glu Thr Lys Asn Leu 2666 TGE GCT TTT GGA GAG AAA CTC AGA GCC AAC TTT GAT GAA ACC AAG AAC CTC Val Leu Gin Thr Ala Gly His Lys Asp Leu Leu Glu Gly Asp Pro Tyr Leu 2717 GTC CTC CAG ACT GCT GGA CAT AAA GAC CTT CTT GAA GCA GAT CCT TAC TTG

Sequence alignments of higher plant PEPcs Figure 4.5

The numbers correspond to amino acid residue number. The shaded box is the PEPc phosphorylation motif conserved among all PEPc sequences known to date. Small black circles (•) in the sequence alignment denote conserved residues. Small black vertical lines ()) denote similar residues.

		10	20	3	30	40	50
		•• [• • • •	•••••	• • •]	•• ••
А.	thalíana	MAGRNISK	м4 33 рА рыя	QLVPAKVS	SEDDKLVEY	CALLER	FLÐILQDLH
F.	trinervia (C4)	MANENVEK	LASSINGLE	LLVPGKV8	SEODKLVEN	DALLLOK	FLDILQDLH
G.	hirsutum	MAGRKVEK	MA SEL	LLAPGKV:	BEODKEVEN	(DAVLDR	FLDILÇDLH
G.	max	MANRNLEK	MANNARQLI	LUVPAKVS	SEDDKLVEY	DALLLDR	FLDILQDLH
М.	crystallinum	MSTVKLDR	LT	LLAPKKVS	SEDDKLIEN	DALLOR	FIDILQNUH
Ν.	tabacum	MATRSLEK	LA DLF	ALVPGXV	SEDDKLVEY	DALLUDR	FLDULQDLH
ς.	tuberosum	MTTRNLDX	LA DIF	QLVPARVS	SEDDKLVEY	DALLEDR	EFDIFÖDPH
s.	vulgare	MPER	IIQ DLF	J.LAPG: (V:	SEDDKILVEN	DALLVOR	FEDILQDLH
ν.	<i>planifolia</i> (air roots)	М	AS	LLAPAKV	SEDDKUVEY	VRLLLØR	ялотторга
v.	planifolia (CRM lesf)	M	LA	LLAPAKV	SEDDKLLEY	(VSLLLDR	FLOTIQUIH
Ζ.	mays (C3)	MAALGPKMER	ls sig lf	MLVPGKV	GEDDKLEEN	(DALLLOR	FEDITODIA
Ζ.	mays (C4)	MASTKAPGPGEK	HHE CLF	QLVPGKV:	GEDDKLIEN	DALLVOR	FUNTLQDLH
Ζ.	mays (root)	MPER	HOSENAOLE	LLAPGKV	SEDDKLVEN	DALLVDR	FLDILQDLH
	-						
		60	70	80	96	3	100
		• • ••; •	•	1 1	· • •	i - 1 (•	111 11 ••
А.	thaliana	GEDLRETVOELY	ELSAEYEGI	REPSKLE	-ELGSVLTS	SLDFGDSI	VISKAFSEM
F.	trinervia (C4)	GEDLKEAVQQCY	ELSARYESI	HDPKKLE	-ELGSILTS	SLDTGDSI	VIAKAFSEM
G.	hirsutum	GEDIRETVOECY	FLSAEYEGH	HDPKILE	-ELGKVLTS	SLDPGDSI	VVTKSFSBM
G.	max	GEDLKETVQEVY	ELSAEYEGE	(HDPKKLE)	-ELGNLITS	BLDAGDSI	VVAKSFSHM
м.	crystallinum	GEDIKETVQELY	EQSAETERI	HDPKKLE	-ELGSMVTS	SLDAGDSI	VVAKSESHM
Ν.	tabacum	GEDLKETVQECY	ELSAETEGI	НОРККЪЕ:	-ELGNVLTS	SLDPGDSI	VIAKAFSHM
5.	tuberosum	GEDLKETVQECY	ELSAEYEAR	KHDPKKLE-	-ELGNVLT:	STOPGDSI	VIAKAFSHM
s.	vulgare	GPHLREFVQECY	ELSAEYENI	DRDEARLG	-ELGSKLTS	SLPFGDSI	VVASSESHM
v.	planifolia (air roots)	GEVVRETVQELY	ELSAEYESH	KHDPKKLD	-FIGNLLIS	SLOFGNSI	VIASSESHM
v.	planifolia (CAM leaf)	GEVLRETVQELY	ELSSEYESE	KHDPKKFG	MNLGILLI	PLDPGNSI	VIASSESHM
Ζ.	mays (C3)	GDELKEMVQECY	EVAAEYETI	RHDLQKLD	-FLGKMIT:	GLDFGDS1	VIAKSLSHM
z.	mays (C4)	SPSLREFVQECY	EVSADYEGI	KGDTTKLG	-ELGAKLTO	GLAPADAI	LVASSILHM
7.	mays (root)	CPHLREFVÇECY	ELSAEYENI	ORDEARLG	-FLGSKLT:	SLFPGDSI	VVASSIFSRM
	-						
		110 1	.20	130	140	150)
		••• ••••• •	1 [+	• • •		• • • • !	• •
А.	thaliana	LNLANLAEEVQJ	AHRRRIKKI	LKKGDFVD	ESSATTESI	DIEETFKR	LVSDLGKSP
<i>P</i> .,	trinervia (C4)	INLANLASELQI	AYRREKL	KS-GDFAD	EANATTES	DIEETFKF	LVHKLNKSP
G.	hirsutum	LNLGNLABEVQI	AYRRRIKL	K-KGDFAD	ESSATTES	DIFETERE	LVGQLNKSP
G.	max	LNLANLABEVO	AHSRRNKL	K-KGDFAD	ENNATTES	DIEETLKE	LVVDMKKSP
М.	crystallinnm	LNLANLAEEVOI	SRRKRVKK	VKKGDFMD	ENTAMTES	CMEETLRF	RLIVDLKKSP
N.	tabacum	LNLANLAEEVOL	AYRRROKL	KR-GDFAD	ENNATTES	DIEETFKF	LVGDLKKSP
s.	tuberosum	LNLANLAEEVOI	AYRRROKL	KKKGDEGD	ESNATTES	DIEETEKE	LVGDLKKSP
s.	vulgare	LNLANLAEEVO	AORRRIKL	KR-GDFAD	EASAPTES	DIEETIKI	
ν. ν	planifolia (air roota)	LNLANLAEEVOI	AHRRRIKL	K-KGDFVD	ENSATTES	DIEETIKF	APAHDI KKSD
v	planifolia (CAM leaf)	LNLANLAEEVOI	AFRRRIKL	K-KGDFAD	EASATTES	DFEETLKF	RIVGKLKKTP
<i>z</i> .	mavs (C3)	INLANLALEVO	AYRRRIKL	K-XGDFAD	ENSAITES	DIEETLKF	APAADPRKSB
z.	mays (C4)	LNLANLAEEVOI	AIIRRRNSK	LKKGGFAD	IGSATTES:	DIEETIKF	RLVSEVGKSP

- Z. mays (C4)
- Z. mays (root)

LNLANLAEEVQIAHRRRIKLKR GDFADEASAFTESDIEE'LKRLVSQLGKSR

160 170 180 190 200 210 EEIFDALKNQTVDLVLTAHPTQSVRRSLLQKHGRIRDCLAQLYAKDITPDDKQ A. thaliana EEVFDALKNQTVELVLTAHPTQSVRRSLLQKHGRIRNCLAQLYAKDITPDDKQ F. trinervia (C4) EEVFDALKNQTVDLVLTAHPTQSVRRSLLQKHGRIRNCLTQLYAKDTTPDDKQ G. hirsutum QEVFDALKNQTVDLVLTAHPTQSVRRSLLQKHGRIRKNLTQLYAKDUTPDDKQ G. max QEIFETLKNQTVDLVFTAHPTQSVRRSLLQKHGRIRDCLAQLYAKDITPDDKQ M. crystallinum OEVFDALKNOTVDLVLTAHFTOSV3RSLLOXHGRIRDCLAOLYAKDITPDDKO N. tabacum QEVFDAIKNQTVDLVLTAHFTQSVRRSLLQXHGRIRDCLACLYAKDITPDDX0 S. tuberosum REVFDALKNQTVDLVFTAHFTQSVRRSLLQKHGRIRNCLRQLYAKDITADDXQ S. vulgare V. planifolia (air roots) EEVFDALKNQTVDLVLTAHPTQSVRRSLLQKHARIRNCLAQLYAKDITPDDKQ V. planifolia (CAM leaf) QEIFDALKSQTVDLVLTANFTQSVRRSLLQKHGRTRNCLTQLYAKOITPDDKQ 2. mays (C3) AEVFDALKSQTVDLVLTAHFTQSVRRSLLQKHSRIRNCLVQLYSKDITPDDKQ Z, mays (C4) EEVFEALKNOTVDLVFTAHFTQSARRSLLCKNARIRNCLTQLNAKDITDDDKQ Z. mays (root) EEVFDALKNCTVDLVFTAHFTQSVRRSLLQKHGX1RNCLRQLYAKDITADDKQ

A. Lhaliana

C. hirsutum

N. tabacum

S. tuberosum

Z. mays (C3)

Z. mays (C4)

A. tnaliana

G. hirsutum

N. tabacum

S. vulgare

S. tuberosum

Z. mays (C3) Z. mays (C4)

Z. mays (root)

G. max

F. trinervia (C4)

M. crystallinum

Z. mays (root)

S. vulgare

G. max

F. trinervia (C4)

M. crystallinum

250 220 230 240 260 ····· ELDESLOREIQAAFRTDEIRRTPPTPQDEMRAGMSYFHETIWKGVPKFLRRVD ELDEALHRELQAAFRTDEIRRTPPTPQDEMRAGMSYFHETIWKGVPXFLRRVD ELDEALQREIOAAFRTDEIRRNPPTPODEMRAGMSYFHETIWKGVPKFLRRVD ELDEALQREIQAAFRTDEIRRTPPTFQDEMRAGMSYFHETTWKGVPTFLRRVD ELDEALQREIQAAFRTDEIRRTQPTPQDEMRACMSYFHETIWNCVPKFLRRLD ELDEALQREIQAAFRTDEIRRTAPTPODEMRAGMSYFHETIWKGVPKFLRRVD ELDEALQREIGAAFRTDEIRRTPPTPODEMRAGMSYFHETINKGVPKFLRRVD ELDEALQREIQAAFRTDEIRRTFPTPQDEMRAGMSYFHETIWKGVPKFLRRID V. planifolia (air roots) ELDEAL-KGGSAAFRTDEIRRTPPTPQDEMRAGMSYFHETIWKGVPKFL3RVD V. planifolia (CAM leaf) ELDEALHREIQAAFRTDEIRRIPPTPQDEMRAGMSYFHETIWKGVPQFLRRVD ELDEALQREIQAAFRTDEIRRTQPTPQDEMRAGMSYFHETIWKGVPKFLRRVD ELDEALQREIQAAFRTDEIRRAQPTPQAEMRYGMSYIHETVWKGVPKFLRRVD ELDEALQREIQAAFRTDEIRRTPPTPQDEMRAGKSYFHETIWKGVPKFLRRID

270 280 290 300 310 ******** TALKNIGIDERVPYNAPLIQFSSWMGGORDGNPRVTPEVTRDVCLLARMMAAN TALKNIGINERFPYNAFLIQFSSWMGGORDGNPRVTPEVTRDVCLLARMMTSN TALKNIGINERVPYNAPLIOFSSWMGGDRDGNPRVTPEVIRDVCLLARMMAAN TALKNIGINERVPYNAPLIQESSWMGGDRDGNPRVTPEVTRDVCLLARMMAAN TALKNIGITERVPYNAPLIQFSSWMGCDRDGNPRVTPEVTRDVCLLARMMAAN TALKNIGINERLPYNAPLJQFSSWMGGDRDGNPRVTLEVTRDVCLLARMMAAN TALKNIGINERVPYNAPLIQFS\$WMGGDRDGNPRVTPEVTRDVCLLARMMAAN TALKNIGINERLPYNAPLIQFSSWMGGDRDGNPRVTPEVTRDVCLLARMMAAN v. planifolia (air roots) TALKNIGINERVPYNAFLIQESSWMGGDRDGNPRVTPEVTRDVCLLARMMAAN V. planifolia (CAM leaf) TALKNLGINERLPYNAPLIQFSFWMGGORDGNPRVTPEVTRDVCLLARMMAAN TALKNIGINERVPYNAPLIOFSSWMGGDRDGNPRVTPEVTRDVCLLARMMASN TALKNIGINERLPYNVSLIRFSSWMGGDRDGNPRVTPEVTRDVCLLARMMAAN TALKNIGINERLPYNAPLIQFSSWMGGDRDGNPRVTPEVTRDVCLLARMMAAN

	:	320	330	340	350	360	370
		• · ·	•• •• • ••	• • • ••	•	• •••••	•
л.	thaliana	LYYN	21enlmfelsm	WRCTDEFRVRAD.	elhrnsrkla-	-AKHYIEFWKI	TIPPTE
F.	trinervia (C4)	MYFSQ	<u>Diedlmiems</u> m	WRCNSELRVRAE	el,yrtarkd	-AKEAIELMKI	Q//PENQ
G.	hirsutum	LYFS	QIEDLMFELSM	WRCSDELRIRAD	ELHRSSKKDA	KHYTEFWKQ	N 65NE
G.	max	LYYS	QIEDLMFELSM	WRCNDELRVRAD	ELNRSSKKNS	VARHYLEFWK	VIPPNE
М.	crystallinum	MYFS	QIDELMFELSM	WRCTDELRERAE	ELHKYSKRDS	KHATEEMKO	DT PSSE
Ν.	tabacum	LYYS	QIEELMFELSM	WRCNDDLRIRAA	ELYRSSRRD-	-TKHYIEFWKI	FTPPSE
s.	tuberosum	LYYS	QIEDLMFELSM	WRCNEELRVRAD	DLQRSSRRD-	-EKHYIBEWK(OVPPNE
s.	vulgare	LYFS	QIEDLMFELSM	WRCSDELRIRAD	ELHRSSKRA	-VKHAISEMKČ	VPPNE
v.	planifolia (air roots)	LYFS	QIEDEMFELSM	WRC5DELRIRAD	ELHRSSKRDA		VPPNE
v.	planifolia (CAM leaf)	LYFS	QIE DLM FELSM	WPCSDELRLRAD	ELNVSSKKDA	KHYTEFWKE	KVPPNE
z.	mays (C3)	LYCS	QIEDLMFELSM	WRCSDELRMRAD	VLHLSTKKDA	KHAIESMRE	WEBNE
z.	mays (C4)	PAID0	QIEELMFELSM	WRCNDELRVRAE	ELHSSSG-SK	VIKYYIEFWK(21 PPNE
Ζ.	mays (root)	LYFS	DIEDLMFELSM	WRCSDELRIRAD	ELHRSSRKA-	-AKEYIEFWK(<u>2006bne</u>

	380	390	400	410	420
	••• • •• ••	•••• • •	•[[[]][[]]	• • • • • • [•
	PYRVILGDVRDKLY	HTRERSROLLS	NGISDIPEEA	r f tnveqflei	PLELCYRS
	PYRVILCOVRDKLY	NTRERSRHLLV	DGKSDIPDEA	VYTNVEQLLEI	PLELCYRS
	PYRIILGDVRDKLY	NTRERARSLLA	NGESDIPEEA	AFTNVEQFLEI	PLELCYRS
	PYRVLLGEVRNRLY	QTRERSRHLLA	HGYSDIPEEE'	ff'invelflei	PLELCYRS
	PYRVILADVRDKLY	YTRERSROLLA	SEVSEIPVEA	PFTEIDQFLEI	PLELCYRS
	PYRVILGDVRDKLY	QTRERTROMLA	HGESDIPEDA	FYNNVEQFLEI	PLELCYRS
	PYRVILGDVRDKLY	QTRERARQULC	HGYSEIPEEA	FYTNIEQFLE	PLELCYRS
	PYRVILGDVRDKLY	YTRERSRHLLS	SGISEIPEEA	TETNVEQELE	PLELCYRS
ots)	PYRVVLADVRDKLY	NTRERSRHULS	SGISDVPEES;	SMINVEQULE	PLELCYRS
aī)	PYRVILGDVRDKLY	HTRERARQILS	HGVSCIPE-A	FFTNVEQFLE	PLELCYRS
	PYRVILSDVRDKLY	NTRERSRELLS	SGHSDIPEEA	FLTNVEQLLE	PLELCYRS
	PYRVILGHVRDKLY	NTRERARHLLA	SGVSEISAES.	SFUSIEEFLE	PLELCIKS
	PYRVILGDVRDKLY	YTRERSRHLL	SGISEILEEA	FENVEQFLE	PLELCYRS

the second second

3

. . .

v

÷

е: -

1

ۇ. ئۇرىن

	430	440	450	460	470	
	•• • E	• • • • • • • • • •	*** !*		•.•[•[•	'
	LCSCGDSFI	ADGSLLDFLRQV	STEGLSLV	RLDIRQESERL	FIDVLDAITK	ELDI
	LCDCGDHVI	ADGSLLDFLRQV	STEGLSLV	KLDIRQESDRI	ITEVLDAITQ	FLGI
	LCACGDRPI	ADGSLLDFLRQV	STEGLSLV	RLDIRQESDRI	ITDVLDAITK	HLDI
	LCACGDRAI	ADGSLLDFLRØV	STFGLSLV	RLDIRQESDRI	ITDVLDAITK	HUEI
	LCACGDRPV	ADGSLLDFMRQV	ATTGLCLV	KLDIRQESERH	ITDVMDAITT	HLGI
	LCECGDRPI	ADGSLLDFLROV	STFGLSFV	RLDIRQESDRH	HTDVLDAITÇ	HLEI
	LCACGDLSI	ADGSLLDFLRQV	STEGLSLV	REDIRQESDRE	HTDVLDAITÇ	HLEI
	LCACGDKPI	ADGSLLDFLRQV	FNFGLALV	KLDIRQESDRH	ITDVLOSITI	HLGI
ots)	LCSCGDRPI	ADGSLLDFMRQV	STFGLALV	KLDIRQESORF	FDVIDALTT	HLGI
af)	LCACAHRPI	ADGSLLDFLRQV	STEGLSLV	KLDIRQESERF	ITDVMDAITR	HIEI
	LCACGDSVI	AGGTLLDFLRQV	STEGLSLV	REDIROESDRE	TTLAGIVETI	YLGI
	LCDCGDKAI	ADGSLLDLLRQV	FTFGLSLV	KLDIRQESERF	TTLAGIVCTE	HLCI
	LCACGEKPI:	ADGSLLEFLRQV	STEGLALV	KLDIRQESDRI	ITDVLDSITT	HLCI

G. max M. crystallinum N. tabacum S. tuberosum S. vulgare V. planifolia (air roc V. planifolia (CAM lea Z. mays (C3) Z. mays (C4) Z. mays (root)

A, thaliana

G. hirsutum

F. trinervis (C4)

A. thaliana
S. trinervia (C4)
G. hirsutum
G. max
M. crystallinum
N. tabacum
S. tubercoum
S. vulgare
V. planifolia (air roV. planifolia (CAM leader)
Z. mays (C3)
Z. mays (C4)
Z. mays (root)

480 490 500 510 520 530 •• •]•[•] ••]••• • •]]••• • • []]*•• GSSYRDWSEEGRQEWLLAEL-SGKRPL-FGPDLPKTEEISDVLDTFKVISELP A. thaliana GS-YREWSEEKRQEWLLAEL-SGKRPL-IGPD PKTEEVKDCLDTFKVLAELP F. trinervia (C4) GS YREWPEERRQEWLLSEL-RGKRPL-FGPDLPKTEEVADVLDTFHVISELP G. hirsutum GS-YQEWSEEKRQQWLLSEL-SGKRPL-FGPDLPQTEEIRDVLETFHVIAELP G. max GS-YRDWTEEKRQDWLLSEL-RGKRPL-FGPDLPRTDEIADVLDTINVIAELP M. crystallinum CS-YREWSEERROEWLLSEL SGKRPL-FGPDLPRTEEIADVLDTLHVIAELP N. tabacum CS-YRDWSEERRQEWLLSEL -SGKRPL-FGPDLPKTEEIADVLDTFHVIAELP S. tubercsum CS-YAEWSEEKRODWLLSEL RGKRPL-FGSDLPOTEETADVLGTFHVLAELP S. vulgare V. planifolia (air roots) GS-YREWSED-RGEWLLSELASGKGDLQIGSDLPATCSV-DVIDTFLVIAELP V. planifolia (CAM leaf) GS-YREWSEEQRQQWLLSELAE-KRFL-FGPDLPRTEEIADVLDTFQVIAELP Z. mays (C3) GS-YREWTEERROEWLLSEL-NGKRPL~FGSDIPKTEETSDVLDTF9VLAELP 2. mays (C4) GS-YREWPEDKROEWLLSEL-RGKRPL-LPPDLPQTDEIADVIGAPHVLAELP Z. mays (root) GS-YAEWSEEKRQDWLLSEL-RGKRPL-FGSDLPQTEETADVLCTFHVLAELP

570 540 550 560 580 SDCFGAYITSMATSPSDVLAVELLQRECHVKNPLRVVPLFEKLADLEAAPAAV SDCFGAYIISMATSTSDVLAVELLQREYHIKHPLRVVPLFEKLADLEAAPAAM SDSFGAYIISMATAPSDVLAVELLQRECHVKQPLRVVPLFEKLADLEAAPAAV LONFGAYIISMATAPSDVLAVELLQRECHVKHPLRVVPLFEKLADLEAAPAAL SDSFGAYVISMATAPSDVLAVELLQRECKVKKPLRVVPLFEKLADLEAAPASM SOCFCAYIISMATAPSDVLAVELLQRECHVKQPLRVVPLFEKLDDLESASAAV ADCFGAYIISMATAPSDVLAVELLQRECRVRQPLRVVPLFEKLADLDAAPAAV ADCFGAY1ISMATAPSDVLAVELLQRECHVKQPLRVVPLFEKLADLSAAPAAV V. planifolia (air roots) PDNEGAYIISMATAPSDVLAVELLQRECHVRQPLRVVFLFEKLADLEAAPAAL SDGFGAYITSMATSSSDVLAVELLQREFRVPKPLRVVPLFEKLADLEAAPAAV SDNFGAY1:SMATAPSDVLAVELLQRECHVKTPLRVVPLFEKLADLEAAPAAL PDSFGPYIISMATAPSDVLAVELLORECGVROPLPVVPLFERLADLQSAPASV ADCFGAYIJSMATAPSDVLAVELLQRECHVKHPLRVVPLFEKLADLEAAPAAV

590	600	610	620	630
••••()••	• • • • • • •	1	•• •	•• • +•
ARLESIDWY	KNRINGKQEVW	IGYSDSGKDAG	RLSAAWELY	(AQEELVKVAKKY
TRLFSMDWY	RNRIDGKOEVM	IGYSDSGKDAG	GRESAAWQLYB	(TQEQIVKIAKEF
ARLESICWY	RDRINGKQEVN	IGYSDSGKÐAG	GRESAAWQEYF	(AQEELVKVAKQY
ARLESVOWY	RNRINGKQEVN	IGYSDSGKDAG	GRESAANQLYE	AOEELIMVAKQY
TRLFSVDWY	KNRIDGKQEVM	IIGYSDSGKDAG	GRLSAAWOMYE	VQEELLKVSKKF
ARLFSIEWY	RNRINGKQEVM	IVGYSDSGXDAG	GRFSAAWQLYB	(VŐEETIKAYKEH
ARLFSIEWY	RNRINGKQEVE	LEGYSDSGKDAG	GRESAAWQLY	(AQEELIQVAKEF
ARLFSIDWY	MNRINGKQEVN	IIGYSDSGKDAG	GRLSAAWQMY	(AQEELIKVAKHY
ARDFSIEWY	RNRIDGKQEVA	ngysdsgkdad	GRLSAAWQLYI	GQEDLIKVAKKY
SELESIDWY	RDRINGKQEVM	ILGYSDSGKDA(GRTFCCLAVY	TQEELVQVAKQF
ARLESIDWY	RORINGKOEVN	IGYSDSGKDA	GRLSAAWQLYI	AQEELIKVAKDF
ERLFSVOWY	MDRIKGKQQV	VGYSDSGKDA	GRLSAAWQLYI	RAQEEMAQVAKRY
ARLESIOWY	MDRINGKÇEVM	IIGYSDSGKDA	GRLSAAWQMYI	KAQEELIKVAKHY

- A. thaliana F. trinervia (C4) G. hirsulam G. RAX M. czystallinum N. Labacum S. tuberosum S. vulgare V. planifolia (CAM leaf) Z, mays (C3) Z. mays (C4)
- Z. mays (root)

A, thaliana

- F, trinervia (C4) G. hirsutum G. max M. crystallinum N. tabacum S. tuberosum S. vulgare V. planifolia (air roots)
- V. planifolia (CAM leaf) Z. mays (C3)
- Z. mays (C4) Z. mays (root)

121
640 650 660 670 680 GVKLTMFHGRGGTVGRGGGFTHLAILSQFPDTVNG5LRVTVQGEVIEQSFGEA A. thaliana F. trinervis (C4) GVKLVIFHGRGGTVGRGGGPTHLALLSQPPDTINGSLRVTVQGEVIEQSFGEE GVKLTMFHGRGGTVGRGGGPTHLAILSQPPDTIHGSLRVTVQGEVIEQSFGEE G. hirsutum **GVKLTMFHGRGGTVGRGGGPTHLAILSQPPETIHGSLRVTVQGEVIEQSFGEQ** G. mex M. crystallinum GVKLTMFHGRGGTVGRGGGPTHLAILAQPAETIGGSLRVTIQGEVIEQSFGEQ GVKLTMFHGRGGTVGRGGGPTHLAILSQPPDTIQGSLRVTVQGEVIEQSFGEE N. tabacum DVKLTMFHGRGGTVGRGGGPAHLAILSQPPETIHGSLRVTVQGEVIEQSFGEE S. tuberosum S. vulgare GVKLTMFHGRGGTVGRGGGPTHLAILSQPPDTIHGSLRVTVQGEVIEHSFGEE V. planifolia (air roots) GVKLTMFHGRGGTVGRGGGPTHLAILSQPPDTIRGSLRVTVQGEVIEQSFGEE V. pianifolis (CAM leaf) GVKLTMFHGRGGTVGRGGGPTHLATLSQPPDTINGSLRVTVQGEVIEQCFGEE Z. mays (C3) GVKLTMFHGRGGTVGRGGGPTHLAILSQPPDTIHGSLRVTVQGEVIEQSFGEE Z. mays (C4) GVKLTLFHGRGGTVGRGGGPTHLAILSQPPDTINGSIRVTVQGEVIEFCFGEE GVKLTMFHGRGGTVGRGGGPTHLAILSQPPDTTHGSLRVTVQGEVIEHSFGEE Z. mays (root)

A. thaliana F, trinervia (C4) G. hirsutum G. max M. crystallinum N, tabacum S. taberosum S. vulgare V. planifolis (CAM leaf) 2. mays (C3)

2. mays (C4)

A. thaliana

Z. mays (root)

690 700 710 720 730 740 ***|****| ********|* | ||***** *|*||**:** ** |* |*|* HLCFRTLQRFTAATLEHGMNPFISPKPEWRALLDEMAVVATEEYRSVVFQEPR HLCFRTLQRFCAATLEHGMNPPISPRPEWRELMDQMAVVATEEYRSVVFKEFR HLCFRTLQRFTAATLEHGMHPPVSPNPEWRALMDEMAVVATKEYRSVVFQEPR HLCFRTLQRFTAATLEHGMNPPISPKPEWRALMDEMAVIATEEYRSIVFKEPR HLCFRTLQRYTAATLEHGMNPPKSPKPEWRALLDQMAVVATEEYRSIVFKEPR HLCFRTLQRFTAATLEHCMAPPVSPKPEWRALMDETAVIATEKYRSIVFKEPR HLCFRTLQRFTAATLEHGMHPPVSPKPEWRALMDEIAVVATEKYRSIVFKEFR LLCFRTLORYTAATLEHGMHPPISPKPEWRALMDEMAVVATKEYRSIVFQEPR V. planifolia (air roots) HLCFRTLQRFTAATLEHGMHPPISPKPEWRTLLDEMAVVATEEYRSIVFQEPR RLCFRTLQRYTAATLEHGMNP-YFSKPEWRSLIJDEMAVIATNEYRWIVFQEPR HLCFRTLQRFTAATLEHGMHPPNAPKPEWRALLDEMAVVATEEYRSIVFKEPR **HLCFOTLORFTAATLEHGMHPPVSPKPEWRKLMDEMAVVATEEYRSVVVKEAR** LLCFRTLQRYTAATLENGMNPPISPKPEWRALMDEMAVVATKEYRSIVFQEPR

- 770 780 750 760 790 FVEYFRLATPELEYGRMNJGSRPSKRKPSGGIESLRAIPWIFAWTQTRFHLPV **FVEYFRLATPELEFGRMNIGSRFSKRKPSGGIESLRAIPWIFSWTQTRFHLPV** FVEYFRLATPELEYGRMNIGSRESKRKESGGIESLRAIPWIFAWTQTRFHLEV FVEYFRIATPELEYGRMNIGSRFAKRPSGGIETLRAIPWIFAWTOTRFHLPV FVEYFRLATPETEYGRMNIGSRPSKRKPSGGTESLRAIPWIFAWTQTRFHLPV FVEYSALATPELEYGRMNIGSRPSKRKPSGG1ESLRA1PW1FAWTQTRFHLPV FVEYFRLATPELEYGRMNIGSRPSKRKPSGGIESLRATPWIFAWTGTRFHLPV FVEYFRSATPETEYGRMNIGSRPSKRKPSGGTESLRA1PW1FAWTQTRFHLPV V. planifolia (air moots) FVEYGRLATPELEYGRMNIGSRPSKRKP3GGIESLRAIPWIFAWTQDRFHLPV V. planifolia (CAM leaf) FVEYFRLATPELEYGRMNIGSRPSKRKPSGGIESLRAIPWIFAWTQTRFHLPV FVEYFRLATPETEYGRMN1GSRPSKRKPSGGIUSLRAIPW1FAWTQTRFHLPV FVEYFRSATPETEYGRMNIGSRPAKRRPGGGITTLRAIPWIFSWTQTRFHLPV FVEYFRSATPETEYGRMNIGSRPSKRKPSGGIESLRAIPWIFANTQTRFHLPV
- F. trinervia (C4) G. hírsotum G. max M. crystallinum N. tabacum S. tuberosum S. vulgare Z. mays (C3) Z. mays (C4)
- Z. mays (root)

800 810 820 830 B40 ••• •|•|| WLGFGAAFRYAIKKDVRNLHMLQDMYKQWPFFRVTIDLIEMVFAKGDPGIAAL A. thaliana WLGFGAAFKHAIQKDSKNLQMLQEMYKTWPFFRVTIDLVEMVFAKGNPGIAAL F. trinervia (C4) WLGFGAAFKHVIQKDIKNLHMLQEMHNQWPFFRVTMDLIEMVFAKGDPGIAAL G. hirsetum WLGFGAAFKHVIEKDVRNIHVLQEMYNQWPFFRVTIDLVEMVFAKGDPGIAAL G. max WLGVGGALKHVLEKDIRNHNMLRDMYNNFPFFRVTIDLLEMVFAKGDPEIAAL M. crystallinum N. tabacum WLGFGAAFKYAIDKDIKNLRMFHEMYNEWPFFRVTIDLVEMVFAKGNPGIAAL S. tuberosum WLGFGAAFKYAIEKDIKNLRMLQEMYNAWPFERVTTDLVEMVFAKGDPGIAAL WLGFGAAIKEIMOKDIRNIHVLKEMYNEWPPERVTLDLLEMVFAKGDPGIAAV S. vulgare V. planifolia (air roots) WLGFGTAFKHVSPKDIKNLRILQEMYNEWPFFRVTIDLVEMVFAKGNPGIASL V. planifolia (CAM leaf) WI.GFGAAFKEVISKONKNLHTLREMYNGWPFFRVTIDLIEMVFTKGNPGIASL 2, mays (C3) WLGFGRAFKNVLQKDIRNLHMLQEMYNEWPFFRVTIDLVEMVFAKCNPGIAA5 Z. mays (C4) WEGVGAAFKFAIDKDVRNFQVLKEMYNEWPFFRVTLDLLEMVFAKGDPGIAGL Z. mays (root) WIGFGAAIKHIMQKDIRNIHILREMYNEWPFFRVILDLLEMVFAKGDPGIAAV

A. thaliana

G. hirsutum

N. tabacum S. tuberosum

S. vulgare

Z. mays (C3)

Z. mays (C4) Z. mays (root)

G. max

F. trinervia (C4)

M. crystallinum

850 860 870 880 890 90C | +++| |+ ||+ ++ | |+ |+ |+ |+ |+ | + - | YDKLLVSEDLWAFGEKLRANFDETKNLVLQTAGHKDLLEGDPYLK-QRLRLRD NDKLLVSEDLRPFGESLRANYEETKNYLLKIAGHKDLLEGDPYLK-ØGIRLRD YDKLLVSKELWPFGENLRANYEDTKRLVLQVAGHRDLLEGDPYLK-CRLRLRD YDRULVSEDLMSFGEQLRTMYEETKELLLQVAGHRDLLEGDPYLK-QRLRLRD YDKLLVSEELQSFCERLRANYEDTKRRLLEVAGHKDLLEGDPYLK-CRLRLRD YDKLLVSEDLLPFGELLRSNYEETRSLLLQTAGHKDLLEGDPYLK-QRLRLRD FDKLLVSEDLWSFGELLRSKYEETKSLLLQIAGHEDLLEGDPYLK-QRLRLRD YDKLLVAEDLQ\$FGEQLRKNYEETKELLLQVAGHKDVLEGDPYLK-QRLRLRE V. planifolia (air roots) YGKLLVSSELWSFGERLRANYQETKDLLLQVAGHKVFLK-ESLSE-AEVGLRD V. planifolia (CAM leaf) YDKLLVSSELLPFGAQLRENYGETKRIJLLVAGHKDLLEGDPYLKDQRLRERD YDKLLVSEELHPLGEKLRANYEETQKLLLQVAGHRDLLEGDLYLK-QRLRLRD YDELLVAEELKPFGKQLRDKYVETQQLLLQIAGHKDILEGDPFLKQG-LVLRN YDKLLVADDLQSFGEQLRKNYEETKELLLQVAGHKDVLEGDPYLK-QRLRLRE

		910	920	930	940	950
] • • • • • • • • • •	•••••]]	• •	- L - L - L - L - L - L - L - L - L - L	1
А.	thaliana	SYITTLNVCQAYT	LKRIRDANY	NVTLEPHISE	EIM-QSSKSA	QELVKLNPTSE
F.	trinervia (C4)	PYITTLNVCQAYT	LKRIRDPNY	HVTLRPHISI	(EYAAEPSKPA	DELIHLNPTSE
G.	hirsutum	AYITTLNVCQAY?	LKRIRDPDY	HVKVRPHLSF	EYM-ESSKAA	AELVKLNPTSE
G,	max	SYITTLNVCQAYT	LKRIRDPNY	NVKLRPHISE	(ESI-EISKPA	DELITLNPTSE
М.	crystallinum	PYITTLNVCQAYT	LKRIRDPDF	KVTERPHLSE	(EIM-DAHKAA	AELVKLNPTSE
Ν.	Labacum	SYITTLNLLQAYT	LKRIRDPNY	EVTLRPHISI	CYM-E-SKSA	ALLVQLNPTSE
s.	tuberosum	SYITTLNVCQAY'T	LKRIRDPDY	SVTPRPHISI	KEYM-E-AKPA	TELVNLNPTSE
s.	vulgare	SYITTLNVCQAYT	LKRIRDFSF	QVSPQPPLSB	(EFT-DESQPV	-ELVQLNQQSE
v.	<i>planifolia</i> (air roots)	SYITTLNVCQAYT	LKRIRDPNF	HVKVRPHISI	(E-ISDASKPA	AELVKLNPTSE
v.	planifolia (CAM leaf)	PYITTLNVCOAYT	LKRIREFGY	HVTARPHLLE	(E-TDESIKSA	AELVKLN2TSE
8.	<i>mays</i> (C3)	AYITTLNVCQAYC	LKRIRDPDY	HVALRPHISI	KETM-DSTKAA	ACVVKLNPGSE
8.	mays (C4)	PYITTLNVFQAYT	LKRIRDPNF	KVTPQPFLSH	KEFADENKF	AGLARINDASE
Ζ.	mays (root)	SYITTLNVCQAYT	LKRIRDPSF	QVSPQPPLSE	KEFT-DESQPA	-ELVQUNQQSE

		960 970
		• { • • • • • • • • • • • • • • • •
A.	thaliana	YAPGLEDTLILTMKGIAAGLQNTG
F.	trinervia (C4)	YAPGLEDTLILTMKGIAAGMQNTG
G.	hiroutum	YAPGLEDTLILTMKGIAAGMQNTG
с.	max	YAPGLEDTLILTMKGIAAGLQNTG
М.	crystallinum	YAPGLEDTLILTMKGVAAGLQNTG
N.	tabacum	YAPGLEDTLILTMKGIAAGLQNTG
s.	tuberosum	YAPGLEDTLILTMKGIAAGMQNTG
s.	vulgare	YAPGLEDTLILTMKGIAAGMONTG
v.	planifolia (air roots)	YAPGLEDTLILTMKGIAAGMQNTG
v.	planífolia (CAM leaf)	YGPGLEDTLIITMKGIAAGLQNTG
z.	<i>mays</i> (C3)	YAPGLEDTLILTMKGIAAGLQNTG
z.	mays (C4)	YPPGLEDTLILTMKGIAAGMQNTG
z.	mays (root)	YAPGLEDILILIMKGIAAGMQNTG

 E_{i}

software (Figure 4.5) and this allows the identification of several conserved sites. These are discussed in section 4.3.

The peptide sequence of the *A. thaliana* PEPc cDNA was used to search the Genbank database of *A. thaliana* sequences using BLAST. The search revealed that the sequence reported in this work is the first full-length PEPc cDNA to be cloned from *A. thaliana*. Six months after submission of this sequence to the Genbank database (accession no. AF071788), the genomic sequence of a PEPc gene cloned from young seedlings of *A. thaliana* var. Columbia (accession no. ATH13170) was submitted to the database (Hartung, F., 1998). A gapped alignment of this genomic sequence with the *A. thaliana* PEPc cDNA nucleotide sequence is shown in Figure 4.6. The genomic sequence contains nine introns as do most PEPc genes (Yanai et al., 1994). All the introns have the conserved intron/exon splice sites (GT....AG) observed in the six plant PEPc genes discussed by Lepiniec et al. (1993). Figure 4.7 is a diagrammatic representation of the genomic sequence of the *A. thaliana* PEPc with some other plant PEPcs, highlighting the conserved location of the introns in the genes with respect to the coding region. However, as shown in Table 4.1, the intron length of the genes varies quite considerably (Lepiniec et al., 1993).

	<i>A. thaliana</i> (ATH131710)	M. crystallinum (x14588)	F. trinervia (x64143)	Z. mays (x15642)
intron I (bp)	187	568	833	111
intron II (bp)	85	103	78	111
intron III (bp)	105	1453	92	898
intron IV (bp)	86	530	89	450
intron V (bp)	82	106	79	126
intron VI (bp)	105	324	83	138
intron VII (bp)	172	260	740	97
intron VIII (bp)	82	377	617	98
intron IX (bp)	82	805	605	89

Table 4.1A comparison of intron length from different PEPc genes

Figure 4.6 Alignment of A. thaliana PEPc cDNA and genomic sequences

The numbers in the figure below correspond to nucleotide number. The top line is the PEPc genomic sequence and the bottom line is the PEPc cDNA sequence. Small black circles (•) between the two sequences show identical nucleotides at that point in the alignment. A series of dashes (---) in the genomic sequence represents an intron.



430 1	44 ([) 4:	50	460 I	470 1	48C	490 I	500 [
AGGGAGTGTC	CTAACGAG	STTTGGATCO	CTGGTGACT	CAATTGTT	ATCTCCAAC	GCTTTCTC	ICACATGETT	AACTTAG
AGGGAGTGTC	CTAACGAG	TTTGGATCO	CTGGTGACT	CAATTGTT	ATCTCCAAG	GCTTTCTC	l'CACATGCTT	AACTTAG
30	50	370	380	390	400	410	0 42	0
51 	10	520 	530 	540 	550 I	56() 57 	0
CCAATTIGGO	CIGAGGAGG	STGCAGATTO	GCTCACCGT	CGCAGGAT	CAAGAAGCI	GAAGAAAG:	etgatticgt	TCATGAG
CCAATTTGGC	TGAGGAG	GTGCAGATIC	GCTCACCGT	CGCAGGAT	CAAGAAGCI	GAAGAAAG	GTGATTTÇGT	TGATGAG
 430	 440	 450	 460	 47	0 4	80	 490	 500
580 1	590 1	600 	61.0 	62 I	0 6	530	640 1	650 I
AGTTCTGCAP	ACTACTGA	ATCCGATAT	IGAAGAGAC	TIT T AAGA	GGCTCGTTI	CGGATCTT	GTAAGTCTC	CTGAAGA
AGTTCTGCM	\CTACTGAI	TCCGATAT	IGAAGACAC	TTTTAACA	GGCTCGTTI	CGGATCTT	GTAAGTCTC	CTGAAGA
l	!						I	
510	520	53() 5	40	550	560	570	580
660	670	68() 6	90	700	710	720	730
GATCTTTGAT	I GCCTTGAI	I IGAATCAGAG	 CTGTSGATC	TCGTTTTG.	 ACTGCTCAI	CCTACTCA	I GTCTGTGCGT	I AGATCAT
				******	•••••			******
GATCITIGAI	GCUTTGAA	IGAATCAGAG	JTGTGGATC	TGGTTTTG.	ACTGCTCA1	CCTACTCA	storgreger I	AGATCAT
590) (500	610	620	630	640	650	I.
74() 7	50	760	770	780	790	800	I
TGCTTCAGA4	AGCARGGGZ	GGTTAGTT	 የሞሞናልርሞዊል	 ՅԳ ԳՅՔ ՓՓՓ	 CTGGAGAGI	i Ngtgagaaa	 גריתריתהיתיתה	ՐͲͲႺͲՅϮ
		•			010000	10101010101	1010101110	0110101
TGCTTCAGA	AGCATGGGA	\G						
660	670							
810 	820 I	830 1	840 I	850 I	8 (50 8	37C I	880 I
GTTGACCTG	CATTAAAI	GTTTCTC	AGGATAAGG	GACTGTCT	TGCTCAACT	CTATGCAA	AGGAÇATTAÇ	TCCTGAT
			GATAAGG	GACTGTCT	TGCTCAACI	CTATGCAN	\GGACATTAC	TCCTGAT
				1	<u> </u>	<u> </u>	!	
			680	690	700) 7.	10 7	20
890	900	910	92	0	930	940	950	960
GACAAGCAGO	 FAGCTAGA1	 IGAGTCTCT(GCAAAGAGA	GGTAAGAA	I CTCAACCTC	 ACTEGATT	 STAACATGTT	ן יGATTTGA
GACAAGCAG	GAGCTAGAT	GAGTCTCT	GCAAAGAGA	G				
 730	 740	750	 760					
970	98	30 9	90	1000	1010	1020	1030	1040
 - ምልሞልልምር ርግሥ	ן יייצעייפטטי	ረርሷሞልርኔምም	 ՅՅՈՐՈւդիսենտ	 አይሮምሮምሮሮ	երիչը, ու գետրերին 	 געראייינינאנקעייי	 ጎምኮልጋይምር	 രഹമദ്യവ
~~********	~~~~	╴━╓╷┈╡┝╚┲┇┇						******
						ATTCA	AGCTOCATTC	CGAACAG
							770	780

6.0

MCGAGETTAGAAGAACACCTCCACAGACGAALCTGG MTGGSATTAGAAGAACCTCCACAGCCACAGACGAGACAALCTGG MTGGSATTAGAAGAACCACAGACCTCCACAGACGAGACGACAALCTGG MTGGSATTAGAAGAACACCACGCCACAGACTCGAAGACGAGACG		1050	1060	1070	1080	1090	1100	1110
ATGASHTTAGAAAAAACACCTICAAAACCAACCTICAAGAATAAACAAACCAGAGAAATCTGG 790 800 810 820 820 840 850 860 1120 1130 1140 1150 1160 1170 1180 1290 AAAGSTGTCCCCAASTTCTTGCGCCGGTGGACACACCTCTGGAAAACGTTGGATGGA	ADGAGADT	I AGAAGAAC	ACCTCCAACCC	CACAAGATG	NANTGAGAGC	TGGAATGAGI	I TATTTCCACCA	GACAATCTGG
190 800 810 820 820 840 850 860 1120 1130 1140 1150 1160 1170 1180 1190 AAAGGTGTCCCCAAGTTCTTGCGCCGTGTGGGACAGCGCTTGAAAAACATTGGGATTGATGAACGTGTTCCTTACAA AAAGGTGTCCCCAAGTTCTTGCGCCGTGTGGACAGCGCTGAAAAACATTGGGATTGATGAACGTGTTCCTTACAA AAAGGTGTCCCCAAGTTCTTGCGCCGTGTGGACAGCGCTGTGAAAAACATTGGGATTGATGAACGTGTTCCTTACAA 870 880 890 900 910 920 930 1200 1210 1220 1230 1240 1250 1260 1270 7GCCCATTGATTGATTCATTCTTTGTGTGGATGGGGGGGGG	ATGAGATT	AGAAGAAC	ACCTCCAACCO	CACAAGATG	AAATGACAGC	TGGAATGAGI	TATTTCCACGA	GACAAICIGG
112.0 113.0 114.0 115.0 116.0 117.0 118.0 11.90 AAASGTGTCCCCCAAGTTCTTGCGCCCGTGTGGACAACACTCTGAAAAACATTGGAATTGAAGAGGTGTTCCTTACAA AAASGTGTCCCCCAAGTTCTTGCGCCGTGTCGGACACACCTCGAAAAACATTGGGATTGATGAACGTGTTCCTTACAA AAASGTGTCCCCCAAGTTCTTGCGCCGTGTCGGACACACCTCGAAAAACATTGGGATTGATGAACGTGTTCCTTACAA AAASGTGTCCCCAAGTTCTTGCGCCGCGTGTCGGCGGTGATCGTGAAAAACATTGGGATTGATGAACGTGTTCCTTACAA AAASGTGTCCCCAAGTTCTTGCTTCGTGGAACGCGGTGATCGTGATGGTGACATGGGTTTTTTTGGATTACTGGGCGGTGATCGTGATGTGGCTTTTTTTGGATTACTGGGGGGGG	 790	 801	0 810	 	 8.3 0	0 84	0 850	 860
1120 1130 1140 1150 1160 1170 1180 1190 MAGGTGTCCCCCAAGTTCTTGCGCCGTGTGGACAGAGCTCGTGAAAAACATTGGGATTGATGAACGTGTTCCTTACAA AAAGTGTCCCCAAGTTCTTGCGCCGTGTGGACAGAGCTCGTGAAAAACATTGGGATTGATGAACGTGTTCCTTACAA B70 880 890 900 910 920 930 1200 1210 1220 1230 1240 1250 1260 1270 TGCCCCATTGATTCAATTCCTTCGTGGATGGGCGGTGACCGTGATGGTGCAACGTGCTACATGCGTTTTTTGGATTACTGGCGAAA 1320 1330 1340 1280 1290 1300 1310 1320 1330 1340 1280 1290 1300 1310 1320 1330 1340 1350 1360 1370 1380 1390 1400 1410 1420 1350 1360 1370 1380 1390 1400 1410 1420 1350 1360 1370 1380 1390 1400 1410 1420 1350 1360 1370 1380 1390 1400 <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>000</td>								000
AAAGGTGTCCCCAAGGTCTTGCCCCGCGTGGGACAGGGCTCTGGAAAAACATTGGGATTGATGAACGTGTTCCTTACAA AAAGGTGTCCCCAAGGTCTTGCCCCGCGGTGGACAGGCTCTGGAAAAACATTGGGATTGATGAACGTGTTCCTTACAA 970 880 950 960 910 920 930 1200 1210 1220 1230 1240 1250 1260 1270 TGCCCCATTGATTCAATTCTTTCGTGGATGGCGGGGGGGG	1120	1130	1140	1150	1160	1170	1180	1190
ARAGGTGTCCCCARAGTTCTGCGCCCTCTCGGCACCGCCCTCCGCARAGACTCGGGGTTTGATGACGTGTTCCTTTGCTTGCGCCCTCTGGGCGGGGTGATCGTGATGGGGGGTGTTTTTGGATGGTTGGT	AAAGGTGI	 'CCCCAAGT'	I TCTTGCGCCGT	 GTGGACACA	I GCTCTGAAAA	 ACATTGGGAI	 TGATGAACGTG	 FTCCTTACAA
970 880 890 900 910 920 930 1200 1210 1220 1230 1240 1250 1260 1270 TGCCCCATTGATTCATTCTTTCGEGATGGGGGGGGGATCGTGATCGTGATGGTACTGGGTTTTTTGGATTACIGGATA 1	AAAGGTGT	·······			SCTCTGAAAA	асаттассая	TRATICALCON	
870 880 900 900 910 920 930 1200 1210 1220 1230 1240 1250 1260 1270 TGCCCCATTGATTCATTCTCTGEGGATCGGCGGTGATCGTGGATCGTGGGTGGCGTTTTTTTGGATTACTGGCATCGTGGATGGGGTGATCGTGGGTTTTTGGATTACTGGATTACTGGCATCGTGGGTTGGTGGGTTGGTGGGGGGGG		70						
1200 1210 1220 1230 1240 1250 1260 1270 TGCCCATTGATTCATTCCTTCGEGGATCGEGGGGGGGTACCGTCGATGGGTACGTGGGGTGATCGTCGTGGGTTTTTTGGATTACTGGGATA 1 1 1 1 940 950 960 970 980 1320 1330 1340 1280 1290 13C0 1310 1420 1420 1420 1350 1360 1370 1380 1390 1400 1410 1420 1310 1020 1030 104C 1050 106C 1070 1080 1430 1440 1450 1460 1470 1480 1490 1500	8	70	88Ç	890	900	910	920	930
TGCCCATTGATTCATTCTTTCTGGGGGGGGGGGATCGGGCGTGATCGTGGTGGGCGGTTTTTTGGATTACTGGCTACTGGGGGGGTGATCGGGGTGGTTTTTTGGATTACTGGCATTCGGGGTGGGGGGGG	120	0 1	210 12	20 1:	230 1	.240 1	.250 12	:60 1270
TGCCCCATTGATTCCATTCCTTCGTGGATGGGCGGCGTGATCGTGATC 940 950 960 970 980 1280 1290 1300 1310 1320 1330 1340 CGANTCACTCTCTTGGTTTCTCTIGGTCACCCTCTAATCGTCGTATCTGAATTCCAGGGATCCCAACCT	J TGCCCCAT	 TGATTCAA	 TTCTCTTCGTC	 GATCCCCCC	 IGATCGTGAT	 GGTACATGCG	 STTTTTTTGGAI	I TACIGGCAAA
940 950 960 970 980 1280 1290 1300 1310 1320 1330 1340 CGAATCACTCTCTTGGTTTGTCT1CGTCACCCTCTAATCGTCGTATCTGAATTCCGAGGGTCACACCT	TGCCCCAT	тсаттсаа	·····	GATEGOOGO	PGN/PCGTCAT	•		
940 950 950 970 980 1280 1290 1300 1310 1320 1330 1340	10000011							
1280 1290 1300 1310 1320 1330 1340 CGANTCACTCTCTTGGTTTCTCTTGGTCACCCTCTAATCGTGATTTCAGGTAATCCGAGGGTCACACCT	940	950	960	970	980			
CGANTCACTCTCTTGGTTTCTCTTGGTCACCCTCTAATCGTCGTACTCTGAATTTCAGGTAATCCGAGGGTCACACCT		1280	1290	1300	1310	1320	1330	1340
COMMENCICICITISTICATE CONCECTERATION CONTROL CONSTRUCT	000000000	 	 	 	 ¤################################	 CTCD D D D D D D D D D		
	COMMON	.10101130	11101011001	CACCULCIA	RICOLOGIAI	CIGMITICF	**********	GGICACACCT
990 1000 1350 1360 1370 1380 1390 1400 1410 1420 CAGGTCACTAGAGATGTGGTGGTTGTTGGCTAGATGTGGCTGCCAATCTCACTATAACCAAATCGAGAATCTGAT SACGTCACTACAGAIGTGGTGGCTTGTTGGCTAGGATGATGGCTGCCAATCTCACTATAACCAAATCGAGAATCTGAAT SACGTCACTACAGAIGTGGTGGTGTGGTGGGCTAGGATGGCTGCCAATCTCACTATAACCAAATCGAGAATCTGAATCTGAT SACGTCACTACAGAIGTGGTGGCTTGGCTAGGATGGATGGCTGCCAATCTCACTATAACCAAATCGAGAATCTGAAT SACGTCACTACAGAIGTTGTGTGGCTTGGCAAGGATGGCTGCCAATCTCACTACTAAACCAAATCGAGAATCTGAAT 1010 1020 1030 1040 1050 1060 1070 1080 1430 1440 1450 1460 1470 1460 1490 1500 1430 1440 1450 1460 1470 1460 1490 1500 1430 1440 1450 1460 1470 1460 1490 1500 167TTGAG 1 1 1 1 1 1 1 1 1510 1520 1530 1540 1550 1560 1570 1 1 1 1 1 1 1 1 1 1 1			- -				GTAATCCGAG 	IGGTCACACCT
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$							990	1.000
1500 1500 1500 1500 1400 1000 1100	1350	1360	1370	1380	1390	1400	1410	1420
CAGGTCACTAGACATGTGTGTGTGGCTTGTTGGCTAGAATGATGACTGCCCACTGTCACTACTACCACAATCGAGAATCTGAT SACGTCACTACACAATGTGGTGTGTGGCTTGTGTGGCTAGAATGATGATGATGGTGGCAAATCGAGAATCTGAT SACGTCACTACACAATGTGGTGTGGGCTTGGCTAGAATGAGAATGTGATGATGCACAAATCGAGAATCTGAT SACGTCACTACACAAATGTGGGCTTGTGGCTAGAATGACGAAATGGCAGGAACTGGAA SACGTCACTACACAATGATGGCTTGGCTAGAATGACGAAATGGCAGGAACTGGCAGGAACTGCACAGGAACTGCACTGGCAGGGATGAACTGCACAGGAACTGCACTGGCAGGGATGAACTGCACAGGAACTGCACAGGAACTCCAA SACGTCACTAGTGTGCGGGGCGGGATGAACTGCACAGGGAACTCCAA SACGTCTGCACTGCACTGCACTGATGAATTCCGTGTGCGGGCGG		:				1100	1	
SAGGTCACIACAGAIGTGTGCTTGTTGGCTAGGATGATGGCTGCCAATCTCACTATAACCAAATCGAGAATCTGAT 1	GAGGTCAC	AGAGATG	TGTGCTTGTTG	GCTAGAATG.	ATGGCTGCCA	ATCTCTACTA	VTAACCAAATC(GAGAATCIGAT
1010 1020 1030 1040 1050 1060 1070 1080 1430 1440 1450 1460 1470 1480 1490 1500 GTTTGAGGTTTTTATCTCTTTTTACTTGTACTCATTCCCGGTTTTCATTCTTTAAGATTCTCTTGATTCATTC	GAGGTCAC	TACAGAIG	TGTGCTTGTTG i	GCTAGAATG.	ATGGCTGCCA	ATCTCTACTA	ATAACCAAATCG	SAGAATCTGAT
1430 1440 1450 1460 1470 1460 1490 1500 GTTTGAGGTTTTTACTTGTACTCGTTTTCCCGGGTTTTCATTCTTTAAGATTCTCTTGATTCATTC	1010	102	o 1030	104	c 105	0 100	50 iova) 1.080
1430 1440 1450 1460 1460 1490 1500 i i i i i i i i i GTTTGAGGTTTTTACTCTTTTTACTTGTACTCATTCCCGGGTTTCATTCTTTAAGATTCTCTTGAFTCATTCGTG GTTTGAG i								
$\begin{array}{c} GTTTGAGGTTTTTATCTCTTTTTACTTGTACTCATTTCCCGGTTTTCATTCTTTAAGATTCTCTTGATTCATTC$	1430 I) 14 	40 14: I	50 14 	60 14 I	170 14	180 149	90 1500 I
GTTTGAG 1.51.0 1.520 1530 1540 1550 1560 1570 TTCTTCTTGGCAGTTATCTATGTGGCGTTGCACTGATGAATTCCGTGTGCGGGCGG	GTTTGAGG	TTTTTATC	TCTTTTTACTI	'GTACTCATT	TCCCGGTTTT	CATTCTTTA	AGATTCTCTTG	ATTCATTCGTG
151.0 1520 1530 1540 1550 1560 1570 TTCTTCTTGGCAGTTATCTATGTGGCGTTGCACTGATGAATTCCGTGTGCGGGGCGGATGAACTGCACAGGAACTCAA TTATCTATGTGGCGTTGCACTGATGAATTCCGTGTGCGGGCGG	GTTTGAG-	·						
151.0 1520 1530 1540 1550 1560 1570 I I I I I I I I I TTCTTCTTGGCAGTTATCTATGTGGCGTTGCACTGATGAATTCCGTGTGCGGGGCGGATGAACTGCACAGGAACTCAA	-							
TTCTTCTTGGCAGTTATCTATGTGGCGTTGCACTGATGAATTCCGTGTGCGGGCGG	ز ا	.51.0	1.520	1530 I	1540 	1550 I	1560 	1570
TTATCTATGTGGCGTTGCACTGATGAATTCCGTGTGCGGGCGG	TTCTTCTT	GGCAGTTA	TCTATGTGGCC	TTGCACTGA	TGAATTCCGI	GTGCGGGCGG	JATGAACTGCA(CAGGAACTCAA
1090 100 110 120 130 1140 1150 1580 1590 1600 1610 1620 1630 1640 1650 1 1 1 1 1 1 1 1 GGAAAGATGCTGCAAAACATTACATAGGTTAGAAGCTTTGGACA1TTTCTTAAAAGAAACACATACTGCAAGTA GGAAAGATGCTGCAAAACATTACATAG 1 1 1 1 1 1 1 1 1 1 1 1 GGAAAGATGCTGCAAAACATTACATAGGTAGAAGCTTTGGACA1TTTCTTAAAAGAAACACATACTGCAAGTA 1 1 1 1 1 1 1		ATT	TCTATGTGGCC	STTGCACTGA	TGAATTCCGI	TETECESÇCE	SATGAACTGCAC	CAGGAACTCAA
1580 1590 1600 1610 1620 1630 1640 1650 GGAAAGATGCTGCAAAACATTACATAGGTTAGAAGCTTTGGACA1TTTCTTCTTAAAAGAAACACATACTGCAAGTA GGAAAGATGCTGCAAAACATTACATAG		1090	1100	 1110	1120) 1130) 1140	1150
1580 1590 1600 1610 1620 1630 1640 1650 GGAAAGATGCTGCAAAACATTACATAGGTTAGAAGCTTTGGACATTFTCTTCTTAAAAGAAACACATACTGCAAGTA GGAAAGATGCTGCAAAACATTACATAG								
GGAAAGATGCTGCAAAACATTACATAG	1580	1590	1600	1610	1620	1630	1640	1650
GGAAAGATGCTGCAAAACATTACATAG	 GCAAAGA3	 Igctgcaaa	 ACATTACATAG	 GTTAGAAGC	 TTTGGAÇA'I'I	 FTCTTCTTA	 \AAGAAACACA1	I FACTGCAAGTA
	GGAAAGA1	GCTGCAAA	ACATTACATAC	- 				
		 	170	_				

i,

,

1660	1670 1	1680) î. 6	590	1700 	1710 I	1720 	1730
TACAATGGI	ATIGGTT	TGCATTCTT	ACTTATT1	TTTTTCTT		GCAGAATT	CTGGAAGACA CTGGAAGACA	ATTCCTC
						11180	1190	 1200
174	10 1	750	1760	1770	1780	1790	1800	1810
CAACTGAGC	CATACCGT CATACCGT	GTGATTCTTC GTGATTCTTC	GTGATGTG GTGATGTG	agggataa Agggataa	GCTGTATCA GCTGTATCA	CACACGTG	AGOGTTOCOG	CCAATTG
12	210	1220	1230	1240	1230	1.26	0 127	0
	1820	1830	1840	1.850	186	0 1	870 1	880
CTGAGTAAT	IGGAATCTC	GGATATTCC GGATATTCC	I IGAAGAAGQ IGAAGAAGQ	TACCTTCA	CTAATGTCC	AACAGGTG	AGCAACCTC1	'A''ACTCT
 1,280	 1290	l 1300	 1310	 132	0 13	30		
1890	1900	1910	192	20 1	930	1940	1950	1960
GAATTTGT	ICCTTACTA	AATTGTTTC	TGATTT	ACAACTTG	TTAGITCTI	i GCCATTAG	AGTTTCTATC	ATTATCC
1970 I AGCCTTAC) 19 : : : : : : : : : : : : : : : : : : :	80 1: :CCTCAATAA	990 IATTCACGO	2000 ;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;	2010 TGTGTTTAC	2020 TGTECTTT	2030 TATTCTA TT 	2040 !ATGCAGT T
TCTTGGAG	2050 CTCTTGAG	2060 CTCTCTTAC	2070 : CGATCACT7	2080 ATGTTCATG	2090 TGGTGACAG) 21 GCCCGATAG	00 21 :CTGATGGAAG	.10 SCCTTCTT
ICTIGGAG	CTCTTGAG	CTCTGTTAC	CGATCACT	ATGTTCATG	TGGTGACAC	CCCGATAG	CTGATGGAAG	SCCTTCTT
1340	1350	1360	131	70 1	380	1390	1400	1410
2120 	2130 	2140 	215: I) <u>2</u> 1	60 2	21.70	2180 	2190
GATTTCTT(GATTTCTT)	GAGGCAAGI GAGGCAAGI	CTCTACCTT	TGGACICT(TGGACICT(CCCTTGTGA CCCTTGTGA	GACTUGACA GACTUGACA	ATCAGGCAA ATCAGGCAA	GAGTCTGAAC GAGTCTGAAC	:GCCACAC IGCCACAC
 142) 0 1.4	30 1	440	[1450	 1460	 1470	 1480	1490
2200 AGATGTCT AGATGTCT	221 J FGGATGCIA EGGATGCIA	.0 22 I ATCACCAAGC.	20 ; ACTTGGAC: ACTTGGAC:	2230 ATCGGTTCC ATCGGTTCC	2240 1 TCCTATAGA	2250 AGACTGGTC AGACTGGTC	2260 TGAAGAAGGO TGAAGAAGGO	2270 ł CCGACAGG
	1500	1510	1520	1530	1540) 15	50 15	560

 2

ş

228	0 2	290	2300	2310	2320	2330	2340	
i AATGGCTTCT	TCCTCAAC	TAAGCCCCC	I AACGTCCA	CTTTTCGGA0	l CCTGATCTI	I DAAAADDD1	CGAAGAAATT	TCTGAT
AATGGCTTCT	TGCTGAAC	TAAGCGGCA	AACGTCCA	CTTTTCGGA(CCTGATCT?	CCCAAAAC	CGAAGAAATT	TCTGAT
 1570	l 1580	 1590	 1600	 161() 16	520	: 1630	 1640
				·				
2350	2360	2370	2380	2390	240)0 2	410 2	420
I GTCCTGGACA	I CATTCAAA	GTCATATCI	GAGCTGCC	TTCAGATTG	TTTGGAG	- TTATATTA	I TCTCTATGGC.	AACTTC
GTCCTGGACA	CATTCAAA	GTCATATCI	IGAGCTGCC	TTCAGATTG:	I'TT' T GGAG(TTATATTA	TCTCTATCGC	AACTTC
 1650	 1660	 161	/0 1	680 -	 690	 1 70 0	 1710	 1720
2430	2440	2450) 24	60 2·	170	2480	2490	2500
ACCTAGTGAT	GTGCTTGC	GGTTGAGCI	TTTACAGC	GCGAATGCC	\TGTGAAA A	ATCCACTT	'AGAGTTGTTC	CACTOP
ACCTAGTGAT	GTGCTTGC	GGTTGAGCI	PTTTACAGC	GCGAATGCC/	ATGTGAAA#	ATCCACTI	AGACTTGTTC	CACTCT
 1,73	10 1	740	 1750	 1760	 1770	 1780	, 1790	1
2510) 25	20 2	2530	2540	2550	2560	2570	2580
TTTGAGAAGCT	AGCTGATC	TTGAAGCA	; GCTCCTGCC	GCTGTTGCA	I AGACTCTT!	I ITCTATAGN	LCTGGTACAAA	AACCGT
TTGAGAAGCI	AGCTGATC	TTGAAGCAG	SCICCIGCC	GCTGTTGCA	AGACTCTT:	PTCTATAGA	CTGGTACAAF	AACCGT
1900	 1910	1820	 1830	1840	181	50 I	860 I	870
1990	1010	1020	1000	1040	10.	,0 I	.000	0.0
2	2590	2600	2610	2620	2630	26	340 26	50
ATTAACGGTA	AACAAGAG	; CTTATGAT.	 FGGTTACTC	; AGATTCAGG	 GAAAGATG(CAGGGCGTC	 TCTCAGCTGC	TTGGGA
ATTAACGGTA	AACAAGAG	GTTATGAT	rggttactc	AGATTCAGG	GAAAGATG	CAGGGCGTC	TCTCAGCTGC	TTOGGA
1 990	1000			 		 1930	: 1940	1950
1000	1030	130	, T3	10 H	220	1990	7040	1000
2660								
 GCTATACAAA	2670	2680	269	0 27	00 :	2710	2720	2730
	2670 AGCTCAAGA	2680 AGAGCTTG	269 IGAAGGTTC	0 27 CTAAGAAAT	00 : ATGGAGTG	2710 MAGCTAACI	2720 !ATGTTCCATC	2730 ;600020
GCTATACAAA	2670 AGCTCAAGA	2680 AGAGCTTG AGAGCTTS	269 IGAAGGTTC IGAAGGTTC	0 27 	00 : ATGGAGTG/ ATGGAGTG/	2710 NAGCTAACI	2720 ATGTTCCATC	2730 GCCC22G
GCTATACAAA	2670 AGCTCAAGA AGCTCAAGA 	2680 I AGAGCTTG AGAGCTTG	269 IGAAGGTTC 000	0 27 	00 : ATGGAGTG/ ATGGAGTG/ 1 2200	2710 MAGCTAACI AAGCTAACI 2010	2720 ATGTTCCATC ATGTTCCATC	2730
GCTATACAAA 1960	2670 AGCTCAAGA AGCTCAAGA)9	2680 AGAGCTTG AGAGCTTG 70	269 IGAAGGTTG IGAAGGTTG 1980	0 27 CTAAGAAAT. CTAAGAAAT. 1990	00 ; ATGGAGTG/ ATGGAGTG/ 1 2000	2710 AAGCTAACI AAGCTAACI 2010	2720 IATGTTCCATC ATGTTCCATC 2020	2730 GCCCC2G GCCGTG 2030
GCTATACAAA 1960 2740	2670 AGCTCAAGA AGCTCAAGA)9 275	2680 I AGAGCTTG AGAGCTTG 70 2	269 IGAAGGTTC IGAAGGTTC 1980 760	0 27 icTTAAGAAAT. icTTAAGAAAT. iCTTAAGAAAT. i 1990 2770	00 : ATGGAGTGJ ATGGAGTGJ 1 2000 2780	2710 AAGCTAACI AAGCTAACI 2010 2790	2720 I ATGTTCCATO ATGTTCCATO 2020 2800	2730
GCTATACAAA 1960 2740 GTGGCACAG9	2670 AGCTCAAGA AGCTCAAGA 09 275 PCGCAAGAG	2680 AGAGCTTG AGAGCTTG 70 2 0 2 GAGGTGGT	269 IGAAGGTTG IGAAGGTTG 1980 760 2CTACTCAT	0 27 CTAAGAAAT. CTAAGAAAT. 1990 2770 CTTGCTATA	00 : ATGGAGTGA ATGGAGTGA 1 2000 2780 1 TTGICTCA	2710 AAGCTAACI AAGCTAACI 2010 2790 GCCACCAGA	2720 ATGTTCCATC ATGTTCCATC 2020 2800 ATACAGTTAAT	2730
GCTATACAAA 1960 2740 GTGGCACAG1	2670 AGCTCAAGA AGCTCAAGA D9 275 PCCCAAGAG	2680 AGAGCTTG AGAGCTTG 70 2 0 2 GAGGTGGT	269 IGAAGGTTC IGAAGGTTC 1980 760 CCTACTCAT	0 27 ictaagaaat. ictaagaaat. 1 1990 2770 i 'CTIGCIATA	00 : ATGGAGTGA ATGGAGTGA 1 2000 2780 1 TTGICTCA	2710 AAGCTAACI AAGCTAACI I 2010 2790 I GCCACCAGA	2720 IATGTTCCATO :ATGTTCCATO i 2020 2800 ITACAGTTAA	2730
GCTATACAAA 1960 2740 GTGGCACAGI GTGGCACAGI	2670 AGCTCAAGA AGCTCAAGA)9 275 PCCCAAGAG CCGGAAGAG	2680 I AGAGCTTG AGAGCTTG 70 2 0 2 I GAGGTGGT I GAGGTGGT	269 IGAAGGTTG IGAAGGTTG 1980 760 CCTACTCAI CCTACTCAI	0 27 CTAAGAAAT CTAAGAAAT 1990 277C CTTGCTATA CTTGCTATA	00 : ATGGAGTG ATGGAGTG 2000 2780 TTGICTCA TTGICTCA	2710 AAGCTAACI AAGCTAACI 2010 2790 3CCACCAGA 3CCACCAGA	2720 PATGTTCCATC PATGTTCCATC 2020 2800 VTACAGTTAAN	2730
GCTATACAAA 1960 2740 GTGGCACAG1 GTGGCACAG1	2670 AGCTCAAGA AGCTCAAGA 09 275 PCCCAAGAG PCCCAAGAG 2040	2680 I AGAGCTTG AGAGCTTG 70 2 0 2 I GAGGTGGT GAGGTGGT I 2050	269 GAAGGTTG 1980 760 CCTACTCAI 2060	0 27 i CTAAGAAAT i 1990 2770 2770 CTTGCTATA CTTGCTATA 2070	00 : ATGGAGTG ATGGAGTG 1 2000 2780 1 TTGICTCA TTGICTCA 1 208	2710 AAGCTAACI 2010 2790 GCCACCAGA GCCACCAGA	2720 INTETTCCATC ATETTCCATC 2020 2800 I ATACAGTTAAT ATACAGTTAAT I)90 2:	2730 GCCGTG 2030 2810 GGCTCT .00
GCTATACAAA 1960 2740 GTGGCACAGI GTGGCACAGI	2670 AGCTCAAGA AGCTCAAGA) 275 PCCCAAGAGAG PCCCAAGAGAG 2040 820	2680 I AGAGCTTS AGAGCTTS 70 20 20 CAGGTGGT I 2050 2830	269 GAAGGTTC 1980 760 CCTACTCAT CCTACTCAT 2060 2840	0 27 CTAAGAAAT CTAAGAAAT 1990 2770 2770 CTTGCTATA CTTGCTATA 2070 2350	00 ATGGAGTG ATGGAGTG 1 2000 2780 1 TTGICTCA TTGICTCA 208 2860	2710 AAGCTAAC1 2010 2790 GCCACCAGA GCCACCAGA 0 20	2720 IATGTTCCAT(IATGTTCCAT(2020 2800 I ATACAGTTAA I 190 281 282 282 2800 200 2	2730 GCCGTG 2030 2810 FGGCTCT .00
GCTATACAAA 1960 2740 GTGGCACAGT GTGGCACAGT 220 24 24 24	2670 AGCTCAAGA AGCTCAAGA 275 CCCCAAGAG CCCCAAGAG CCCGAAGAG 2040 820	2680 AGAGCTTG AGAGCTTG 70 20 20 20 20 20 20 20 20 20 20 20 20 20	269 IGAAGGTTG IGAAGGTTG 1980 760 CCTACTCAI 2060 2840	0 27 CTAAGAAAT CTAAGAAAT 1990 277C CTTGCTATA CTTGCTATA 2070 2350	00 : ATGGAGTG ATGGAGTG 2000 2780 TTGICTCA TTGICTCA 208 2860 1	2710 AAGCTAAC1 AAGCTAAC1 2010 2790 3CCACCAGA 0 20 287 0 20	2720 IATGTTCCATC ATGTTCCATC 2020 2800 I ATACAGTTAAT 1090 21 10 288 I	2730 GCCCCCC GCCCCCC 2030 2810 CGCCTCT CGCCTCT .00
GCTATACAAA 1960 2740 GTGGCACAGT GTGGCACAGT 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	2670 AGCTCAAGA AGCTCAAGA 275 275 275 275 2040 820 ACGGTTCAG	2680 1 AGAGCTTG AGAGCTTG 70 20 20 20 28 30 28 30 1 30 26 26 26 26 26 26 26 26 26 26	269 GAAGGTTC 1980 760 CCTACTCAI 2060 2840 2840 CATTGAGCI	0 27 CTAAGAAAT CTAAGAAAT CTAAGAAAT 1990 2770 2770 CTTGCTATA CTTGCTATA 2070 2350 1 ATCATTEGG	00 ATGGAGTGI ATGGAGTGI 2000 2780 1 TTGICTCA TTGICTCA 208 2860 1 2860 1 2860	2710 AAGCTAACI 2010 2790 GCCACCAGA 0 20 287 ACTTATGCJ	2720 RATGTTCCAT(2020 2800 ATACAGTTAAT ATACAGTTAAT)90 23 / / / / / / / / / / / / / / / / / / /	2730 GCCGTG GCCGTG 2030 2810 TGGCTCT IGGCTCT 100 30 TTCAACG
GCTATACAAA 1960 2740 GTGSCACAGI GTGGCACAGI GTGGCACAGI 2 CTTCGAGTCA CTTCGAGTCA	2670 AGCTCAAGA AGCTCAAGA 275 2040 220 200 	2680 1 AGAGCTTG AGAGCTTG 70 20 20 20 20 20 20 20 20 20 2	269 IGAAGGTTC IGAAGGTTC 1980 760 CCTACTCAI CCTACTCAI 2060 2840 I CATTGAGCI	0 27 CTAAGAAAT CTAAGAAAT 1990 277C CTTGCTATA CTTGCTATA 2070 2350 1 ATCATTEGG	00 : ATGGAGTG ATGGAGTG 2000 2780 1 TTGICTCA 1 208 2860 1 GGAGGCAC GGAGGCAC	2710 AAGCTAACI 2010 2790 3CCACCAGA 3CCACCAGA 0 20 285 i ACTTATGCJ ACTTATGCJ	2720 IATGTTCCATC IATGTTCCATC 2020 2800 I ATACAGTTAAT I 190 280 I ITAGAACAC I ITTAGAACAC	2730 GCCCCCC GCCCCCC 2030 2810 CGCCTCT CGCTCT 100 30 FTCAACG ITCAACG

2890 TITCACACCA TTTCACAGCA	2900 AGCTACTCTAC	2910 EAGCACGGAAT	2920 GAACCCTC	2930 i CGATTTCA CGATTTCA	2940 ССААААСССС ССААААСССС	2950 I GAGTGGCGTG) 2960 GCTTTGCTTGATG
: 2190	 2200	 2210	 222	0 2:	230	 2240	2250 2260
2970 I AAATGGCGG1 AAATGGCCGT	2930 TGTTGCAAC TGTTGCAAC	2990 IGAGGAATACC IGAGGAATACC	3000 GATCTGTC GATCTGTC	30: J GTTTTCCA, GTTTTCCA;	10 3 AGAACCTCG	020 : ATTCGTCGA(ATTCGTCGA(3030 3040 J STATTTCCGCCTC
22	270 22	280 22	90	2300	2310	2320	2330
305 GTAAGTATTC	0 30) CATTCATATCA	50 307 +ATCCAAATTT	0 3 CACCAAAG	080 TTTTTGTG	3090 TGTITTAAG	3100 TACCTAAAA(3110 CCGTTGGTTTGGT
3120 ITGCAGGCTA	3130 \CTCCGGAEC'	3140 GGAGTATGGA	3150 CGTATGAA	3160 I TATTGGAA	3170 I GTAGACCTT GTAGACCTT	3180 i CAAAGCGAA CAAAGCGAA	3190 AACCAAGCGGTGG AACCAAGCGGTGG
 234	1 1 23	50 236	i0 2	370	1 2380	 2390	 2400
3200 GATCGAAIC1 GATCGAAIC1 2410	3210 rctccgtgca, rctccgtcca; 2420	3220 ATCCCATGGAT ATCCCATGGAT I 2430	3230 CCTTTGCTT CCTTTGCTT 2440	324 J GGACGCAA GGACGCAA 2450	0 32 ACAAGATTC ACAAGATTC I 2460	50 3: CATCTTCCT(CATCTTCCT) 2470	260 3270 JTATGGTTAGGTT GTATGGTTAGGTT GTATGGTTAGGTT 2480
3280 TCGGAGCAGO TCGGAGCAGO 2490) 329 CATTTAGGTA CATTTAGGTA 2500	0 3300 j IGCGATCAAGA IGCGATCAAGA 2510) 33 HAGGATGTG AAGGATGTG 2520	10 AGAAACCT AGAAACCT ! 253	3320 I TCACATGCT TCACATGCT I 0 25	3330 GCAAGATAT GCAAGATAT 40 2	3340 3350 STATAAACAATGG GTATAAACAATGG 550 2560
COOTTITTCC COOTTITTCC COOTTITTCC I 2570	3360 CGAGTCACCA CGAGTCACCA I D 258	3370 3 , TCGATCTAAT TCGATCTAAT I 0 2590	3380 CGAAATGGT CGAAATGGT) 2 2 6	3390 I CITEGECA GTTEGECA	3400 AGGGAGACC AGGGAGACC 2610	3410 J CCGGGATCG CCGGGATCG J 2620	3420 I CTGCTTTGTACGA CTGCTTTGTACGA I I 2630 2640
3430 L CAAACTICT CAAACTICT	3440 retetcagaa estetcagaa 2650	3450 GATTTATGGGG GATTTATGGGG 2660 2	3460 CTTTTGGAG CTTTTGGAG 2670	347) I Agaaacic Agaaacic I 2690	0 34 - AGAGCCAAC - 2690	80 34 TTTGATGAA TTTIGATGAA 1 2700	190 3500 I ACCAAGAACCTCG ACCAAGAACCTCG I 2710

: Ø)

3510	352	20 33	530	3540	355	50 B	560	3570	3580
TCCTCCAGGT	 ATAACAT#	I ACAAAAAA	CAGGGTCC	 TCTGTT'	CATATTI	 GTIGGGTI	TGTCTTGC	I AATAACA(CTAAAC
******** "POCTOCAG									
2720									
35	90	3600	3610	362	20	3630	3640	365/	0
	രനാനതന്നാം					 າດຫຼາວ ດາ ຫຼາວ			ግጠ እር ስለካጥ
FIGCGTTGTI	GCAGACTO	SCTGGACAT	MAAGACC1	TOTIGA	AGGAGAT(********	* 4 * • • • • •	UTAUG3
	ACTO	CTGGACAT	AAAGACCI	TCTTGAZ	AGGAGATO	CCTTACTTO	AAACAGAG	ACTAAGG(CTACGT
		2730	2740	279	50	2760	2770	278	0
								_	
3660 	3670	3680 I	365	0	3700	3710	372	.0 .	3730
GACTCTTACA	TTACGACO	CTCAACGT	ITGCCAAG	CCTACAC	CATTGAA	GAGGATCCO	TGATGCAA	ACTACAA	TGTGAC
GACTCTTACA	TTACGACO	CTCAACGT	ITGCCAAG	CCTACA	CATTCAA	CAGGATCCC	TGATGCAA	ACTACAA:	rgugac
 2790	2800	 2810	 282	:0	2830	2840	 285	io :	 2860
3740	3750) 37	60	3770	3783	o <u>37</u>	90	3800	3810
I TCTGCGACCA	 CACATTTO	TAAAGAGA'	FCATGCA	I TCAAGCA	I AAATCAG	l Cacaagag(TCGTCAAG	 	CCACGA
TCTGCGACCA	•••••••• САСАТТТС	тааасаса:	• • • • • • • • • FCATCCAR		AAPCAG	ACAAGAGC	PEGTCAAG	CTTAACC	CCACGA
					1				1
2870	2880	28	90	2900	2910	1 25	120	2930	2940
382	0 3	3830	3840	385:) :	3860	3870	3880	
	<u>ം</u> ന്നുവരും] מידי א רידי שים איז	ן ריקידאסרי	787728863	 2621F234PPCC5	 	ן הבגנגרסי	BOCGGT.
4444444444	•••••••••			••••			• • • • • • • •		•••••
GTGAATACGC	GCCTGGAG	CTTGAGGAÇ)	ACACTTAJ	CITAACO I	CATGAAG	gtattoc: 	GCAGGATI	GCMMACI	ACCGGT
295	0 2	2960	2970	298	D 2	2990	3000	3010	
7.0.0									
3890									
TAA									
TAAGTGAGTC	AGTGAAAG	JAAAACAAA	ACTTCGA	TCTCTC	PTTTTTA'	ICTACCCTI	TTTAATA	ATCTCTTT	TTTTCT
 3020	1 3030	1 3040		2	1 3060	3070	3080) 3	090
AGAATCCAAA	ATAATTA	CGGTTGGAT	TACAGTT	FACTTA	IGTATCC:	ACCGTTGAZ	ATCTTAAI	CTTCCAT [,]	TGTATC
3100	3110	312	0 3	3130	3140	315	50 3	3160	3170
AAACGTCACT	GACTCTG	ITTCTGGAA	GTGTAAA(AGAGACA	GTGAATCT	TAATGTTAI	CTTCTTT	GTCTAA
3 1 .80	3:	190	3200	3210	3.	220	3230	3240	3250
<u>"""""""""""""""""""""""""""""""""""""</u>	3 3 3 5 5 5 5 5 5 5 5	- -	CTUT IN COM COM	ייייא אריי	സൗഹം				
	ppereneration (1					
3	260	3270	3280	3.	290				

A comparison of intron length and location in PEPc genes from different species Figure 4.7 In the diagram below introns are numbered 1 to 9 in Roman numerals and exons are represented by coloured boxes.



A chromosome location has not yet been assigned to this gene and unfortunately no sequence on the 5' or 3' side of the coding sequence has been submitted to allow identification of the promoter sequence for the gene or other regulatory elements. Recently (Lin et al., 1999), the sequence of an *A. thaliana* BAC clone (accession no. AC007087) from chromosome 2 was submitted to GenBank. This clone contains a PEPc gene different from the one for the PEPc cDNA clone 4.2 (81% identity) thus assigning an *A. thaliana* PEPc gene to chromosome 2 and providing 5'-UTR and 3'-UTR sequence for future analysis.

The PEPc cDNA (accession no. AF071788) also showed homology to several other A. *thaliana* sequences. Those sequences with greater than 90% identity to the PEPc cDNA were presumed to be PEPc sequences, and of these sequences, those with greater than 90% identity to one another were presumed to be part of the same gene. Using these criteria, the sequences were analysed and 7 distinct PEPc sequences were found from this search of the A. *thaliana* Genbank database (Figure 4.8). However, only the genomic (accession no. ATH131710) and cDNA (accession no. AF071788) sequences for the PEPc gene discussed in this chapter, and the other PEPc genomic clone from A. *thaliana* (accession no. AC007087) contained the characteristic start codon nucleotides ATG indicative of a full-length PEPc sequence.

4.2.4 Tissue expression study of A. thaliana PEPc clone

Rosette leaves, bracts, bolts, flowers and buds were taken from 5-week old *A. thaliana* var. Landsberg erecta plants. Root tissue was grown as described in section 2.2.2(i). After 14 days the roots were removed from the seedlings, washed with distilled water and carefully blotted dry with paper towel. All plant tissue was snap-frozen upon harvesting and stored at -80° C until required. Figure 4.9 is a labelled diagram of an *A. thaliana* var. Landsberg erecta plant showing the different tissues used for the Northern analysis of PEPc and PEPc kinase transcripts. Total RNA was isolated from 1-2 g of tissue by the method described in section 2.10.2(i) and from this poly A⁺ RNA was isolated using Promega's Poly ATract kit (section 2.10.3). These poly A⁺ RNA samples were run on a denaturing formaldehyde gel (section 2.7.5) and the RNA then immobilized onto Hybond-N membrane by overnight capillary transfer.

Primers were designed to the 3'-untranslated region (3'-UTR) of the *A. thaliana* PEPc clone (section 2.9.7 and Figure 2.1):

Figure 4.8 Alignment of *A. thaliana* PEPc cDNA with other *A. thaliana* PEPc sequences

All sequences represented have $\geq 90\%$ identity to the *A. thaliana* PEPc cDNA (accession no. AF071788) and are therefore assumed to be PEPc. Those sequences with $\geq 90\%$ identity to AF071788 and $\geq 90\%$ identity to one another are assumed to be the same PEPc gene and are coloured blue (). All other sequences are assumed to represent different *A. thaliana* PEPc genes and as such are coloured differently. The sequences are given by their accession numbers. A scale is provided to show the length of the sequences in amino acid residues.



Figure 4.9 Labelled photograph of a flowering A. thaliana var. Landsberg erecta plant



Figure 4.10 Expression of *A. thaliana* PEPc transcript in different tissues of *A. thaliana* var. Landsberg erecta

Poly A^+ RNA was isolated from different tissues harvested from *A. thaliana* var. Landsberg erecta plants. RNA samples were separated by denaturing agarose electrophoresis, immobilized on Hybond-N membrane and probed with a radiolabelled fragment of the 3'- untranslated region of *A. thaliana* PEPc cDNA. A phosphorimage of the membrane after hybridization is shown below. The samples in the lanes of the phosphorimage are as follows:

Lane 1 - roots

- Lane 2 siliques
- Lane 3 leaves
- Lane 4 bolts

Lane 5 - bracts

Lane 6 - flowers and buds

Lane 7 - A. thaliana PEPc kinase transcript



5' - ACCGGTTAAGTGAGTCAGTG - 3'

5' - ATTCACTGTCTCTGTTCTTG - 3'

These primers were used in a PCR with some *A. thaliana* PEPc template DNA to amplify an approximately 200 bp fragment of the PEPc. This fragment was then radiolabelled as described previously and used to probe the nylon membrane containing the *A. thaliana* tissue poly A^{\dagger} RNA.

Figure 4.10 shows the autoradiograph of the nylon membrane after probing with the radiolabelled 3'UTR PEPc fragment. There is a clear hybridization signal with the root poly A⁺ RNA. There is no apparent hybridization signal with any other tissue.

root poly A⁺ RNA. There is no apparent hybridization signal with any other tissue.

4.2.5 Phylogenetic analysis of A. thaliana PEPc

A phylogenetic analysis of the *A. thaliana* PEPc cDNA described in this chapter was carried out. PEPc nucleotide sequences from different species and tissues were aligned using Clustalx software and a phylogenetic tree produced using the Neighbour-joining method. The tree was viewed using Treeview software. A PEPc sequence from the gymnosperm *P. ahies* (spruce) was specified as the root of the tree based on the assumption that angiosperms diverged from gymnosperms at some point in evolution. The phylogenetic tree is shown in Figure 4.11. It is difficult to draw any conclusions from this analysis. However, the information is consistent with previous data suggesting that the *A. thaliana* PEPc discussed in this chapter is different from the PEPc contained within the *A. thaliana* chromosome 2 BAC (section 4.2.3) and with the northern blot data presented in Figure 4.10 which suggests that the *A. thaliana* PEPc cloned is a root isoform.

4.3 DISCUSSION

It is a generally accepted fact that the diverse functions of PEPc are attributable to the existence of different PEPc isoforms which have specific functional and regulatory properties. However, one property common to plant PEPcs is their control by phosphorylation. For example, similar to their C₄ and CAM photosynthetic counterparts (e.g. Nimmo et al., 1984; Brulfert et al., 1986; Budde and Chollet, 1986; Nimmo et al., 1987a; Weigend, 1994), non-photosynthetic PEPcs from C₃ and C₄ plants undergo

Figure 4.11 Phylogenetic tree of *A. thaliana* PEPc and various other PEPc isoforms

The data was obtained as described in section 4.2.5. The different isoforms are given by the name of the species and tissue that they were obtained from and their accession number. The accession number of spruce (*P. abies*) is x79090.



regulatory phosphorylation e.g. wheat (Champigny and Foyer, 1992), tobacco leaves (Wang and Chollet, 1993), soybean root nodules (Schuller and Verner, 1993), *Vicia* guard cells (Schnabl et al., 1992) and *Sorghum* roots (Pacquit et al., 1993). PEPc kinase has been partially purified from tobacco leaves (Wang and Chollet, 1993) and detected in several different C_3 tissues (e.g. Schuller and Verner, 1993; Pacquit et al., 1993) including *A. thaliana* leaf, root, bracts, bolts, flower and bud (see Chapter 5). These observations, together with the strict conservation of an N-terminal phosphorylation site in all known plant PEPc sequences, support the hypothesis that this type of reversible covalent control is a general regulatory feature of the different higher plant PEPcs.

The first full-length PEPc cDNA from the C₃ species A. thaliana, cloned and sequenced as described in this chapter, contains the phosphorylation motif SIDAO, with serine¹¹ as the phosphorylated residue (Figure 4.4). Previous alignments of deduced PEPc amino acid sequences have revealed several highly conserved residues and motifs other than the phosphorylation site. These likely contribute to the domains involved in the active site and/or regulation of the enzyme (Lepiniec et al., 1994; Toh et al., 1994). The importance of some of these conserved sites has been confirmed by site-directed mutagenesis (Duff et al., 1995; Wang et al., 1992) and chemical modification studies which have also helped to relate the primary structure of the enzyme to its function. Recently, the structure of PEPc from E.coli was determined (Kai et al., 1999) and this information will greatly advance understanding of the structure/function relationship of the enzyme. In general, the C-terminal half of the ~110kD PEPc polypeptide contains most of these presumed active-site determinants, whereas the N-terminal half appears to include the motifs that are regulatory in nature (Jiao and Chollet, 1990; Jiao et al., 1990; Terada et al., 1990; Toh et al., 1994). The histidine residue of the amino acid motif VLTAHPT (Figure 4.5, amino acid residue no. 176 - 182) is essential for the carboxylation activity of PEPc (Terada et al., 1992). The highly conserved sequence QqVMvGYSDSgKDaG (Figure 4.5, amino acid residue no. 602 - 616; the residues noted with capital letters being absolutely conserved) contains the species invariant lysine residue implicated in the active site by Jiao et al. (1990) and the recent determination of the three-dimensional structure of E. coli PEPc confirms the involvement of this lysine in the putative active site (Kai et al., 1999).

The glycine-rich motif FHGRGGtvGRGGgP (Figure 4.5 no. 645 - 658) has been proposed to be part of the substrate binding site (Terada et al., 1992). The second arginine residue (R) interacts with aspartate which is known to inhibit *E. coli* PEPc (Kai et al., 1999). In the active enzyme this region probably folds over the active site i.e. the inhibitor site is quite close to the active site. Toh et al. (1994) postulated that this region is a phosphate binding site after comparing it with the conserved region in the PEPcs of *E. coli* and *A. nidulans* which were shown to share some common features with known nucleotide- or phosphate-binding motifs. Such motifs construct flexible loops containing conserved Gly residues. In addition, the motifs contain conservative positively charged residues to interact with phosphate. The conserved region of PEPc referred to contains two-fold short repeats, Gly-Arg-Gly-Gly (Figure 4.5 no. 647 - 650 and 653 - 656). The pattern of the conservative sequence required to construct a phosphate-binding flexible loop in PEPc is therefore present and as the reaction catalysed by PEPc includes liberation of phosphate from phosphoenolpyruvate, it is quite plausible that PEPc may contain a phosphate-binding motif in its primary structure.

Of the preserved areas of unknown function, the conservation of the carboxy- terminus with regard to length as well as sequence should be noticed. The C-terminal sequence motif LTMKGIAAGMQNTG (Figure 4.5 no. 967 - 980) exists slightly modified in the carboxy-terminus of all plant PEPcs and the PEPcs of several bacteria (Relle and Wild, 1996 and references therein). It seems to be an essential component serving the stability of the enzyme (Izui et al., 1986). The binding sites of allosteric effectors have not been identified yet (Toh et al., 1994).

Further insight into the structure/function relationships of PEPc must await continued mutagenesis of these and other highly conserved domains and, most importantly, high resolution crystallographic analysis of the plant and microbial proteins.

Consistent with the enzyme's functional diversity, small multigene families of PEPc have been found (see Lepiniec et al., 1993 and references therein). *A. thaliana* has previously been suggested to have a PEPc gene family (Newman et al., 1994). A search and analysis of an *A. thaliana* sequence database with the *A. thaliana* PEPc cDNA clone described in this chapter, suggested that there are at least 7 *A. thaliana* PEPc sequences (section 4.2.3 and Figure 4.8). The Southern analysis of *A. thaliana* genomic DNA digested with a variety of restriction enzymes would also give some indication of the number of PEPc genes.

As has been reported previously for some C_4 - and C_3 -type genes, the typical TATA or CAAT box sequences, a common feature of eukaryotic genes, are not found in the 5'-flanking sequence of the *A. thaliana* PEPc sequence presented here. This is in contrast to the data of Lepiniec et al., (1993) who reported a typical CAAT box and TATA box in *Sorghum* PEPc. Neither does the *A. thaliana* PEPc sequence contain evidence of the multiple polyadenylation sites reported in the 3'-UTR of some other PEPc sequences (Crétin et al., 1991; Matsuoka and Minami, 1989; Yanagisawa et al., 1988).

Having obtained the first full-length *A. thaliana* PEPc cDNA, it was of immediate interest to determine which isoform of PEPc had been cloned. Northern analysis of poly A^+ RNA from different *A. thaliana* plant tissues led to the hybridization of the sequence specific probe to root tissue only (Figure 4.10). This expression pattern is similar to the root-specific isoform reported in maize by Kawamura et al. (1990) and is distinct from the maize anaplerotic isoform which was constitutively expressed in green and etiolated leaves and roots (Kawamura et al., 1992). The result of phylogenetic analysis of the *A. thaliana* PEPc (Figure 4.11) is consistent with the Northern data suggesting that the PEPc cloned is a root isoform. However, it is not possible to draw any definitive conclusions regarding the origin of the clone from this analysis. Comparison of the hydropathy profile of the *A. thaliana* PEPc with those of known PEPc isoforms e.g. the root and anaplerotic maize isoforms, would highlight local structural differences or similarites in regions which may be associated with differences in catalytic and regulatory properties of the isozymes, and thus help confirm the identity of the *A. thaliana* PEPc clone.

The *A. thaliana* PEPc clone pKP42 was one of three clones of >3 kb isolated in the screening of the λ PRL2 cDNA library (Figure 4.2). It would be worthwhile to check if the three clones are different PEPc isoforms as it is valuable to have more than one member of a gene family from a single plant species (e.g. Lepiniec et al., 1993) for comparisons of isoform function and regulation. *In situ* hybridization would clarify the tissue expression of each gene.

Now that clones for both a PEPc gene and PEPc kinase from *A. thaliana* are available, the over-expression of the genes is a desirable target. The purified proteins would not

only facilitate the investigation of PEPc regulation at the protein level, for example by providing a good control for kinase assays, but for the first time would enable studies of the interaction of the kinase with PEPc to be carried out. Mr J. Jardine has been trying different vectors to express *Kalanchoë fedtschenkoi* PEPc with the intention of then using any successful system for the expression of the *A. thaliana* PEPc and PEPc kinase. At present a suitable vector has still not been found.

The recent Genbank submission of an *A. thaliana* BAC containing a PEPc gene (accession no. AC007087) has revealed the presence of a PEPc gene on chromosome 2 of *A. thaliana*. However, the chromosomal location of the gene for the PEPc cDNA cloned during this research is not yet known but could be determined by restriction fragment length polymorphism (RFLP) analysis using the specific probe for the *A. thaliana* PEPc cDNA and other PEPc probes whose chromosomal location is known. In all C_{3} - and C_{4} -monocots and dicots studied so far, the various PEPc genes are not located together (Lepiniec et al., 1994). Kawamura et al. (1992) found that the genes for the C_{4} and C_{3} isoenzymes of PEPc in maize are located apart on different chromosomes. Chao et al. (1989), located PEPc to chromosome number 7 in the C_{3} species wheat, although the RFLP analysis also gave hybridization of the *Sorghum* probe used to a number of fragments from the group 3 chromosomes, suggesting that there are also several other related sequences in wheat.

To summarise therefore, sequence analysis and comparison of the first full-length PEPc cDNA from *A. thaliana* with other PEPc sequences is consistent with the importance assigned to certain conserved sites in the regulation and activity of PEPc. The tissue expression study (Figure 4.10) and phylogenetic analysis (Figure 4.11) suggest that the isoform cloned is root-specific and a search of an *A. thaliana* sequence database indicates that the clone is only one of at least seven PEPc genes. The isolation of clones for these other genes will provide great insight into the different expression patterns of the isoforms of PEPc in the C_3 species *A. thaliana* and their regulation.

Chapter Five

CHARACTERIZATION OF PHOSPHOENOLPYRUVATE CARBOXYLASE AND ITS KINASE FROM THE C₃ SPECIES *ARABIDOPSIS THALIANA*

5.1 INTRODUCTION

The work described in Chapter 1 highlights the fact that the regulation of PEPc and its kinase in C_3 species appears to be significantly different from C_4 and CAM species. It was therefore of interest to investigate the nature and extent of PEPc regulation in C_3 species and more specifically the involvement of PEPc kinase in the regulation.

The purification and characterization of PEPc from photomixotrophically cultured green tobacco cells (Sato et al., 1988) showed the purified enzyme to display typical C_{3^-} , as opposed to C_{4^-} , plant type behaviour with regards to substrate affinity and substrate saturation kinetics. Furthermore, immunochemical studies suggested that the cultured cells possibly have a very similar enzyme to intact tobacco leaves. Therefore, for the purpose of this present research, a photomixotrophic culture of green *A. thaliana* cells was chosen as a source of the C_3 PEPc isoform and its kinase(s). *A. thaliana* is a system in which biochemical parameters can be defined alongside gene cloning and genetic analyses and the culture provides a simple model to study because of the undifferentiated state of the cells and the ease with which metabolic conditions can be manipulated Experiments were also carried out using *A. thaliana* plant tissue.

During the course of this work, a full length cDNA for PEPc kinase from *A. thaliana* became available. This was identified following the cloning of PEPc kinase from a *K. fedtschenkoi* cDNA library. The first full length cDNA for PEPc from *A. thaliana* was also cloned and sequenced as described in Chapter 4. These discoveries facilitated the study of the expression of both the PEPc and PEPc kinase genes, and the steady-state transcript levels of both PEPc and its kinase were examined in the culture and different tissues of *A. thaliana*. In consideration of the supposed involvement of C_3 PEPc activity with nitrogen metabolism, the effects of nitrate and sucrose concentrations on the transcript level of PEPc and PEPc kinase in the culture were also investigated.

5.2 RESULTS

5.2.1 Development of an extraction protocol for PEPc activity from *A. thaliana* cell culture

The *A. thaliana* cell culture was established and maintained as described (section 2.2.1). As the culture had not previously been assayed for PEPc activity, a method had to be developed for the extraction of the enzyme activity.

2 ml samples of 3 day old cells were spun in a microfuge and the cell pellet resuspended in 1 ml of sonication buffer (section 2.4). The cell suspension was then sonicated on ice for 30 s at an amplitude of 5 microns using a Soniprep sonicator. When a little of the sonicated sample was viewed under a light microscope it was seen that the majority of the *A. thaliana* cells had been broken open by the sonication. The sonicated cells were then microfuged and the superpatant removed by micropipetting to a clean 1.5 ml microfuge tube. The supernatant was assayed for PEPc, but very little activity, approximately 0.02 units/ml, was detected. By comparison, the same samples had approximately 0.5 units/ml of MDH activity (section 2.6.3). These values suggested that the culture was metabolically active but exhibited low levels of PEPc activity under normal growth conditions. It was therefore necessary to develop the extraction protocol to obtain samples in which the PEPc was more concentrated.

After sonication, 750 μ l of sample was ultracentrifuged for 20 mins at 30,000 rpm and 4°C to remove as much cell debris as possible. The resulting supernatant was brought to 75% saturation with ground ammonium sulphate and left to precipitate on ice for 30 mins. The sample was then spun in a microfuge for 2 mins at 13,000 rpm and 4°C and the pellet resuspended in 50 μ l sonication buffer.

When the sample was assayed for PEPc activity, an activity of 0.3 units/ml was obtained, that is an approximately 15-fold concentration. It was now also possible to measure the malate sensitivity of the culture sample. The sensitivity of PEPc to L-malate is measured as the apparent K_i of the enzyme i.e. the inhibitor concentration which reduces the velocity of the enzyme by 50%. Figure 5.1 shows that an apparent K_i of 0.4 mM L-malate was obtained for the culture PEPc.

In subsequent experiments the sonicate was microfuged and the ammonium sulphate added without ultracentrifugation of the supernatant. The extraction protocol for PEPc activity from *A. thaliana* cell culture is summarised in section 2.5.1.

Figure 5.1 Determination of the malate sensitivity (apparent K_i) of PEPc in Λ . *thaliana* cell culture

Cell culture samples were prepared and assayed for PEPc activity in the presence of different concentrations of L-malate (see section 2.6.1 for assay conditions).

The graph below was used to determine the malate sensitivity or apparent K_i of the PEPc i.e. the L-malate concentration at 50% maximum PEPc activity.



L-malate concentration (mM)

The assay mixture used to measure PEPc activity in the cell culture was adapted from Van Quy et al. (1991) (section 2.4). The assay mixture was at pH 7.6 and contained 0.3 mM PEP. An extract of cell culture was made as described above and then assayed for PEPc activity at pH 7.0, 7.2, 7.4 and 7.6. The activity of PEPc detected was highest at pH 7.6 (data not shown). The malate sensitivity of PEPc from cell culture extract was then measured at pH 7.6 and 0.3 mM PEP and and at pH 8.0 and 3.0 mM PEP. The results in Table 5.1 show that higher PEPc activities were detected under "optimal" assay conditions (higher pH and PEP concentration) but the enzyme was not sensitive to malate. By comparison, under the standard assay conditions of pH 7.6 and 0.3 mM PEP, PEPc activity was slightly lower but the enzyme displayed malate sensitivity, having an apparent K_i between 0.5 and 1.0 mM L-malate.

 Table 5.1
 The activity and malate sensitivity of PEPc from A. thaliana cell culture assayed under different conditions

	PEPc activity (units/ml)	PEPc activity (units/ml)
	pH 8.0 and 3.0 mM PEP	pH 7.6 and 0.3 mM PEP
0 mM L-malate	0.23	0.17
0.1 mM L-malate	0.29	0.16
0.3 mM L-malate	0.32	0.18
0.5 mM L-malate	0.28	0.10
1.0 mM L-malate	0.30	0.07

5.2.2 Characterization of *A. thaliana* cell culture growth, PEPc activity and the malate sensitivity of PEPc

The *A. thaliana* cell culture was maintained by subbing 20 ml of culture into 180 ml of fresh medium every 7 days. The growth of the culture throughout this 7 day period was recorded as the fresh weight (g/L). 2 ml of freshly subbed cells were aliquoted into twenty-one sterile 50 ml glass conical flasks each containing 18 ml of culture medium

Figure 5.2 Time course of cell growth in an A. thaliana cell culture

Cell culture samples were collected every day throughout the growth phase of the culture, spun down and the fresh weight measured.

The graph below shows the results obtained.



time (hours)

and shaken as for the 200 ml volumes of culture. Each day of the time course, duplicate flasks were taken and the contents filtered using Buchner apparatus, then weighed to measure fresh weight. Figure 5.2 shows that the growth of the cell culture was exponential during this period. Culture maintained longer than 7 days entered a stationary phase by day 10 (data not shown).

The specific activity of PEPc and its sensitivity to malate were also followed throughout the 7-day growth phase of the culture. One 50 ml flask containing 20 ml culture was taken each day and an extract prepared (section 2.5.1). 5 μ l samples of the extract were used to measure the activity and malate sensitivity of PEPc (section 2.6.1). The protein concentration of samples was determined by the dye-binding assay of Bradford (section 2.12) to enable PEPc specific activity to be calculated. No clear trends were apparent for either kinetic characteristic throughout the time course (Figure 5.3). The average values obtained were a specific activity of 0.13 units/mg protein and an apparent K_i of 0.54 mM L-malate when measured under sub-optimal conditions of pH 7.6 and 0.3 mM PEP. This value for specific activity is consistent with those obtained for PEPc from other C₃ tissues e.g. 0.16 units/mg protein from tobacco cell culture (Sato et al., 1988) and 0.04 -0.13 units/mg protein from a range of C₃ species (Gupta et al., 1994).

A desalted, ammonium sulphate precipitated extract of rosette leaves from flowering *A*. *thaliana* plants var. Columbia (prepared as described in section 2.5.2) was also assayed for PEPc activity and malate sensitivity. The apparent K_i for malate (0.5 mM) obtained from the leaf extract was comparable to that obtained from the *A*. *thaliana* cell culture.

5.2.3 Manipulation of the malate sensitivity of PEPc in *A. thaliana*: effects of the protein phosphatase inhibitor cantharidin and light

The malate sensitivity of PEPc is a qualitative measure of the *in vivo* phosphorylation state of PEPc. It is therefore possible to assess whether particular treatments affect the phosphorylation state of PEPc by monitoring the apparent K_i for malate.

To investigate whether the cell culture contained the machinery to phosphorylate PEPc, the apparent K_i of PEPc for L-malate was assayed after incubation of the cells with cantharidin which is an inhibitor of protein phosphatase type 1 and 2A activity (MacKintosh and MacKintosh, 1994; Smith and Walker, 1996). It should therefore

Figure 5.3 The specific activity and malate sensitivity of PEPc in *A. thaliana* cell culture

Cell culture samples were collected every day throughout the growth phase of the culture and assayed for the specific activity and malate sensitivity of PEPc.

The graph opposite shows the results obtained: specific activity (\blacklozenge); apparent K_i(\blacksquare).

The specific activity and malate sensitivity of PEPc in A. thaliana cell culture





time (hours)



increase the phosphorylation state of PEPc and decrease its sensitivity to feedback inhibition by L-malate.

10 ml volumes of 3 day old culture were incubated for 6 hours in 25 ml tissue culture bottles with or without 100 μ M cantharidin and shaken gently at 75 rpm. The PEPc activity from the different incubation conditions was assayed in the presence and absence of 0.5 mM L-malate. The means of the results agreed to within 15% and, as shown in Figure 5.4, PEPc from cells incubated with cantharidin showed significantly less sensitivity to malate than control cells. The cantharidin was dissolved in ethanol, so as a further control some cells had been incubated in ethanol but this did not appear to affect the malate sensitivity of PEPc.

Since a pharmacological agent had been effective in manipulating the malate sensitivity of the culture PEPc, the potential of light, an environmental factor, to alter the malate sensitivity of the culture PEPc was also investigated. 10 ml volumes of 3 day old cell culture were aliquoted into 25 ml tissue culture bottles and shaken for 24 hours at 75 rpm under normal temperature and light intensity growth conditions for culture. Some of the tissue culture bottles were wrapped in aluminium foil so that the cells were shaken in the dark. After 24 hours, extracts of the light- and dark- incubated culture samples were prepared and assayed for the malate sensitivity of PEPc. The light and dark samples had apparent K_i values of approximately 0.5 mM and 0.7 mM L-malate respectively, that is, there appeared to be no significant difference in the malate sensitivity of the PEPc from the two treatments (Figure 5.5).

Detached leaves of *A. thaliana* var. Columbia were also used to study the effect of light on the malate sensitivity of C₃ PEPc. Flowering plants grown under a 16 hour photoperiod (50 μ moles/m²/s white light) at 20°C were transferred to a higher intensity white light (500 μ moles/m²/s) and 27°C growth regime at the beginning of a 12 hour photoperiod. Leaves were detached 8 hours into the dark/18°C period (dark samples) and some were put into water and incubated for a further 4 hours darkness followed by 3 hours in the light (light samples). Desalted, ammonium sulphate precipitated extracts of the leaves were then prepared and assayed for the malate sensitivity of PEPc. No appreciable difference in malate sensitivity was observed between the light and dark samples (Figure 5.5). Therefore, consistent with the results using the cell culture, no evidence was found to suggest light/dark changes in the malate sensitivity of *A. thaliana*

Figure 5.4 The effect of the phosphatase inhibitor cantharidin on the malate sensitivity of PEPc

Cell culture samples were incubated in either ethanol or 100 μ M cantharidin. The samples were then assayed for sensitivity to 0.5 mM L-malate expressed as a percentage of PEPc activity from control samples from the same incubation conditions assayed with and without L-malate.

The graph below shows the results obtained.



% PEPc activity at 0.5 mM L-malate

Figure 5.5 The effect of light and dark on the PEPc malate sensitivity in *A. thaliana* cell culture and leaves

3-day cell culture samples were incubated in the light (20 μ moles/m²/s, 20°C) or dark for 24 hours and then the malate sensitivity of PEPc was assayed. Leaves were detached from flowering *A. thaliana* plants 8 hours into a 12 hour dark period (18°C). Some detached leaves were then assayed for the malate sensitivity of dark-treated samples. Other leaves were incubated in water for the remainder of the dark period and were then collected and assayed for malate sensitivity 3 hours into a subsequent 12 hour light period (500 μ moles/m²/s, 27°C).

The graph below shows the results obtained: light-treated culture (\clubsuit) ; dark-treated culture (\clubsuit) ; light-treated leaves (\clubsuit) ; dark-treated leaves (\circledast) .



L-malate concentration (mM)

PEPc from culture or leaves. The increases in light intensity and temperature were used because other plant species demonstrating a light enhancement of PEPc activity had been grown under similar conditions (e.g. Jiao and Chollet, 1988). However, it is not impossible that the dramatic change in light intensity and temperature stressed the leaves of the *A. thaliana* plants thus preventing the effect of the light/dark treatment alone to be studied.

5.2.4 PEPc and PEPc kinase transcript levels in *A. thaliana* cell culture and plant tissue

The failure to detect clear changes in the malate sensitivity of PEPc raised questions about the occurrence of phosphorylation of the protein in *A. thaliana*, at least under the experimental conditions employed here. The isolation of the *A. thaliana* PEPc kinase cDNA (see 5.1) meant that it was possible to test whether the PEPc kinase gene was expressed. Northern analysis of total RNA from the culture and leaf tissue failed to detect the kinase transcript (data not shown).

The more sensitive approach of RT-PCR (see section 2.10.5) allowed the detection of the PEPc kinase transcript in the culture and 3 week-old A. thaliana var, Landsberg erecta plants that had been grown under continuous high white light (500 μ moles/m²/s). Figure 5.6 is a photograph of some RT-PCR products separated on a 1% agarose gel (section 2.7.4) and visualised by ethidium bromide staining. Total RNA was isolated from culture and plant tissue (section 2.10.2(i)) and oligodT (polythymine oligonucleotide) was used to prime the reverse transcription reaction catalysed by Promega's AMV Reverse Transcriptase (RT reactions 2-5). Some transcript was also made from the A. thaliana PEPc kinase cDNA after excision from the plasmid pZL1 by digestion with Hind III and Pst I (Figure 5.7). This transcript was then reverse transcribed using a primer to the antisense strand of the PEPc kinase cDNA (RT1). PCR (section 2.9.5) was performed on the different reverse transcription reaction products to amplify PEPc kinase, PEPc and tubulin from the different samples using primers designed to A. thaliana PEPc kinase cDNA, a conserved region of PEPc (Honda et al., 1996) and a-tubulin from Chlamydomonas reinhardtii (Silflow et al., 1985), respectively (section 2.9.7). Plasmid and water controls were included where appropriate. Conserved PEPc kinase primers were used in reverse transcription reaction 1 and control PCR (Figure

Figure 5.6 RT-PCR of PEPc kinase, PEPc and α -tubulin steady-state transcript from *A. thaliana* culture

Total RNA was isolated from *A. thaliana* cell culture and 3-week old *A. thaliana* plants var. Landsberg erecta which had been grown under a high white light (~500 μ moles/m²/s, 27°C) 12 hour photoperiod. The following reverse transcription (RT) reactions were performed as described in section 2.10.5:

RT1. 2 μ l A. thaliana kinase transcript + 3.6 μ l antisense conserved PEPc kinase primer

RT2. 1 μ l RNA from 3% sucrose stationary culture cells + 3.6 μ l oligo dT primer

RT3. 0.9 µl RNA from 3% sucrose exponential culture cells + 3.6 µl oligo dT primer

RT4. 0.75 µl RNA from 1% sucrose exponential culture cells + 3.6 µl oligo dT primer

RT5. 0.7 µl RNA from A. thaliana plants + 3.6 µl oligo dT primer

5 μ l of each reverse transcription reaction product was then used as the template in a PCR. The products of each PCR were run out on a 1% agarose gel with 1 kb DNA ladder (outside lanes) and are shown in the photographs opposite. The reactions in the lanes of each photograph are as follows:

(A)	Lane 1 - 1 kb DNA ladder	_	
	Lane 2 - RT 1		
	Lane 3 - RT 2		A. thaliana kinase-specific
	Lane 4 - RT 3		primers
	Lane 5 - RT 4		
	Lane 6 - RT 5		
	Lane 7 - linearized A. thaliana PEPc kinase cl	DNA 🗆	
	Lane 8 - RT 1		
	Lane 9 - RT 2		
	Lane 10 - RT 3	PEP	c primers
	Lane 11 - RT 4		
	Lane 12 - RT 5		
	Lane 13 - linearized A. thaliana PEPc cDNA -		
	Lane 14 - 1 kb DNA ladder		
(B)	Lane 1 - 1 kb DNA ladder		
	Lane 2 - RT 1 + conserved PEPc kinase prime	ers	
	Lane 3 - water + conserved PEPc kinase prim	ers	
	Lane 4 - water + A. thaliana specific PEPc kin	hase prime	ers
	Lane 5 - water + PEPc primers		
	Lane 6 - RT 1		
	Lane 7 - RT 2		
	Lane 8 - RT 3 α-tubulin g	orimers	
	Lane 9-RT4		
	Lane 10 - RT 5		
	Lane 11 - water		
	Lane 12 - 1 kb DNA ladder		


1 2 3 4 5 6 7 8 9 10 11 12 13 14





1 2 3 4 5 6 7 8 9 10 11 12





Diagram of the A. thaliana PEPc kinase EST in plasmid pZL1 Figure 5.7

5.6) because they had previously successfully primed the reverse transcription of PEPc kinase from A. thaliana RNA (data not shown) and the A. thaliana PEPc kinase-specific primers had not yet been used in any reaction. A putative PEPc kinase was amplified from the reverse transcription products of RNA from exponential cells grown in 1%- and 3%-sucrose grown A. thaliana culture and the A. thaliana leaf tissue (Figure 5.6A lanes 4-6). PCR with an annealing temperature of 55°C using the tubulin specific primers unexpectedly resulted in more than one amplification product from culture and leaf RNA RT products (Figure 5.6B lanes 7-10). Surprisingly, when the annealing temperature of the reaction was lowered to 46°C (Jackson et al., 1995), the higher molecular weight amplification product was no longer apparent (data not shown). The PEPc kinase and PEPc plasmid controls gave the expected bands (Figure 5.6A lanes 7 and 13) but the water control for the PEPc primers (Figure 5.6B lane 5) showed contamination making it impossible to interpret anything from the bands present in the products of PCR using the PEPc primers from RT reactions (Figure 5.6A lanes 8-12). Such problems were not consistently encountered but overall difficulties experienced obtaining reproducible data meant that it was not possible to quantify transcript levels of PEPc or PEPc kinase using RT-PCR.

Northern analysis of poly A⁻ RNA (sections 2.8 and 2.10) was therefore employed, initially to detect the transcripts of both PEPc and PEPc kinase in the culture, and then to investigate the level of the transcripts in *A. thaliana* plant tissue. The plant tissue was harvested as described in Chapter 4 and section 2.2.2(i).

Total RNA was isolated from 1-2 g of tissue by the method described in section 2.10.2(i) and from this poly A^{*} RNA was isolated using Promega's Poly ATract kit (section 2.10.3). Poly A^{*} RNA was also isolated by the same methods from approximately 950 μ g of total RNA from 4-day old *A. thaliana* cell culture grown in 3% sucrose (a gift from O. Oswald).

Poly A⁺ RNA samples were separated by denaturing agarose gel electrophoresis and the samples then transferred to Hybond-N membranes (sections 2.7.5 and 2.8.1).

The *A. thaliana* PEPc kinase cDNA (Hartwell et al., submitted) was cut out of the plasmid pZL1 with *Hind* III and *Pst* I to give a fragment of PEPc kinase cDNA to be radiolabelled for use as a probe. Figure 5.7 is a diagram of the plasmid pZL1 with the PEPc kinase cDNA insert. Primers to the 3'-untranslated region (3'-UTR) of the *A*.

thaliana PEPc clone (Chapter 4 and section 2.9.7)) were used in a PCR to amplify an approximately 200 bp fragment from *A. thaliana* PEPc cDNA (section 4.2.4.). An approximately 400 bp fragment of coding sequence conserved between different PEPc genes was also isolated from *A. thaliana* PEPc cDNA (section 2.9.7).

2

These three cDNA fragments and a fragment of an α -tubulin gene from *Chlamydomonas reinhardtii* (Silflow et al., 1985) (a gift from H. Wade) were radiolabelled (section 2.8.2 and 2.8.3) and used to probe the nylon membranes (section 2.8.4) containing poly A⁺ RNA from *A. thaliana* tissue and culture. The specific activity of each probe was 1.14×10^9 dpm/µg or greater (section 2.8.3).

Phosphorimages (section 2.7.3) of the membranes containing the tissue and culture poly A^+ RNA after hybridization with the four different radiolabelled probes are shown in Figures 5.8 and 5.9 respectively. A hybridization signal was seen in the culture poly A^- RNA with all four probes (Figure 5.9). A clear hybridization signal was apparent in all the plant tissues with the PEPc kinase probe (Figure 5.8). As mentioned already, the PEPc kinase cDNA was excised from the plasmid pZL1 by digestion with *Hind* III and *Pst* I, restriction enzymes with sites in the multiple cloning site of plasmid pZL1 (section 2.10.6). Since there is also a *Hind* III site in the 3'UTR of the kinase cDNA, the kinase transcript run off from this is smaller (<300 bp smaller) than the *in vivo* transcript. The hybridization signals of kinase with plant tissue in Figure 5.8C would confirm this.

Hybridization with the PEPc 3'UTR probe was seen in root tissue (Figure 5.9A lane 1). The conserved PEPc sequence probe only hybridized to poly A^+ RNA from the root and flower/bud tissue and culture (Figure 5.8B lanes 1 and 6; Figure 5.9B lane 2). The tubulin probe (Figure 5.8D) was used as a constitutive control to determine the relative loading of poly A^+ RNA for each tissue. The intensity of the tubulin hybridization signal with each tissue was then used to standardize the level of the other three transcripts in each tissue.

5.2.5 The effect of sucrose and nitrate on the transcript levels of PEPc and PEPc kinase in *A. thaliana* cell culture.

A. thaliana cell culture was grown in 3% or 1% sucrose. After 3 days, 555 ml of each culture was transferred to sterile centrifuge bottles and spun at 5000 rpm in a Beckman centrifuge using a JA-14 rotor for 10 mins at 20° C. The pellets of both cultures were

Figure 5.8 Northern blot analysis of PEPc and PEPc kinase steady-state transcript levels in *A. thaliana* tissue

Poly A^* RNA was isolated from different tissues harvested from *A. thaliana* var. Landsberg crecta plants. RNA samples were separated by denaturing agarose electrophoresis, immobilized on Hybond-N membrane and probed with radiolabelled fragments as indicated below. Phosphorimages of the membrane after each hybridization are shown opposite. The intensity of the α -tubulin signal for each lane was used to standardize all other signals for RNA loading. Where relevant, the corrected intensities of bands relative to the most intense band for each probe is shown below the phosphorimages. The samples in the lanes of each phosphorimage are as follows:

- Lane 1 roots
- Lane 2 siliques
- Lane 3 leaves
- Lane 4 bolts
- Lane 5 bracts
- Lane 6 flowers and buds
- Lane 7 A. thaliana PEPe kinase transcript

(A) Phosphorimage of membrane after hybridization to a radiolabelled fragment of the 3'-untranslated region of A. *thaliana* PEPc.

(B) Phosphorimage of membrane after hybridization to a radiolabelled fragment of an *A. thaliana* PEPc conserved between different PEPc genes.

(C) Phosphorimage of membrane after hybridization to a radiolabelled fragment of A. *thaliana* PEPc kinase.

(D) Phosphorimage of membrane after hybridization to a radiolabelled fragment of α -tubulin.



Figure 5.9 Northern blot analysis of PEPc and PEPc kinase steady-state transcript levels in *A. thaliana* culture

Poly A^+ RNA was isolated from *A. thaliana* cell culture. The poly A^+ RNA and supernatant (containing transfer RNA and ribosomal RNA) from the Poly ATract kit column were separated by denaturing agarose electrophoresis, immobilized on Hybond-N membrane and probed with radiolabelled fragments as indicated below. Phosphorimages of the membrane after each hybridization are shown opposite. Transcript sizes are as shown in Figure 5.8. The samples in the lanes of each phosphorimage are as follows:

Lane 1 - A. thaliana PEPc kinase transcript (1:1000 dilution)

- Lane 2 poly A⁺ RNA
- Lane 3 Poly ATract column supernatant

(A) Phosphorimage of membrane after hybridization to a radiolabelled fragment of the 3'-untranslated region of *A. thaliana* PEPc.

(B) Phosphorimage of membrane after hybridization to a radiolabelled fragment of an *A. thaliana* PEPc conserved between different PEPc genes.

(C) Phosphorimage of membrane after hybridization to a radiolabelled fragment of A. *thaliana* PEPc kinase.

(D) Phosphorimage of membrane after hybridization to a radiolabelled fragment of α -tubulin.



then washed three times with MS salts made up without any nitrate, centrifuging to remove the supernatant after each wash. After the final centrifugation step, the pellets were resuspended in MS salts (section 2.4) with either 3% or 1% sucrose. 60 ml volumes of the 3% sucrose culture were then transferred to sterile 200 ml glass conical flasks containing filter sterilised NH4NO3/KNO3 solution to make the medium to a final nitrate concentration of 0 mM, 0.6 mM, 6 mM or 60 mM. 100 ml volumes of the 1% sucrose culture were transferred to sterile 250 ml glass conical flasks containing sterile nitrate solution to the specified concentration. One flask from both the 3% and 1% sucrose cultures containing 60 mM nitrate was wrapped in tin foil to control for the effect of light, and then all the flasks were shaken as normal for a further 48 hours. Prior to washing the 3% sucrose culture, 60 ml had been removed as a control for washing, and, immediately following washing, 60 ml of 3% sucrose culture containing 60 mM nitrate was removed as a control for growing the culture in the smaller flasks. Both controls were immediately filtered using Buchner apparatus and Whatman 3MM filter paper and the cells snap-frozen in liquid nitrogen, then stored at -80°C until required. After incubation, all the other culture samples were harvested and stored in the same way, Total RNA was isolated from each sample (by Mr. J. Jardine) and Promega's polyAtract system was then used to obtain poly A⁺ RNA from each of the samples. The poly A⁺ RNA was separated using denaturing agarose gel electrophoresis and the RNA immobilised on nylon membrane as described previously. The nylon membrane was probed with the radiolabelled fragments used in section 5.2.4. The phosphorimages of the washed membrane after hybridization to each probe are shown in Figure 5.10, Hybridization of each probe to the membrane was evident and the hybridization with tubulin (Figure 5.10D) was used to standardize the level of the other three transcripts. The hybridization signals in Figure 5.10A are rather faint but this is most likely due to the nylon membrane being stripped (section 2.8.5) several times before hybridization with the 3'UTR probe. For ease of interpretation, the standardised intensity of hybridization for each sample is presented in graphical form for both the 3% sucrose and 1% sucrose cultures in Figure 5.11.

The only consistent and easily interpretable observation from these results is that the level of transcripts detected by the PEPc kinase probe and both PEPc probes was higher when the cells were grown in 1% sucrose compared with 3% sucrose. Light-treated cells

Figure 5.10 Northern blot analysis of PEPc and PEPc kinase steady-state transcript levels in *A. thaliana* culture in response to altered sucrose and nitrate levels

3-day old *A. thaliana* cell culture was washed and samples then resuspended in different media and shaken for a further 24 hours as described in section 5.2.5. Poly A^* RNA was isolated from the *A. thaliana* cell culture after treatment. The poly A^+ RNA from the Poly A'Tract kit column was separated by denaturing agarose electrophoresis, immobilized on Hybond-N membrane and probed with radiolabelled fragments as indicated below. Phosphorimages of the membrane after each hybridization are shown opposite. Transcript sizes are as shown in Figure 5.8. The percentage intensities of bands relative to the most intense band for each probe is shown below the phosphorimages. The intensity of the α -tubulin signal for each lane was used to standardize all other signals for RNA loading. The samples in the lanes of each phosphorimage are as follows:

- Lane 1 unwashed
- Lane 2 washed
- Lane 3 3% sucrose + 60mM NO₃
- Lane 4 3% sucrose + 60mM NO₃ + dark
- Lane 5 3% sucrose NO₃⁻
- Lane 6 3% sucrose + 6mM NO₃⁻
- Lane 7 3% sucrose + 0.6mM NO₃⁻
- Lane 8 1% sucrose + 60mM NO₃
- Lane 9 1% sucrose + 60mM NO_3 + dark
- Lane 10 1% sucrose NO3
- Lane 11 1% sucrose + 6mM NO₃⁻
- Lane 12 1% sucrose + 0.6mM NO₃⁻
- Lanc 13 A. thaliana PEPc kinase transcript

(A) Phosphorimage of membrane after hybridization to a radiolabelled fragment of the 3'-untranslated region of *A. thaliana* PEPc.

(B) Phosphorimage of membrane after hybridization to a radiolabelled fragment of an *A. thaliana* PEPc conserved between different PEPc genes.

(C) Phosphorimage of membrane after hybridization to a radiolabelled fragment of A. *thaliana* PEPc kinase.

(D) Phosphorimage of membrane after hybridization to a radiolabelled fragment of α -tubulin.



Figure 5.11 PEPc and PEPc kinase steady-state transcript levels in *A. thaliana* culture in response to altered sucrose and nitrate levels

The data shown in Figure 5.10 as northern blot analysis is displayed opposite in graphical form. Unwashed cells showed the highest level of each transcript, including the control α -tubulin, and so were used as a control for relative intensity of hybridization (100%). The different transcripts are represented thus: *A. thaliana* PEPc (\blacklozenge); conserved PEPc (\clubsuit); *A. thaliana* PEPc kinase (\blacktriangle).

(A) The effect of nitrate on the transcript level of PEPc and PEPc kinase in 3% sucrose-grown *A. thaliana* culture.

(B) The effect of nitrate on the transcript level of PEPc and PEPc kinase in 1% sucrose-grown A. thaliana culture.





NO3⁻ concentration (mM)

The effect of nitrate on the transcript level of PEPc and PEPc kinase in 1% sucrose-grown A. thaliana culture



relative intensity of hybridization (%)

NO3⁻ concentration (mM)

A

grown in 1% sucrose had a higher level of the *A. thaliana* PEPc transcript than darktreated cells. In contrast, the PEPc kinase transcript level in 1% sucrose-grown cells was higher with the dark treatment.

The introduction of nitrate at 0.6 mM concentration to the culture medium, supplemented with either 1% or 3% sucrose, increased the transcript level of PEPc and PEPc kinase (Figure 5.10A, B and C lanes 5, 7, 10 and 12). In 1% sucrose-grown cells, all three transcript levels then feil below the level in the absence of nitrate as the concentration was increased to 6 mM nitrate. The levels then increased again as the nitrate concentration was increased to 60 mM. Likewise, in 3% sucrose-grown cells, the *A. thaliana*-specific PEPc transcript level initially rose, then declined only to rise again as the nitrate concentration increased. The conserved PEPc transcript also initially rose with the introduction of 0.6 mM nitrate to the culture medium but then steadily declined as the nitrate concentration decreased. The PEPc kinase transcript level increased between 0 and 6 mM nitrate but no further as the nitrate concentration was increased to 60 mM.

5.3 DISCUSSION

The first objective of the work described in this chapter was to investigate the use of the *A. thaliana* cell culture as as experimental system in which the effects of metabolite availability on PEPc and PEPc kinase could be tested. To some extent this was successful but progress was hindered by fungal contamination - a problem in other laboratories throughout the project - hence not as much progress was made in this research as had been hoped. Nevertheless some positive results did emerge. In general, the work showed that the cell culture is indeed a useful system for the study of the regulation of PEPc and PEPc kinase from a C_3 species.

The response of PEPc from various sources to malate and glucose 6-phosphate is known to be affected by assay pH (Israel and Jackson, 1982; Podesta and Andreo, 1989; Schuller et al., 1990; Wedding et al., 1990). Malate is a stronger inhibitor (Schuller et al., 1990; Wedding et al., 1990) and glucose 6-phosphate a stronger activator (Schuller et al., 1990) at pH 7.0 than at pH 8.0. The sensitivity of PEPc from the *A. thaliana* culture to the effector malate was also found to be greater at an assay pH of 7.6 than 8.0 (Table 5.1). These data indicate that pH fluctuations may be significant for the regulation of

171

N . 1997-1 PEPc by malate, consistent with the proposed pH stat function of soybean nodule PEPc (Israel and Jackson, 1982).

No apparent trend was evident in either the malate sensitivity or the specific activity of the culture PEPc over the 7 days of the culture growth phase (Figure 5.3). This contrasts with the results of Sato et al. (1988) who showed that the PEPc activity in a photomixotrophic culture of tobacco cells was highest during or at the onset of exponential growth. However, the comparability of the malate sensitivity of PEPc in A. *thaliana* culture with that in A. *thaliana* leaf tissue (section 5.2.2) suggests that the culture is a convenient experimental system to use for the purposes of this research.

After the characterization of the specific activity and malate sensitivity of PEPc in the culture, the potential to manipulate the malate sensitivity of PEPc from both culture and leaf tissue was investigated as a tool by which to elucidate some of the factors regulating C_3 PEPc. The result of the cantharidin experiment (section 5.2.3) is consistent with the view that reversible phosphorylation is the means by which the C₃ isoform of PEPc, like the C4 and CAM isoforms, modulates its sensitivity to allosteric effectors. In the presence of the phosphatase inhibitor cantharidin, the PEPc will remain in its malate insensitive (high apparent K_i), phosphorylated state and so will demonstrate less inhibition by malate. However, there is no evidence that light can cause a change in the K_i of A. thaliana PEPc for malate (section 5.2.3). Fluorescence studies on the culture have shown that the cells' photosystems I and II have photosynthetic capacity (Dr. PJ Dominy, personal communication). It is therefore possible that the high photosynthate (3% sucrose) in the culture medium inhibits the photosynthetic activity of the cells. Further information might be obtained by repeating the study of the light effect on culture PEPc using cells washed with medium to remove sucrose and then resuspended in medium containing 3% sucrose or 1.6% mannitol. Mannitol is not metabolized and so at this concentration would act only to maintain the osmotic pressure of the cells, thus controlling for any involvement of photosynthate in a light effect on the malate sensitivity of PEPc.

No effect of light on the malate sensitivity of PEPc from the cell culture was seen (section 5.2.3) suggesting that the *A. thaliana* culture is not a good model for studying leaf PEPc. In contrast, barley leaf protoplasts were found to be a suitable model for the demonstration of light activation of C_3 PEPc (Lillo et al., 1995), resulting in a reduction

172

in the enzyme's sensitivity to the inhibitor L-malate (Smith et al., 1996). However, PEPc kinase activity increased with time after isolation of the barley leaf protoplasts in both light and darkness, although more so in the light than in the dark. The increase in kinase activity in dark-incubated protoplasts contrasts with the increase in malate sensitivity of PEPc over the same period, and indicates that, in contrast with the situation in C_4 and CAM plants, in C_3 barley leaf protoplasts PEPc kinase activity and the malate sensitivity of PEPc are not necessarily linked (Smith et al., 1996).

The availability of full-length cDNAs for both PEPc (see Chapter 4; Cloning and Sequencing of First Full-Length cDNA for Phosphoenolpyruvate Carboxylase from A. *thaliana*) and PEPc kinase from A. *thaliana* (see section 5.1) made it possible to study expression of the enzymes using a molecular approach.

Northern analysis of poly A⁺ RNA from different A. thaliana tissues provided information on the level of transcript for the A. thaliana-specific PEPc cDNA (discussed in Chapter 4), any other PEPc isoforms and A. thaliana-specific PEPc kinasc cDNA in each tissue. A radiolabelled fragment from the 3'-untranslated region of the A, thalianaspecific PEPc cDNA resulted in hybridization to root tissue only (Figure 5.8A lane 1). As the probe used was specific to the sequence of that PEPc cDNA, it would appear that the cDNA cloned is a root isoform of PEPc. Phylogenetic analysis of this PEPc sequence (section 4.2.5) is consistent with this expression data. A PEPc sequence conserved between different PEPc genes was amplified from the A. thaliana PEPc cDNA and also used to probe the tissue poly A7 RNA. This resulted in hybridization to root and flower and bud tissue only (Figure 5.8B lanes 1 and 6). As the sequence used is highly similar to other PEPcs, it would appear that PEPc isoforms are simply not detectable at the transcript level in silique, leaf, bract and bolt tissue. In contrast, the A. thaliana PEPc kinase probe gave a hybridization signal in all plant tissues (Figure 5.8C). These observations of the two enzymes at the transcript level are consistent with the knowledge that PEPc kinase protein turns over very rapidly (Nimmo, 1993; Hartwell et al., 1996).

Tubulin was employed as a constitutive control for the quantification of PEPc and PEPc kinase transcript levels using poly A^+ RNA northerns. It has been shown using RNA gel blot hybridizations and gene-specific probes, that the transcript levels of all fifteen tubulin genes in *A. thaliana* are decreased by 24 hours of continuous white light, although to different degrees (Leu et al., 1995). Tubulin is therefore, strictly speaking, not

173

constitutively expressed but, in the absence of any known constitutively expressed gene, tubulin was used for the purpose of standardizing the levels of the other transcripts examined.

A hybridization signal was also detected using both PEPc probes and the PEPc kinase probe on poly A⁻ RNA from *A. thaliana* cell culture. As discussed earlier, one of the attractions of using cell culture is the ease with which metabolic conditions can be altered. The ability to measure the transcript level of both PEPc and PEPc kinase in the *A. thaliana* culture meant that the system could be used to investigate the effect of different metabolites on the transcript level of the two enzymes. PEPc is one of many genes known to be regulated by both sugars and nitrate (see Koch, 1996 for review), not suprisingly considering the central role it is assumed to have in the control of photosynthetic carbon flow between sucrose synthesis and amino acid biosynthesis (Champigny and Foyer, 1992; Huber et al., 1994). Both photosynthetic carbon metabolism and nitrogen metabolism have also been implicated in the regulation of PEPc kinase activity in C₃ species (Duff and Chollet, 1995; Li et al., 1996; Scheible et al., 1997; Wadham et al., 1996). The regulatory effect of sucrose and nitrate on PEPc and PEPc kinase transcript level was therefore studied in the *A. thaliana* cell culture.

When the sucrose concentration of the culture medium was altered, the level of transcripts detected by the PEPc kinase probe and both PEPc probes was higher when the cells were grown in 1% sucrose compared with 3% sucrose. This would suggest that high photosynthate levels cause a reduction in the level of transcript for PEPc and PEPc kinase in the culture. Considering the presumed anaplerotic role of PEPc in the C_3 cell culture, it is possible that PEPc kinase acts as a "pull" for carbon and so when less carbon is available there would be higher levels of PEPc and PEPc kinase transcript to utilise what available carbon there is for amino acid biosynthesis. There was no significant effect of light on the transcript levels in 3% sucrose grown cells, consistent with the results obtained when investigating the effect of light on the malate sensitivity of culture PEPc (section 5.2.3). However, there was a definite light enhancement of the 3°-UTR transcript in 1% sucrose grown cells (Figure 5.10A lanes 8 and 9). This is rather surprising considering that this transcript encodes a putative root isoform of PEPc. Under the same metabolic conditions, the PEPc kinase transcript level was higher in the dark than the light (Fig 5.10C lanes 8 and 9).

In addition to the effects of sucrose mentioned above, it would be interesting to study the effect of nitrogen supply on PEPc and PEPc kinase. Duff and Chollet (1995) studied the short-term effect of nitrate feeding on the activity and malate sensitivity of C_3 PEPc, and observed little difference between illuminated or darkened wheat leaves detached from low-N-grown plants. In comparison, a rather dramatic increase in PEPc kinase activity was observed with illuminated excised leaves fed exogenous nitrate. However, no trend was apparent in the response of either of the PEPc transcripts or PEPc kinase transcript to increasing concentrations of nitrate in the cell culture medium.

It would appear that washing the cells to remove traces of growth medium affects the level of all three transcripts and the tubulin transcript grown under 3% sucrose (Figure 5.10A-D lanes 1 and 2). An effect of resuspending the cells, even in similar medium, can also be seen on transcript level (Figure 5.10A-D lanes 1 and 3). Therefore, it is quite plausible that the cells are stressed as a result of the experimental treatments and this might explain the absence of any consistent response of PEPc and PEPc kinase transcript to the alteration of nitrate levels. Equally these data might just reflect a lack of precision in this type of experiment, especially when the tubulin control is not ideal. In either case, it is questionable whether manipulation of the metabolic conditons of the cell culture is a suitable experimental system for studying the control of PEPc

Close inspection of Figure 5.10B reveals the presence of a band much lower in its molecular weight than the PEPc hybridization signal (lanes 8-10 and 12). Its size would suggest that it is not an isoform of PEPc and the hybridization pattern is not similar to that of any other transcript analyzed to be explained by filters not being stripped completely of previously hybridized probes. The band is only evident in 1% sucrose-grown cells but again there is no obvious trend in response to nitrate concentration. Its identity is still unclear.

The aim of these experiments was to discover something of the regulation of PEPc in the C_3 species *A. thaliana*. The work described in this chapter has established that the activity of PEPc and its malate sensitivity in the *A. thaliana* suspension culture is comparable to that of other C_3 species and is affected by the phosphorylation state of the PEPc protein. However, the light activation of PEPc demonstrated previously in other C_3 species was not seen with either *A. thaliana* culture or leaves (Rajagopalan et al., 1993; Lillo et al., 1995; Smith et al., 1996). Northern analysis of poly A⁺ RNA from *A*.

100

thaliana culture and different plant tissues showed the presence of PEPc kinase transcript, a putative root isoform PEPc transcript and the transcripts of other PEPc isoforms. The level of all three transcripts was higher in culture grown in 1% sucrose as opposed to 3% sucrose (Figure 5.10) but there was no obvious trend in the effect of nitrate level on any of the transcripts.

Therefore, whilst these experiments have not facilitated any further understanding of the regulation of PEPc in C_3 species, they have confirmed what was already known from studies using other C_3 species and have highlighted the realistic potential and pitfalls of using the *A. thaliana* culture as an experimental system for future research.

-

Chapter Six

DISCUSSION

The challenge of scientific research lies not only in understanding the working of the world in which we live but how our improved understanding can be applied to better the way we live. In consideration of its important role in photosynthesis, its involvement in nitrogen metabolism and its contribution to good "house-keeping" in the plant cell, the regulation of PEPe has been the focus of intensive research for about 15 years. As knowledge and understanding of PEPe regulation increases so does appreciation of the complex mechanisms involved and the ability to manipulate these mechanisms to modify plant metabolism.

The specific objectives of this present work as outlined in Chapter 1 were to investigate the involvement of cytosolic pH in the regulation of the CAM isoform of PEPc and to study the regulation of PEPc in C_3 species using an *A. thaliana* cell culture. The findings of these investigations and the conclusions drawn from them are presented in the preceding chapters.

The regulatory phosphorylation of PEPc is probably the most studied example in plants of this form of reversible covalent modification. However, very little is known about the signal transduction processes that lead to phosphorylation of PEPc, particularly in CAM and C_3 plants. The work described in Chapter 3 tested the idea that intracellular pH changes may be involved in this signalling in a CAM plant, as they are in C_4 species. While an effect of cytosolic acidification was observed, it seemed to be too slow to play a role in signal transduction.

The cloning of the first *A. thaliana* PEPc cDNA is described in Chapter 4. The sequence confirms that it can be phosphorylated. Unfortunately the clone was not expressed *in vitro* to provide a substrate for further studies during the course of this project. Use of an *A.* thaliana cell culture confirmed that the machinery required to phosphorylate PEPc is present in this C_3 species. Furthermore, it seems that there is more PEPc kinase present in culture grown in 1% than 3% sucrose.

The biggest development during the course of this project has been the cloning of the first PEPc kinase. This allows numerous developments. Results from previous investigations of PEPc kinase activity in different plant species suggest transcriptional regulation of the enzyme. However, the kinase activity assay of *in vitro* translation products and RNA synthesis inhibitors used in these experiments only in fact gives information about the steady-

presence of 32P-unit and then isolating the total RNA and using as a prebe for slot Hols containing 5-5 DNA (Dean of al, 1969)

---- Þ-

state level of the transcript i.e. the balance between synthesis and degradation. The RT-PCR and Northern analyses of PEPc kinase described in Chapter 5 do not overcome this limitation. The existence of a PEPc kinase clone will now enable nuclear run-on assays to be performed which should clarify the exact nature of the regulation of PEPc kinase at the transcriptinal level. It is also possible to make constructs of the PEPc kinase promoter and a reporter gene like β -glucuronidase which will facilitate studies of PEPc kinase expression in different tissues, under different conditions and at different stages of development. Isolation of the promoter would also allow investigation of transcription factors which regulate PEPc kinase, providing access to steps of the PEPc signalling pathway upstream of those already examined. It is difficult to extrapolate in vitro study of enzyme regulation to the in vivo situation of an intact plant. The genetic transformation of plants, although having its own limitations, provides an excellent means of studying the regulation and importance of an enzyme in its usual environment. Given the importance of protein phosphorylation in the regulation of PEPc in CAM plants and other species, it is likely that large-scale changes in the activity of PEPc kinase by overexpression or suppression of the PEPc kinase gene would result in easily observed changes in the plant phenotype which might enhance understanding of the role of the kinase in the plant. Transgenic plants in which the control of PEPc, a key enzyme of central metabolism, is altered could have advantageous phenotypic changes. The cloning of a PEPc and PEPc kinase from A, thaliana is particularly exciting due to the advantages of this species for genetic studies; it grows from seed to seed in about 6 weeks, it has a relatively small genome which has been extensively mapped, it has been successfully transformed and collections of A. thaliana tagged lines exist which could be screened for PEPc or PEPc kinase mutants. Knockout mutations of PEPc or PEPc kinase would be more revealing than the plants in which a gene has been antisensed.

A second recent development in PEPc research is the determination of the first crystallographic structure of PEPc from *E. coli* (Kai et al., 1999). Nucleotide sequence comparisons have been indispensable in understanding the biochemistry and regulation of PEPc. However, it is hoped that plant PEPc crystallographic structures will soon be available which will facilitate further site-directed mutagenesis studies and analysis of structure/function relationships.

Therefore two major hurdles in understanding the regulation of PEPc appear to have been cleared - the cloning of PEPc kinase and the elucidation of the first three-dimensional structure of PEPc. These achievements will make possible many research strategies which could greatly advance the understanding of PEPc regulation in higher plants in the next few years.

8 C.

-:.

ر معمد البنان

Chapter Seven REFERENCES

ALLEN, JF (1995) Thylakoid protein phosphorylation, state 1-state 2 transition, and photosystem stoichiometry adjustment: redox control at multiple levels of gene expression. Plant Physiol. <u>93</u>: pp 96-105

ANDERSON, CM and WILKINS, MB (1989) Period and phase control by temperature in the circadian rhythm of carbon dioxide fixation in illuminated leaves of *Bryophyllum fedtschenkoi*. Planta <u>177</u>: pp 456-469

ANDREO, CS, GONZALEZ, DH and IGLESIAS, AA (1987) Higher plant phosphoenolpyruvate carboxylase. Structure and regulation. FEBS Letts <u>213(1)</u>: pp 1-8

BAKRIM, N, ECHEVARRIA, C, CRETIN, C, ARRIO-DUPONT, M, PIERRE, J-N, VIDAL, J, CHOLLET, R and GADAL, P (1992) Regulatory phosphorylation of *Sorghum* leaf phosphoenolpyruvate carboxylase. Identification of the protein-serine kinase and some elements of the signal-transduction cascade. Eur. J. Biochem. <u>204</u>: pp 821-830

BENTON, WD and DAVIES, RW (1977) Screening λ gt recombinant clones by hybridization to single plaques *in situ*. Science 196: pp 180

BORLAND, AM, HARTWELL, J, JENKINS, GI, WILKINS, MB and NIMMO, HG (1999) Metabolite control over rides circadian regulation of phosphoenolpyruvate carboxylase kinase and CO₂ fixation in Crassulacean Acid Metabolism. Plant Physiol. (accepted).

BRULFERT, J, VIDAL, J, LE mARECHAL, P, GADAL, P, QUEIROZ, O, KLUGE, M and KRUGER, I (1986) Phosphorylation-dephosphorylation process as a probable mechanism for the diurnal regulatory changes of phosphocnolpyruvate carboxylase in CAM plants. Biochem. Biophys. Res. Commun. <u>136(1)</u>: pp 151-159

BUDDE, RJA and CHOLLET, R (1986) In vitro phosphorylation of maize leaf phosphoenolpyruvate carboxylase. Plant Physiol. <u>82</u>: pp 1107-1114

CANVIN, DT and SALON, C (1997) Photorespiration and CO₂-concentrating mechanisms. In: Plant Metabolism. (eds. Dennis, DT, Turpin, DH, Lefebvre, DD and Layzell, DB) p300. Addison Wesley Longman Ltd., Harlow, England. ISBN 0-582-25906-1

CARTER, PJ, NIMMO, HG, FEWSON, CA and WILKINS, MB (1990) Bryophyllum fedtschenkoi protein phosphatase type 2A can dephosphorylate phosphoenolpyruvate carboxylase. FEBS Letts <u>263</u>: pp 233-236

CARTER, PJ, NIMMO, HG, FEWSON, CA and WILKINS, MB (1991) Circadian rhythms in the activity of a plant protein kinase. EMBO J. <u>10</u> (8): pp 2063-2068

CARTER, PJ, WILKINS, MB, NIMMO, HG and FEWSON, CA (1995) Effects of temperature on the activity of phosphoenolpyruvate carboxylase and on the control of CO₂ fixation in *Bryophyllum fedtschenkoi*. Planta <u>196</u>: pp 375-380

CARTER, PJ, WILKINS, MB, NIMMO, HG and FEWSON, CA (1995) The role of temperature in the regulation of the circadian rhythm of CO_2 fixation in *Bryophyllum fedtschenkoi*. Planta <u>196(2)</u>: pp 381-386

CARTER, PJ, FEWSON, CA, NIMMO, GA, NIMMO, HG and WILKINS, MB (1996) Roles of circadian rhythms, light and temperature in the regulation of phosphoenolpyruvate carboxylase in Crassulacean Acid Metabolism. Ecological Studies <u>114</u> **CHAMPIGNY, M-L and FOYER,C** (1992) Nitrate activation of cytosolic protein kinases diverts photosynthetic carbon from sucrose to amino acid biosynthesis. Plant Physiol. <u>100</u>: pp. 7-12

CHAO, S, RAINES, CA, LONGSTAFF, M, SHARP, PJ, GALE, MD and DYER, TA (1989) Chromosomal location and copy number in wheat and some of its close relatives of genes for enzymes involved in photosynthesis. Mol. Gen. Genet. <u>218</u>: pp 423-435

CHARDOT, TP and WEDDING, R (1992) Role of cysteine in activation and allosteric regulation of maize phosphoenolpyruvate carboxylase. Plant Physiol. <u>98</u>: pp 780 - 783

CHOLLET, R, VIDAL, J and O'LEARY, MH (1996) Phosphoenolpyruvate carboxylase: A ubiquitous, highly regulated enzyme in plants. Ann. Rev. of Plant Physiol. and Plant Mol. Biol. <u>47</u>: pp 185-214

CRÉTIN, C, SANTI, S, KERYER, E, LEPINIEC, L, TAGU, D, VIDAL, J and GADAL, P (1991) The phosphoenolpyruvate carboxylase gene family of Sorghum: promoter structures, amino acid sequences and expression of genes. Gene 99: pp 87-94

CUSHMAN, JC, MEYER, G, MICHALOWSKI, CB, SCHMITT, JM and BOHNERT, HJ (1989) Salt stress leads to differential expression of two isogenes of

phosphoenolpyruvate carboxylase during Crassulacean Acid Metabolism induction in the common ice plant. The Plant Cell <u>1</u>: pp 715-725

1

D'ALESSIO, JM (1992) Lambda ZipLox: Automatic subcloning of cDNA. Focus <u>14(3)</u>: pp 76-79

DEROCHE, M-E and CARRAYOL, E (1988) Nodule phosphoenolpyruvate carboxylase: a review. Physiol. Plant. <u>74</u>: pp 775-782

DEVI, MT and RAGHAVENDRA, AS (1992) Light activation of phosphoenolpyruvate carboxylase in maize mesophyll protoplasts. J. Plant Physiol. <u>139</u>(4): pp 431-435

DIAZ, A, MAZA, H, GONZALEZ-MORO, B, LACUESTA, M, GONZALEZ-MURUA, C and MUNOZ-RUEDA, A (1995) Phosphinothricin reverts the ammoniadependent enhancement of phosphoenolpyruvate carboxylase activity. J. Plant Physiol. 145(1-2): pp 11-16

DIAZ, A, LACUESTA, M and MUNOZ-RUEDA, A (1996) Comparative effects of phosphinothricin on nitrate and ammonium assimilation and on anaplerotic CO_2 fixation in N-deprived barley plants. J. Plant Physiol. <u>149</u>(1-2): pp 9-13

DONCASTER, HD and LEEGOOD, RC (1987) Regulation of phosphoenolpyruvate carboxylase activity in maize leaves. Plant Physiol. <u>84</u>: pp 82-87

DOUGLAS, P, MORRICE, N and MacKINTOSH, C (1995) Identification of a regulatory phosphorylation site in the hinge region of nitrate reductase from Spinach (*Spinacea oleracea*) leaves. FEBS Lett. <u>377(2)</u>: pp 113-117

DOUGLAS, P, PIGAGLIO, E, FERRER, A, HALFORD, NG and MacKINTOSH, C (1997) Three spinach leaf nitrate reductase-3-hydroxy-3-methylglutaryl-CoA reductase kinases that are regulated by reversible phosphorylation and/or Ca²⁺ ions. Biochem. J. <u>325</u>: pp 101-109

DOUGLAS, P, MOORHEAD, G, YONG, Y, MORRICE, N and MacKINTOSH, C (1998) Purification of a nitrate reductase kinase from *Spinacea oleracea* leaves, and its identification as a calmodulin-domain protein kinase. Planta <u>206</u>: pp 435-442

DU, **ZR**, **AGHORAM**, **K** and **OUTLAW JR.**, **WH** (1997) *In vivo* phosphorylation of phosphoenolpyruvate carboxylase in guard cells of *Vicia faba* L. is enhanced by fusicoccin and suppressed by abscisic acid. Arch. Biochem. Biophys. <u>337</u>(2): pp 345-350

DUFF, SMG and CHOLLET, R (1995) *In vivo* regulation of wheat leaf phosphoenolpyruvate carboxylase by reversible phosphorylation. Plant Physiol. <u>107</u>: pp. 775-782

DUFF, SMG, ANDREO, CS, PACQUIT, V, LEPINIEC, L, SARATH, G, CONDON, SA, VIDAL, J, GADAL, P and CHOLLET, R (1995) Kinetic analysis of the nonphosphorylated, *in vitro* phosphorylated and phosphorylation site mutant (Asp8) forms of intact recombinant C₄ phosphoenolpyruvate carboxylase from *Sorghum*. Eur. J. Biochem. <u>228(1)</u>: pp 92-95

DUFF, SMG, GIGLIOLI-GUIVARC'H, N, PIERRE, J-N, VIDAL, J, CONDON, SA and CHOLLET, R (1996) In situ evidence for the involvement of calcium and bundlesheath-derived photosynthetic metabolites in the C_4 phosphoenolpyruvate carboxylase kinase signal transduction chain. Planta <u>199(3)</u>: pp 467-474

ECHEVARRIA, C, VIDAL, J, JIAO, J-A and CHOLLET, R (1990) Reversible light activation of the phosphoenolpyruvate carboxylase protein-serine kinase in maize leves. FEBS Letts. <u>275(1,2)</u>: pp 25-28

ECHEVARRIA, C, PACQUIT, V, BAKRIM, N, OSUNA, L, DELGADO, B, ARRIO-DUPONT, M and VIDAL, J (1994) The effect of pH on the covalent and metabolic control of C_4 phosphoenolpyruvate carboxylase from Sorghum leaf. Arch. Biochem. Biophys. <u>315(2)</u>: pp 425-430

FERL, RJ (1996) 14-3-3 proteins and signal transduction. Ann. Rev. Plant Physiol. Plant Mol. Biol. <u>47</u>: pp 49-73

GIGLIOLI-GUIVARC'II, N, PIERRE, JN, VIDAL, J and BROWN, S (1996) Flow cytometric analysis of cytosolic pH of mesophyll cell protoplasts from the crabgrass Digitaria sanguinalis. Cytometry 23(3): pp 241-249

GIGLIOLI'GUIVARC'H, N, PIERRE, J-N, BROWN, S, CHOLLET, R, VIDAL, J and GADAI, P (1996) The light-dependent transduction pathway controlling the regulatory phosphorylation of C_4 phosphoenolpyruvate carboxylase in protoplasts from *Digitaria sanguinalis*. Plant Cell <u>8</u>(4): pp 573-586

GUPTA, SK, KU, MSB, LIN, J-H, ZHANG, D and EDWARDS, GE (1994) Light/dark modulation of phosphoenolpyruvate carboxylase in C_3 and C_4 species. Photosynthesis Research <u>42</u>: pp. 133-143

HARDIE, DG, CORTON, J, CHING, YP, DAVIES, SP and HAWLEY, S (1997) Regulation of lipid metabolism by the AMP-activated protein kinase. Biochem. Society Transactions 25(4): pp 1229-1231

HARDIE, DG (1999) Plant protein serine/threonine kinases: Classification and functions. Annu. Rev. Plant. Physiol. Plant Mol. Biol. <u>50</u>: pp 97-131

HANKS, SK and HUNTER, T (1995) Protein kinases and the eukaryotic protein-kinase superfamily - kinase (catalytic) domain structure and classification. FASEB J. <u>9</u>(8): pp 576-596

HARMON, AC, PUTNAMEVANS, C and CORMIER, MJ (1987) A calcium-dependent but calmodulin-independent protein-kinase from Soybean. Plant Physiol. <u>83</u>(4): pp 830-837

HARTWELL, J, SMITH, LH, WILKINS, MB, JENKINS, GI and NIMMO, HG (1996) Higher plant phosphoenolpyruvate carboxylase kinase is regulated at the level of translatable mRNA in response to light or a circadian rhythm. Plant Journal. <u>10</u>(6): pp 1071-1078

HARTWELL, J, JENKINS, GI, WILKINS, MB and NIMMO, HG (1999) The light induction of maize phosphoenolpyruvate carboxylase kinase translatable mRNA requires transcription but not translation. Plant, Cell Environ. <u>22</u>: pp 883-889

HEMERLEY, A, ENGLER, JD, BERGOUNIOUX, C VANMONTAGU, M and ENGLER, G (1988) Dominant negative mutants of the cdc2 kinase uncouple cell division from iterative plant development. EMBO J. <u>14</u>: pp 3925-3936

HONDA, H, OKAMOTO, T and SIIIMADA, H (1996) Isolation of a cDNA for a phosphoenolpyruvate carboxylase from a monocot CAM plant, *Aloe arborescens*: structure and its gene expression. Plant and Cell Physiology <u>37</u>(6): pp 881-888

HORTON, P et al. (1981) Regulation of phosphorylation of chloroplast membrane polypeptides by the redox state of plastiquinone. FEBS Lett. <u>125</u>: pp 193-196

HUBER, SC and HUBER, JL (1992) Role of sucrose-phosphate synthase in sucrose metabolism in leaves. Plant Physiol. <u>99</u>: pp 1275-1278

HUBER, SC, HUBER, JL and KAISER, WM (1994) Differential response of nitrate reductase and sucrose phosphate synthase activation to inorganic salts, *in vitro* and *in situ*. Physiol. Plant. <u>92</u>(2): pp 302-310

HUBER, SC and HUBER, JL (1995) Metabolic activation of spinach leaf nitrate reductase - effects on enzymatic activity and dephosphorylation by endogenous phosphatases. Planta <u>196(1)</u>: pp 180-189

HUBER, SC, BACHMANN, M and HUBER, JL (1996) Post-translational regulation of nitrate reductase activity: a role for Ca^{2+} and 14-3-3 proteins. TIPS 1(12): pp 432 - 438

HUBER, SC, HUBER, JL, LIAO, P-C, GAGE, DA, McMICHAEL JR., RW, CHOUREY, PS, HANNAH, LC and KOCH, K (1996)Phosphorylation of serine-15 of maize leaf sucrose synthase. Plant Physiol. <u>112</u>: pp 793-802

IGLESIAS, AA and ANDREO, CS (1984) Involvement of thiol groups in the activity of phosphoenolpyruvate carboxylase from maize leaves. Photosyn. Res. <u>5</u>(3): pp 215-226

ISRAEL, MW and JACKSON, WA (1982) Ion balance, uptake and transport processes in N_2 -fixing and NO_3 -dependent and urea-sependent soybean plants. Plant Physiol. <u>69(1)</u>: pp 171-178

IZUI, K, ISHIJIMA, S, YAMAGUCHI, Y, KATAGIRI, F, MURATA, T, SHIGESADA, K, SUGIYAMA, T and KATSUKI, H (1986) Cloning and sequence analysis of cDNA encoding active phosphoenolpyruvate carboxylase of the C4-pathway from maize. Nucleic Acids Research <u>14</u>(4): pp 1615-1628

JACKSON, JA, FUGLEVAND, G, BROWN, BA, SHAW, MJ and JENKINS, GI (1995) Isolation of *Arabidopsis* mutants altered in the light-regulation of chalcone synthase gene expression using a transgenic screening approach. The Plant Journal <u>8</u>(3): pp 369-380

JIAO, J and CHOLLET, R (1988) Light/dark regulation of maize leaf phosphoenolpyruvate carboxylase by *in vivo* phosphorylation. Archives Biochem. Biophs. <u>261(2)</u>: pp 409-417

JIAO, J-A and CHOLLET, R (1989) Regulatory seryl-phosphorylation of C4 phosphoenolpyruvate carboxylase by a soluble protein kinase from maaize leaves. Arch. Biochem. Biophys. 269(2): pp 526-535

JIAO, J-A, PODESTA, FE, CHOLLET, R, O'LEARY, MH and ANDREO, CS (1990) Isolation and sequence of an active site peptide from maize leaf phosphoenolpyruvate carboxylase inactivated by pyridoxal 5'-phosphate. Biochem. Biophys. Acta <u>1041</u>: pp 291 - 295

JIAO, J-A and CHOLLET, R (1991) Posttranslational regulation of phosphoenolpyruvate carboxylase in C₄ and Crassulacean Acid Metabolism plants. Plant Physiol. <u>95</u>: pp 981-985

JIAO, J-A, ECHEVARRIA, C, VIDAL, J and CHOLLET, R (1991) Protein turnover as a component in the light/dark regulation of phosphoenolpyruvate carboxylase proteinserine kinase activity in C₄ plants. Proc. Natl. Acad. Sci. USA <u>88</u>: pp 2712-2715

JIAO, J and CHOLLET, R (1992) Light activation of maize phosphoenolpyruvate carboxylase protein-serine kinase activity is inhibited by mesophyll and bundle sheath-derived photosynthesis inhibitors. Plant Physiol. 28: pp 152-156

JOSHI, CP (1987) Putative polyadenylation signals in nuclear genes of higher plants - a compilation and analysis. Nucl Acid Res <u>15</u>: pp 9627-9641

KAI, Y (1997) X-ray structure of phosphoenolpyruvate carboxylase. Plant Physiol. <u>114</u>(3)SS: pp 220

KAI, Y, MATSUMURA, H, INOUE, T, TERADA, K, NAGARA, Y, YOSHINAGA, T, KIHARA, A, TSUMURA, K and IZUI, K (1999) Three-dimensional structure of phosphoenolpyruvate carboxylase: A proposed mechanism for allosteric inhibition. Proc. Natl. Acad. Sci. USA <u>96</u>: pp 823-828

KAWAMURA, T, SHIGESADA, K, YANAGISAWA, S and IZUI, K (1990) Phosphoenolpyruvate carboxylase prevalent in maize roots: Isolation of a cDNA clone and its use for analyses of the gene and gene expression. J. Biochem. <u>107</u>: pp 165-168

KAWAMURA, T, SHIGESADA, K, TOH, H, OKUMURA, S, YANAGISAWA, S and IZUI, K (1992) Molecular evolution of phosphoenolpyruvate carboxylase for C_4 photosynthesis in maize: Comparison of its cDNA sequence with a newly isolated cDNA encoding an isozyme involved in the anaplerotic function. J. Biochem. <u>112</u>: pp 147-154

KOCH, KE (1996) Carbohydrate-modulated gene expression in plants. Annu. Rev. Plant Physiol. Plant Mol. Biol. <u>47</u>: pp 509-540

KROMER, S, GARDESTROM, P and SAMUELSSON, G (1996) Regulation of the supply of cytosolic oxaloacetate for mitochondrial metabolism via phosphoenolpyruvate carboxylasein barley leaf protoplasts. 1. The effect of covalent modification on PEPc activity, pH response, and kinetic properties. Biochem. Biophys. Acta - General Subjects **1289**(3): pp 343-350

KROMER, S, GARDESTROM, P and SAMUELSSON, G (1996) Regulation of the supply of oxaloacetatefor mitochondrial metabolism via phosphoenolpyruvate carboxylase in barley leaf protoplasts. 2. Effects of metabolites on PEPc activity at different activation states of the protein. Biochem. Biophys. Acta - General Subjects <u>1289(3)</u>: pp 351-361

60

LAEMMLI, UK (1970) Cleavage of structural protein during the assembly of the head of bacteriophage T4. Nature <u>227</u>: pp 680-685

LEEGOOD, RC, VON CAEMMERER, S and OSMOND, CB (1997) Metabolite transport and photosynthetic regulation in C4 and CAM plants. In: Plant Metabolism. (eds. Dennis, DT, Turpin, DH, Lefebvre, DD and Layzell, DB) p 342. Addison Wesley Longman Ltd., Harlow, England. ISBN 0-582-25906-1

LEPINIEC, L, KERYER, E, PHILIPPE, H, GADAL, P and CRETIN, C (1993) Sorghum phosphoenolpyruvate carboxylase gene family: structure, function and molecular evolution. Plant Mol. Biol. <u>21</u>: pp. 487-502

LEPINIEC, L, VIDAL, J, CHOLLET, R, GADAL. P and CRETIN, C (1994) Phosphoenolpyruvate carboxylase: structure, regulation and evolution. (Review article) Plant Science <u>99</u>: pp 111-124

LEPORT, L, KANDLBINDER, A, BAUR, B and KAISER, WM (1996) Diurnal modulation of phosphoenolpyruvate carboxylation in pea leaves and roots as related to tissue malate concentrations and to the nitrogen source. Planta <u>198</u>(4): pp 495-501

LEU, W-M, CAO, X-L, WILSON, TJ, SNUSTAD, DP and CHUA, N-H (1995) Phytochrome A and phytochrome B mediate the hypocotyl-specific downregulation of *TUB1* by light in Arabidopsis. The Plant Cell <u>7</u>: pp 2187-2196

LI, BN and CHOLLET, R (1993) Resolution and identification of C_4 phosphoenolpyruvate carboxylase protein kinase polypeptides and their reversible light activation in maize leaves. Arch. Biochem. Biophys. <u>307</u>(2): pp 416-419

LI, BN and CHOLLET, R (1994) Salt induction and the partial purification/characterization of phosphoenolpyruvate carboxylase protein-serine kinase from an inducible Crassulacean-acid-metabolism (CAM) plant, Mesembryanthemum crystallinum L. Arch. Biochem. Biophys. <u>314</u>: pp 247-254

LI, BN and CHOLLET, R (1995) Reversible light activation of tobacco leaf phosphoenolpyruvate carboxylase (PEPc) kinase. Plant Physiology <u>108</u>(2) SS: pp 141

LI, B, ZHANG, X-Q and CHOLLET, R (1996) Phosphoenolpyruvate carboxylase kinase in tobacco leaves is activiated by light in a similar but not identical way as in maize. Plant Physiol. <u>111(2)</u>: pp 497-505

LILLO, C, SMITH, LH, NIMMO, HG and WILKINS, MB (1996) Regulation of nitrate reductase and phosphoenolpyruvate carboxylase activities in barley leaf protoplasts. Planta 200(2): pp 181-185

MACDONALD, FD and BUCHANAN, BB (1997) The reductive pentose phosphate pathway and its regulation. In: Plant Metabolism. (eds. Dennis, DT, Turpin, DH, Lefebvre, DD and Layzell, DB) p 300. Addison Wesley Longman Ltd., Harlow, England. ISBN 0-582-25906-1

MacKINTOSH, C and COHEN, P (1989) Identification of high levels of type 1 and type 2A protein phosphatases in higher plants. Biochem. J. <u>262</u>: pp 335-339

MacKINTOSH, C, COGGINS, J and COHEN, P (1991) Plant protein phosphatases. Subcellular distribution, detection of protein phosphatase 2C and identification of protein phosphatase 2A as the major quinate dehydrogenase phosphatase. Biochem. J. <u>273</u>: pp 733-738

MacKINTOSH, C (1992) Regulation of Spinach leaf nitrate reductase by reversible phosphorylation. Biochem. Biophys. Acta 1137(1): pp 121-126

MacKINTOSH, RW, DAVIES, SP, CLARKE, PR, WEEKES, J, GILLESPIE, JG, GIBB, BJ and HARDIE, DG (1992) Evidence for a protein kinase caseade in higher plants - 3-hydroxy-3-methylglutaryl-CoA reductase kinase. Eur. J. Biochem. 209: pp 923-931

MacKINTOSH, C and MacKINTOSH, RW (1994) Inhibitors of protein kinases and phosphatases. TIBS <u>19</u>: pp 444-448

MacKINTOSH, C, DOUGLAS, P and LILLO, C (1995) Identification of a protein that inhibits the phosphorylated form of nitrate reductase from Spinach (*Spinacia oleracea*) leaves. Plant Physiol. <u>107</u>(2): pp 451-457

MacKINTOSH, C (1997) Regulation of C/N metabolism by reversible phophorylation. In: A Molecular Approach to Primary Metabolism in Higher Plants. (eds. Foyer, CH and Quick, PW) pp 179-201. Taylor & Francis Ltd., London, England. ISBN 0-784-0419-8

MacKINTOSH, C (1998) Regulation of cytosolic enzymes in primary metabolism by reversible protein phosphorylation. Current Opinion in Plant Biology $\underline{1}(3)$: pp 224-229

MANH, CT, BISMUTH, E, BOUTIN, J-P, PROVOT, M and CHAMPIGNY, M-L

(1993) Metabolite effectors for short-term nitrogen-dependent enhancement of phosphoenolpyruvate carboxylase activity and decrease of net sucrose synthesis in wheat leaves. Physiol. Plant. <u>89</u>: pp 460-466

MATSUOKA, M and HATA, S (1987) Comparative studies of phosphoenolpyruvate carboxylase from C₃ and C₄ plants. Plant Physiol. <u>85</u>: pp 947-951

MATSUOKA, M and MINAMI, E (1989) Complete structure of the gene for phosphoenolpyruvate carboxylase from maize. European J. Biochem. <u>181</u>(3): pp 593-598

MAY, MJ and LEAVER, CJ (1993) Oxidative stimulation of glutathione synthesis in *Arabidopsis thaliana* suspension cultures. Plant Physiol. <u>103</u>: pp 621-627

McMICHAEL Jr., RW, KOCHANSKY, J, KLEIN, RR and HUBER, SC (1995) Characterization of the substrate specificity of sucrose-phosphate synthase protein kinase. Arch. Biochem. Biophys. <u>321(1)</u>: pp 71-75

McNAUGHTON, GAL, MacKINTOSH, C, FEWSON, CA, WILKINS, MB and NIMMO, HG (1991) Illumination increases the phosphorylation state of maize leaf phosphoenolpyruvate carboxylase by causing an increase in the activity of a protein kinase. Biochem. Biophys. Acta <u>1093</u>: pp 189-195

MELZER, E and O'LEARY, MH (1987) Anapleurotic CO_2 fixation by phosphoenolpyruvate carboxylase in C₃ plants. Plant Physiol. <u>84</u>: pp 58-60

MIZOGUCHI, T, GOTOH, Y, NISHIDA, E, YAMAGUCHISHINOZAKI, K and HAYASHIDA, N (1984) Characterization of 2 cDNAs that encode MAP kinase homologs in *Arabidopsis thaliana* and analysis of the possible role of auxin in activating such kinase activities in cultured cells. Plant J. 5: pp 111-122

MOORHEAD, G, DOUGLAS, P and MacKINTOSH, C (1997) Regulation of nitrate reductase and other enzymes of nitrogen metabolism by phosphorylation and 14-3-3 proteins. In: Protein Phosphorylation in Plants - Second European Symposium. (eds. Gadal, P, Kreis, M, Dron, M, Brulfert, J, Bergounioux, C and Vidal, J) pp 108-116 Institut de Biotechnologie des Plantes, Orsay, Paris, France.

NANMORI, T, TAGUCHI, W, KINUGUSA, M, OJI, Y, SAHARA, S, FUKAMI, Y and KIKKAWA, U (1994) Purification and characterization of protein kinase C from a higher plant, *Brassica campestris* L. Biochem. Biophys. Res. Commun. <u>203</u>: pp 311-318 NEWMAN, T, DE BRUIJN, FJ, GREEN, P, KEEGSTRA, K, KENDE, H, McINTOSH, L, OHLROGGE, J, RAIKHEL, N, SOMMERVILLE, S, THOMASHOW, M, RETZEL, E and SOMMERVILLE, C (1994) Genes galore: a summary of methods for accessing results from a large-scale partial sequencing of anonymous *Arabidopsis* cDNA clones. Plant Physiol. <u>106</u>: pp 1241-1255

NHIRI, M, BAKRIM, N, PACQUIT, V, HACHIM, ZE, OSUNA, L and VIDAL, J (1998) Calcium-dependent and -independent phosphoenolpyruvate carboxylase kinases in *Sorghum* leaves: Further evidence for the involvement of the calcium- independent protein kinase in the *in situ* regulatory phosphorylation of C₄ phosphoenolpyruvate carboxylase. Plant and Cell Physiol. <u>39</u>(2): pp 241-246

NIMMO, GA, NIMMO, HG, FEWSON, CA and WILKINS, MB (1984) Diurnal changes in the properties of phosphoenolpyruvate carboxylase in *Bryophyllum* leaves: a possible covalent modification. FEBS Letts. <u>178</u>(2): pp199-203

NIMMO, GA, NIMMO, HG, HAMILTON, ID, FEWSON, CA and WILKINS, MB (1986) Purification of the phosphorylated night form and dephosphorylated day form of phosphoenolpyruvate carboxylase from *Bryophyllum fedtschenkoi*. Biochem. J. <u>239</u>: pp 213-220

NIMMO, GA, McNAUGHTON, GAL, FEWSON, CA, WILKINS, MB and NIMMO, HG (1987) Changes in the kinetic properties and phosphorylation state of phosphoenolpyruvate carboxylase in *Zea mays* leaves in response to light and dark. FEBS Letts. <u>213(1)</u>: pp 18-22

NIMMO, GA, WILKINS, MB, FEWSON, CA and NIMMO, HG (1987) Persistent circadian rhythms in the phosphorylation state of phosphoenolpyruvate carboxylase from *Bryophyllum fedtschenkoi* leaves and in its sensitivity to inhibition by malate. Planta <u>170</u>: pp 408-415

NIMMO, HG (1993) The regulation of phosphoenolpyruvate carboxylase by reversible phosphorylation. In: Post-translational modification in plants (eds. Battey NH, Dickinson HG and Hetherington AM) pp 161-170. Cambridge University Press. ISBN 0-521-41181-5 NIMMO, HG (1998) Circadian regulation of a protein kinase. Chrono. Biol. Int. <u>15</u>(2): pp 109-118

OGAWA, N, YABUTA, N, UENO, Y and IZUI, K (1998) Characterization of a maize Ca+-dependent protein kinase phosphorylating phosphoenolpyruvate carboxylase. Plant Cell Physiol. <u>39</u>(10): pp 1010-1019

O'LEARY (1982) Phosphoenolpyruvate carboxylase: an enzymologist's view. Ann. Rev. Plant Physiol. <u>33</u>: pp 297-315

OSUNA, L, GONZALEZ, M-C, CEJUDO, FJ, VIDAL, J and ECHEVARRIA, C (1996) *In vivo* and *in vitro* phosphorylation of the phosphoenolpyruvate carboxylase from wheat seeds during germination. Plant Physiol. <u>111(2)</u>: pp 551-558

PACQUIT, V, SANTI, S, CRETIN, C, BUI, VL, VIDAL, J and GADAL, P (1993) Production and properties of recombinant C_3 -type phosphoenolpyruvate carboxylase from *Sorghum vulgare: In vitro* phosphorylation by leaf and root PyrPC protein serine kinases. Biochem. Biophys. Res. Comms. <u>197</u>(3): pp 1415-1423

PACQUIT, V, GIGLIOLI-GUIVARC'H, N, CRETIN, V, PIERRE, J-N, VIDAL, J and ECHEVARRIA, C (1995) Regulatory phosphorylation of C_4 phosphoenolpyruvate carboxylase from *Sorghum* - An immunological study using specific anti-phosphorylation site antibodies. Photosynthesis Research <u>43</u>(3): pp 283-288

PIERRE, JN, PACQUIT, V, VIDAL, J and GADAL, P (1992) Regulatory phosphorylation of phosphoenolpyruvate carboxylase in protoplasts from *Sorghum* mesophyll cells and the role of pH and Ca^{2+} as possible components of the light - transduction pathway. Eur. J. Biochem. <u>210</u>: pp 531-537

PODESTA, FE and ANDREO, CS (1989) Maize leaf phosphoenolpyruvate carboxylase - oligomeric state and activity in the presence of glycerol. Plant Physiol. 90(2): pp 427-433

RAGHAVENDRA, AS, YIN, ZH and HEBER, U (1993) Light-dependent pH changes in leaves of C_4 plants - Comparison of the pH response to CO_2 and O_2 with that of C_3 plants. Planta <u>189(2)</u>: pp 278-287

RAJAGOPALAN, **AV**, **DEVI**, **MT** and **RAGHAVENDRA**, **AS** (1993) Patterns of phosphoenolpyruvate carboxylase and cytosolic pH during light activation and dark deactivation in C_3 and C_4 plants. Photosynthesis Research <u>38</u>(1): pp 51-60

RAJAGOPALAN, AV, DEVI, MT and RAGHAVENDRA, AS (1994) Molecular biology of C_4 phosphoenolpyruvate carboxylase - Structure, regulation and genetic engineering. Photosynthesis Research <u>39</u>: pp 115-135 **RELLE, M and WILD, A** (1996) Molecular characterization of a phosphoenolpyruvate carboxylase in the gymnosperm *Picea abies* (Norway Spruce). Plant Mol. Biol. <u>32</u>: pp 923-936

1974

ROESKE, CA and CHOLLET, R (1989) Role of metabolites in the reversible light activation of pyruvate, orthophosphate dikinase in Zea mays mesophyll cells in vivo. Plant Physiol. <u>90</u>: pp 330-337

SAMARAS, Y, MANETAS, Y and GAVALAS, NA (1988) Effects of temperature and photosynthesis inhibitors on light activation of C₄ PEPc. Photosyn. Research <u>16(3)</u>: pp 233-242

SATO, F, KOIZUMI, N and YAMADA, Y (1988) Purification and characterization of phosphoenolpyruvate carboxylase of photomixotrophically cultured green tobacco cells. Plant Cell Physiol. <u>29</u>(2): pp 329-337

SCHEIBLE, WR, GONZALEZ-FONTES, A, LAURER, M, MULLER-ROBER, B, CABOCHE, M and STITT, M (1997) Nitrate acts as a signal to induce organic acid metaboliam and repress starch metabolism in tobacco. Plant Cell <u>9</u>(5): pp 783-798

SCHULLER, KA, TURPIN, DH and PLAXTON, WC (1990) Metabolite regulation of partially purified soybean nodule phosphoenolpyruvate carboxylase. Plant Physiol. <u>94</u>: pp 1429-1435

SCHULLER, KA and WERNER, D (1993) Phosphorylation of soybean (*Glycine max* L.) nodule phosphoenolpyruvate carboxylase *in vitro* decreases sensitivity to inhibition by L-malate. Plant Physiol. <u>101</u>: pp 1267 - 1273

SHEEN, J (1993) Protein phosphatase activity is required for light-inducible gene expression in maize. EMBO J. <u>12(9)</u>: pp 3497-3505

SILFLOW, CD, CHISHOLM, RL, CONNER, TW and RANUM, LP (1985) The two alpha-tubulin gene of *Chlamydomonas reinhardi* code for lightly different proteins. Mol. Cell Biol. <u>5</u>: pp 2389-2398

SMITH, CM, DUFF, SMG and CHOLLET, R (1994) Partial purification and characterization of maize-leaf pyruvate, orthophosphate dikinase regulatory protein: a low abundance, mesophyll chloroplast stromal protein. Arch. Biochem. Biophys. <u>308</u>: pp 200-206
SMITH, LH, LILLO, C, NIMMO, HG and WILKINS, MB (1996) Light regulation of phosphoenolpyruvate carboxylase in barley mesophyll protoplasts is modulated by protein ynthesis and calcium, and not necessarily correlated with phosphoenolpyruvate carboxylase kinase activity. Planta 200(2): pp 174-180

SMITH, RD and WALKER, JC (1996) Plant protein phosphatases. Ann. Rev. plant Physiol. and Plant Mol. Biol. <u>47</u>: pp 101-125

STONE, JM, COLLINGE, MA, SMITH, RD, HORN, MA and WALKER, JC (1994) Interaction of a protein phosphatase with an Arabidopsis serine-threonine receptor kinase. Science <u>266(5186)</u>: pp 793-795

STONE, JM and WALKER, JC (1995) Plant protein kinase families and signal transduction. Plant Physiol. <u>108</u>: pp 451-457

SUBRAMANIAM, R, DESPRÉS, C and BRISSON, N (1997) A functional homolog of mammalian protein kinase C participates in the elicitor-induced defense response in potato. The Plant Cell **2**: pp 653-664

SUGANO, S, ANDRONIS, C, GREEN, RM, WANG, Z-Y and TOBIN, EM (1998) Protein kinase CK2 interacts with and phosphorylates the *Arabidopsis* circadian clockassociated 1 protein. Proc. Natl. Acad. Sci. USA <u>95</u>: pp 11020-11025

SUGDEN, C, CRAWFORD, RM, HALFORD, NG and HARDIE, DG (1999) Regulation of spinach SNF1-related (SnRK1) kinases by protein kinases and phosphatases is associated with phosphorylation of the T loop and is regulated by 5'-AMP. The Plant J. <u>19(4)</u>: pp 433-439

TARCZYNSKI, MC and OUTLAW JR., WH (1993) The interactive effects of pH, Lmalate, and glucose-6-phosphate on guard cell phosphoenolpyruvate carboxylase. Plant Physiol. <u>103</u>: pp 1189-1194

TERADA, K, KAI, T, OKUNO, S, FUJISAWA, H and IZUI, K (1990) Maize leaf phosphoenolpyruvate carboxylase: phosphorylation of Ser¹⁵ with a mammalian cyclic AMP-dependent protein kinase diminishes sensitivity to inhibition by malate. FEBS Letts. **259**(2): pp 241-244

TOH, H, KAWAMURA, T and IZUI, K (1994) Molecular evolution of phosphoenolpyruvate carboxylase. Plant Cell and Environment <u>17(1)</u>: pp 31-43

VAN QUY, L, FOYER, C and CHAMPIGNY, M-L (1991) Effect of light and NO₃⁻ on wheat leaf phosphoenolpyruvate carboxylase activity. Plant Physiol. <u>97</u>: pp. 1476-1482

VAN QUY, L LAMAZE, T and CHAMPIGNY, M-L (1991) Short-term effects of nitrate on sucrose synthesis in wheat leaves. Planta <u>185</u>: pp 53-57

VAN QUY, L and CHAMPIGNY, M-L (1992) NO₃⁻ enhances the kinase activity for phosphorylation of phosphoenolpyruvate carboxylase and sucrose phosphate synthase proteins in wheat leaves. Plant Physiol. <u>99</u>: pp. 344-347

VIDAL, J and CHOLLET, R (1997) Regulatory phosphorylation of C₄ PEP carboxylase. TIPS <u>2</u>(6): pp230-237

WADHAM, C, WINTER, H and SCHULLER, KA (1996) Regulation of soybean nodule phosphoenolpyruvate carboxylase *in vivo*. Physiol. Plant. <u>97(3)</u>: pp 531-535

WANG, Y-H, DUFF, SMG, LEPINIEC, L, CRETIN, C, SARATH, G, CONDON, SA, VIDAL, J, GADAL, P and CHOLLET, R (1992) Site-directed mutagenesis of the phosphorylatable serine (Ser⁸) in C4 phosphoenolpyruvate carboxylase from Sorghum. J. Biol. Chem. <u>267</u>: pp 16759 - 16762

WANG, Y-H and CHOLLET, R (1993) In vitro phosphorylation of purified tobacco leaf phosphoenolpyruvate carboxylase. FEBS Letts. <u>328</u>: pp 215 - 218

WANG, YH and CHOLLET, R (1993) Partial purification and characterization of phosphoenolpyruvate carboxylase protein-serine kinase from illuminated maize leaves. Arch. Biochem. Biophys. <u>304</u>: pp 496-502

WEDDING, RT, BLACK, MK and MEYER, CR (1990) Inhibition of phosphoenolpyruvate carboxylase by malate. Plant Physiol. <u>92</u>: pp 456-461

WEI, L, HUBBARD, SR, SMITH, RF and ELLIS, L (1994) Protein kinase superfamily -Comparisons of sequence data with 3-dimensional structures. Current Opinion in Structural Biology <u>4</u>(3): pp 450-455

WEIGEND, M (1994) In vitro phosphorylation of phosphoenolpyruvate carboxylase from the facultative CAM plant Mcsembryanthemum crystallinum. J. Plant Physiol. <u>144</u>(6): pp 654-660

WEINER, II (1997) Protein phosphatase 2A in spinach leaves: A sensor for the metabolic phosphate and sugar status? In: Protein Phosphorylation in Plants - Second European

Symposium. (eds. Gadal, P, Kreis, M, Dron, M, Brulfert, J, Bergounioux, C and Vidal, J) pp 136-139 Institut de Biotechnologie des Plantes, Orsay, Paris, France.

WILKINS, MB (1959) An endogenous rhythm in the rate of carbon dioxide output of *Bryophyllum*. I. Some preliminary experiments. J. Exper. Bot. <u>10</u>: pp 377-390

WILKINS, MB (1962) An endogenous rhythm in the rate of carbon dioxide output of Bryophyllum. II. The effects of temperature on the phase and period of the rhythm. Proceedings of the Royal Society, London <u>156</u>:pp 220-241

WILKINS, MB (1984) A rapid circadian rhythm of carbon-dioxide metabolism in Bryophyllum fedtschenkoi. Planta 161; pp 381-184

WILKINS, MB (1992) Tansley Review No. 37 Circadian rhythms: their origin and control. New Phytol. <u>121</u>: pp 347-375

WILLEFORD, KO and WEDDING, RT (1992) Oligomerization and regulation of higher plant phosphoenolpyruvate carboxylase. Plant Physiol. <u>99(2)</u>: pp 755-758

WINTER, K (1982) Properties of phosphoenolpyruvate carboxylasc in rapidly prepared, desalted leaf extracts of the Crassulacean acid metabolism plant *Mesembryanthemum* crystallinum L. Planta <u>154</u>: pp 298-309

WU, MX and WEDDING, RT (1985) Regulation of phosphoenolpyruvate carboxylase from Crassula by interconversion of oligomeric forms. Arch. Biochem. Biophys. <u>240</u>(2): pp 655-662

WU, MX, MEYER, CR, WILLEFORD, KO and WEDDING, RT (1990) Regulation of the aggregation state of maize phosphoenolpyruvate carboxylase - Evidence from dynamic light-scattering measurements. Arch. Biochem. Biophys. <u>281</u>(2): pp 324-329

XING, T, HIGGINS, VJ and BLUMWALD, E (1996) Regulation of plant defense responses to fungal pathogens: Two types of protein kinases in the reversible phosphorylation of the host plasma membrane H^+ -ATPase. The Plant Cell <u>8</u>: pp 555-564

YANAGISAWA, S, IZUI, K, YAMAGUCHI, Y SHIGESADA, K and KATSUKI, H (1988) Further analysis of cDNA clones for maize phosphoenolpyruvate carboxylase involved in C4 photosynthesis - Nucleotide sequence of entire open reading frame and evidence for polyadenylation of messenger RNA at multiple sites *in vivo*. FEBS Letts. **229**(1): pp 107-110

YANAI, Y et al. (1994) Structure of Brassica napus phosphoenolpyruvate carboxylasc genes - Missing introns causing polymorphisms among gene family members. Bioscience Biotechnology and Biochemistry <u>58</u>(5): pp 950-953

с Ц

1282

VIN, Z-H, NEIMANIS, S and HEBER, U (1990) Light-dependent pH changes in leaves of C₃ plants. Planta <u>182</u>: pp 253-261

YIN, Z-H, HEBER, U and RAGHAVENDRA, AS (1993) Light-induced pII changes in leaves of C_4 plants - Comparison of cytosolic alkalization and vacuolar acidification with that of C_3 plants. Planta <u>189</u>(2): pp 267-277

ZHANG, SQ, OUTLAW JR, WH, CHOLLET, R (1994) Lessened malate inhibition of guard cell phosphoenolpyruvate carboxylase velocity during stomatal opening. FEBS Lett. 352: pp 45-48

ZHANG, XQ, LI, B and CHOLLET, R (1995) *In vivo* regulatory phosphorylation of soybean nodule phosphoenolpyruvate carboxylase. Plant Physiol. <u>108</u>: pp 1561-1568

ZHANG, XQ and CHOLLET, R (1997) Phosphoenolpyruvate carboxylase protein kinase from soybean root nodules; partial purification, characterization, and up/down regulation from photosynthate supply from the shoots. Arch. Biochem. Biophys. <u>343</u>(2); pp260-268

